

## Decision Notice

# Retail minus wholesale price control for the WBA market

Decision No:	D01/06
Document No:	06/01
Date:	13 January 2006

## **Contents**

1	Executive summary	3
2	Background	5
3	Decisions	8
	3.1 SUBMISSIONS RECEIVED IN RESPONSE TO DOCUMENT 05/88	
	3.2 DECISIONS	8 9
	3.2.3 Dynamic issues – historic costs or forecasted future costs?	
	3.2.5 What time horizon?	10
	3.2.7 How often should the margin be reassessed?	11
	3.2.9 Individual services or a portfolio?	12
	3.2.10 Wholesale products without retail equivalents	13
	3.2.12How should we approach costs?3.2.13Whose costs?	14
	3.2.14 Assessment of compliance - existing products	
	3.2.16 Price control on connection fee	16
	3.2.18 Retail discount schemes	17
	3.2.20 Other price controls – service establishment, cessation charge and bitstream connection service	
	3.2.21 Publication / notification to the wholesale market	20
	3.2.22 Changes to the decision instrument	
4	Decision Instrument.	
	4.1 STATUTORY POWERS GIVING RISE TO THIS DECISION INSTRUMENT	22
	4.3 SMP OBLIGATIONS IN FORCE	
	4.4 PRICE CONTROL OBLIGATION: GENERAL	
	4.6 NEW PRODUCTS	
	4.7 PROMOTIONS	
	4.8 DISCOUNTS AND BUNDLES	
	4.9 STATUTORY POWERS NOT AFFECTED	
	4.10 EFFECTIVE DATE	
	opendix A – List of Directions and Decisions	
Αŗ	opendix B – Regulatory Impact Assessment	32
	opendix C - Consultation on retail minus wholesale price control for the WE	
m	arket – Document No 05/67	34

Appendix D – Response to consultation and draft direction on retail minus wholesale price control for the WBA market – Document Number 05/88	35
Appendix E – European Commission letter – Case IE/2005/0313: Details of remedies related to case IE/2004/0093	36

## 1 Executive summary

The Commission for Communications Regulation ("ComReg") published document 05/11r on 17 February 2005¹ designating eircom as having Significant Market Power ("SMP") in the market for wholesale broadband access ("WBA") and imposed a number of SMP obligations upon eircom, including obligations in relation to a wholesale price control. In document 05/11r, ComReg imposed an interim "retail minus" price control and committed, following further consultation, to introduce a permanent retail minus price control.

The objective of a permanent price control is to provide predictability and transparency to the marketplace, while preventing the possibility of eircom foreclosing the retail market by means of a margin squeeze. In accordance with this objective, ComReg embarked upon a consultation process. This process commenced with the publication of a consultation document namely, document 05/67. There followed the publication of a response to consultation paper and draft decision notice (document 05/88). The consultation process is now being brought to a conclusion with the publication of this decision notice.

ComReg's overall conclusions may be summarised as follows:

- The price control will be applied ex ante rather than ex post. An ex ante control will preclude a potential margin squeeze, and will offer greater predictability for investment, and transparency in its implementation and operation.
- The overall approach will be forward-looking, assessing economic costs and revenues over time and building in judgements about the likely future value of variables
- eircom's costs and revenues will be used as the basis for establishing those of a similarly efficient operator<sup>2</sup>, and these will be modified to take account of differences in scale, costs which would be borne by efficient new entrants but not by eircom, and of the evolution of prices.
- A discounted cash flow ("DCF") analysis will be adopted, with a DCF analysis being carried out over a 5 year period, and truncated with a terminal value.
- The margin squeeze test will be applied on a product by product basis with a separate control for each wholesale and retail product pair.

ComReg 06/01

<sup>&</sup>lt;sup>1</sup> Document No 05/11r; published on 17/02/05 - Market Analysis - Wholesale Broadband Access (Decision Notice - Designation of SMP and Related Remedies) Decision Notice D3/05.

<sup>&</sup>lt;sup>2</sup> Defined in the "consultation issue leading to Q2" in Section 4.2 of document 05/67 as an operator "which shared the same cost function as eircom's own downstream businesses but which did not yet necessarily enjoy the same economies of scale and scope as eircom's overall business."

- The margin will be reviewed annually, unless circumstances in the market change significantly.
- Where eircom makes changes to the prices of existing products or to promotions relating to existing products, eircom will not be required to submit the proposed changes to ComReg prior to notifying the wholesale market.
- Where eircom introduces new products or new discount schemes, eircom will be required to submit proposals in relation to them to ComReg in advance, and will be required to provide all of the information necessary to assess whether or not there is a potential margin squeeze.
- Where eircom is supporting multiple retail products from a single wholesale product, ComReg concludes that it is appropriate when assessing compliance with the wholesale price control to use the weighted average (by number of subscribers) of the retail products as the reference point for the retail minus control
- The wholesale market should be notified 15 working days in advance of any proposed changes to wholesale prices.

## 2 Background

ComReg published document 05/11r<sup>3</sup> on 17 February 2005 designating eircom Limited ("eircom") as having SMP in the WBA market and imposed a number of SMP obligations upon eircom, including obligations in relation to a wholesale price control.

As part of its analysis of the WBA market, ComReg identified competition problems, in particular, the possible leverage of market power by eircom in the upstream market into the downstream market for broadband services, by way of a margin squeeze (also known as a price squeeze).

In order to address the competition problems identified in the market, ComReg imposed a retail minus price control on an interim basis<sup>4</sup>. At that stage, ComReg also committed to undertaking further consultations in order to establish a permanent retail minus wholesale price control.

Accordingly, in order to determine an appropriate, permanent, wholesale price control, ComReg initiated the current consultation process. The objective of the consultation process was to develop a methodology for calculating a retail minus price control which would be transparent in its methodology and operation, and which would offer greater predictability to the market.

ComReg published document 05/67<sup>5</sup> attached to this document in Annex C. In this consultation document, ComReg set out a number of reasoned proposals for both the methodology for computing appropriate margins and for the operation of a wholesale price control. ComReg asked for industry and other interested parties to respond to this consultation with their views in relation to ComReg's proposals and where relevant, to propose alternative views.

In response to document 05/67, ComReg received submissions from the following organisations<sup>6</sup>:

- Alto:
- BT Ireland;
- Digiweb;
- eircom;
- Ireland Offline; and
- Netsource.

<sup>3</sup> Document No 05/11r; published on 17/02/05 - Market Analysis - Wholesale Broadband Access (Decision Notice - Designation of SMP and Related Remedies) Decision Notice D3/05.

5

ComReg 06/01

<sup>&</sup>lt;sup>4</sup> The imposition of the interim price control was appealed to the Electronic Appeals Panel by eircom and both parties have made various submissions during the course of the appeal.

<sup>&</sup>lt;sup>5</sup> Document No 05/67; published on 19/08/2005 – consultation on retail minus wholesale price control from WBA market.

<sup>&</sup>lt;sup>6</sup> Non-confidential versions of these submissions are published on the ComReg website as document No 05/67s.

ComReg carefully considered the written submissions and in addition conducted numerous bilateral meetings with interested parties, including eircom. ComReg then issued document  $05/88^7$  which was a response to consultation and which also contained a draft decision instrument. Document 05/88 is attached to this decision notice at Annex D.

In document 05/88, ComReg set out its consideration of the issues raised by interested parties in the consultation process. Where, because of issues raised by interested parties, ComReg had revised its proposals, it set out its rationale for the new position. In relation to issues where ComReg had not been convinced by alternative proposals by respondents, ComReg indicated where it found no compelling reason to change its position, as originally put forward in document 05/67 (which is attached in Annex C).

Document 05/88 was published for the purposes of national consultation. As required by the European Communities (Electronic Communications Networks and Services) (Framework) Regulations 2003, the draft decision instrument was simultaneously notified to the European Commission.

In response to the national consultation, ComReg received submissions from the following organisations:

- BT Ireland; and
- Eircom.

The full texts of the responses received by those organisations in relation to document 05/67 and document 05/88 are not annexed to this document but may be viewed on ComReg's website.

The European Commission also responded to document 05/88 by letter dated 21 December, 2005. The EU Commission had no comments in relation to ComReg's proposed measures.

The issues raised in these two submissions are addressed in Section 3 of this document.

The period for consultation has now closed. In this decision notice ComReg (i) summarises the views contained in the responses it received in relation to document 05/88 (ii) sets out its responses to those views (iii) sets out its evidence—based conclusions and issues individual decisions and (iv) finalises the decision instrument which formalises the eircom's SMP obligation of retail minus price control and the obligations with respect to procedures that must be followed by eircom for notifications.

It should be noted that this decision notice relates solely to the wholesale price control obligation in respect of the WBA market. Document 05/11r imposed a

<sup>&</sup>lt;sup>7</sup> Document No 05/88; published on 23/11/2005 – Response to consultation and draft direction on retail minus wholesale price control for the WBA market.

number of other SMP obligations, namely those of access, non-discrimination, transparency and accounting separation.

It should be noted that this decision notice and the decision instrument contained in section 4 supersedes Section 9 of ComReg document 05/11r, being the section entitled "Price Control".

It should also be noted that the retail minus price control obligation, as detailed in this decision notice and its accompanying decision instrument, is not to be viewed in isolation from the other extant SMP obligations (access, non-discrimination, transparency and accounting separation) which were imposed by document 05/11r.

#### 3 Decisions

## 3.1 Submissions received in response to document 05/88

ComReg notified document 05/88 (containing also the draft decision instrument) to the European Commission on 23 November, 2005. The European Commission responded by letter dated 21 December 2005. The European Commission's letter is attached at Annex E of this document.

In its letter, the European Commission described ComReg's procedure to date and the substance of the draft measure. In relation to the procedure adopted and the substance of the draft measure, the European Commission had "no comment" except to state that "ComReg may adopt the resulting draft measures and, where it does so, shall communicate them to the Commission."

ComReg also received two submissions in response to document 05/88 from BT Ireland and eircom. The issues raised in these two submissions are largely reiterations of views already expressed by the same respondents in response to the previous consultation (document 05/67).

While ComReg's deliberations have been ongoing in relation to those views, it is not convinced by some of the alternatives proposed by these respondents and in such instances has found no compelling reason to change its position as a consequence of the views expressed.

In the following section 3.2, ComReg sets out each of the decisions made in document 05/88, and where relevant, responds to views expressed by respondents. Section 4 contains the formal decision instrument which incorporates the decisions in relation to the procedures for the introduction of new products and the implementation of price changes to existing products.

#### 3.2 Decisions

#### 3.2.1 Principles of ex ante and ex post application of retail minus

In document 05/88 ComReg proposed that the wholesale retail minus price control should be applied ex ante.

One respondent responded to this decision and reiterated its views that the risks of inefficient entry in an ex ante test outweighed the benefits of protecting new entrants from a margin squeeze.

ComReg has already responded to this point in section 3.5 of document 05/67 and section 3.2.1 of document 05/88. Those sections set out the rationale for ComReg's decision

While ComReg's deliberations in relation to this point have been ongoing, ComReg has nonetheless not been convinced by the alternative proposed by the respondent

and has thus found no compelling reason to change its position as a consequence of the views expressed.

#### Decision No. 1.

An ex ante retail minus test will be applied.

## 3.2.2 Similarly efficient operator

In document 05/88 ComReg proposed the use of a similarly efficient operator standard as an appropriate benchmark in deriving costs for the wholesale retail minus price control.

One respondent responded to this decision and agreed that a similarly efficient operator was an appropriate benchmark.

ComReg has detailed the reasoning underlying its decision to use a similarly efficient operator in section 4.2 of document 05/67 and section 3.2.2 of document 05/88.

In view of the agreement with its proposal and in the absence of appropriate alternatives, ComReg does not propose to alter its position. ComReg will therefore adopt the proposal as its final decision.

#### Decision No. 2.

A "similarly efficient operator" constitutes an appropriate benchmark for a margin squeeze analysis.

## 3.2.3 Dynamic issues – historic costs or forecasted future costs?

In document 05/88 ComReg proposed the use of forward looking assumptions on the evolution of costs and revenues.

One respondent responded to this decision and agreed that it was appropriate to use future costs and revenues.

ComReg detailed its reasoning around historic versus forecasted costs in section 4.3 of document 05/67 and section 3.2.3 of document 05/88.

In view of the agreement with its proposal and having given further consideration to it, ComReg does not propose to alter its position. ComReg will therefore adopt the proposal as its final decision.

#### Decision No. 3.

Future costs and revenues will be used in ComReg's analysis.

#### 3.2.4 Discounted cash flow analysis

ComReg proposed the use of a DCF methodology to assess margins.

One respondent responded to this decision and stated that a DCF method should be used when an ex post test of margins is undertaken across a portfolio of services offered by the benchmark operator.

ComReg's reasoning for using a DCF analysis is set out in its consultation in section 4.4 and is further elaborated upon in its response to consultation in section 3.2.4.

In relation to the issue of ex post versus ex ante, ComReg has already responded in section 3.5 of document 05/67 and section 3.2.1 of document 05/88 and section 3.2.1 in this decision notice. It has also responded in relation to the issue of portfolio versus individual product basis, in section 4.9 of document 05/67 and section 3.2.9 of document 05/88.

In view of the agreement with its proposal and having given further consideration to the issue, ComReg does not propose to alter its position. ComReg will therefore adopt its proposal as its final decision.

#### Decision No. 4.

A discounted cash flow model will be used to assess margins.

#### 3.2.5 What time horizon?

In document 05/88, ComReg proposed that five years was the minimum credible period to model a hypothetical operator from entry in the market to maturity in the market

Two respondents responded to this proposal. One respondent reiterated its point (made above in section 3.2.4) that where a DCF method was used in an ex post margin test across a portfolio of services offered by the benchmark operator, five years was an appropriate period.

The second respondent restated its preference for a shorter time period reiterating that in this market fixing the DCF at a very rigid timescale would be problematic. It accepted the need for stability and certainty in regulation but stated that what it perceived as such a long period, might hamper the regulator's ability to adjust to changes.

ComReg has already responded to these views in section 3.2.4 above, in document 05/67 in section 4.5 and in document 05/88 in section 3.2.5. Those sections also set out the rationale for ComReg's decision.

ComReg reiterates that while the modelling period is five years, the model is being reassessed on an annual basis - see in this regard section 3.2.7.

ComReg's deliberations in relation to this point have been ongoing. ComReg however remains unconvinced by the alternative proposed by the respondent and has therefore found no compelling reason to change its position as a consequence of the views expressed. ComReg will therefore adopt the proposal as its final decision.

#### Decision No. 5.

The discounted cash flow analysis will be carried out over five years.

#### 3.2.6 Terminal value

In document 05/88, ComReg proposed the use of a terminal value. It also proposed using the net cash flow of the final year, carried forward for three subsequent years as an appropriate method of computing the terminal value.

One respondent indicated its support for the methodology used by ComReg to calculate the terminal value and reiterated that it was also the option that it had advocated in its submission in response to ComReg's original consultation 05/67.

ComReg has set out the rationale underlying its decision in relation to the terminal value methodology in section 4.6 of document 05/67 and section 3.2.6 of document 05/88.

In view of the agreement with its proposal and having ComReg's continued belief that the methodology proposed is the most appropriate, ComReg does not propose to alter its position. ComReg will therefore adopt the proposal as its final decision.

#### Decision No. 6.

ComReg will apply a terminal value. The terminal value will be calculated by using the net cash flow of the fifth year of the discounted cash flow analysis, carried forward for three subsequent years.

#### 3.2.7 How often should the margin be reassessed?

In document 05/88 ComReg proposed that the margin should be reassessed annually. However, ComReg also stated that significant unforeseen developments in the market could require an amendment to the control.

One respondent reiterated its point that reviews should occur more frequently than on an annual basis, in order to combat the risk of inefficient entry which can occur due to excessive margins.

A second respondent welcomed ComReg's "pragmatic" approach in reviewing the price control process in the event of significant unforeseen changes in the market place.

ComReg has already responded to the request for more frequent reviews made by the first respondent in document 05/88 in section 3.2.7. ComReg's rationale underlying this decision was further elaborated upon in section 4.7 of document 05/67.

While ComReg's deliberations in relation to this issue have been ongoing, ComReg has nonetheless not been convinced by the alternative proposed by the respondent and has therefore found no compelling reason to change its position as a consequence of the views expressed. ComReg remains of the view that the margin should be subject to annual review. ComReg will therefore adopt the proposal as its final decision.

#### Decision No. 7.

The margin will be subject to annual review. In the event of significant unforeseen changes in the marketplace, ComReg will review the price control and may issue amendments.

#### 3.2.8 What discount rate should be applied?

In document 05/88, ComReg proposed the use of eircom's cost of capital as the discount rate for the DCF analysis.

One respondent reiterated its view that eircom's cost of capital was not a suitable discount factor for a reseller of wholesale broadband services with minimal investment in its own network. This respondent stated that a lower discount factor taking into account the cost of commercial debt would be more appropriate.

ComReg has already responded to this point in document 05/88 in section 3.2.8. ComReg's original rationale for this decision is also elaborated in section 4.8 of document 05/67.

While ComReg's deliberations in relation to this point have been ongoing, ComReg has nonetheless not been convinced by the alternative proposed by the respondent and has thus found no compelling reason to change its position as a consequence of the views expressed. ComReg will therefore adopt the proposal as its final decision.

#### Decision No. 8.

eircom's cost of capital will be used as the basis for calculating the discount factor in the discounted cash flow analysis.

#### 3.2.9 Individual services or a portfolio?

In document 05/88 ComReg proposed that wholesale products should be reviewed on an individual wholesale product by wholesale product basis.

One respondent restated its view that the margin test should be applied on a portfolio basis rather than on the basis of individual products.

ComReg has previously responded to this point in section 3.2.9 of document 05/88. It has also set out its reasoning for this decision in section 4.9 of document 05/67.

While ComReg's deliberations in relation to this point have been ongoing, ComReg is not convinced of the appropriateness of the alternative proposed by the respondent and has thus found no compelling reason to change its position as a consequence of the views expressed. ComReg will therefore adopt the proposal as its final decision.

#### Decision No. 9.

The margin squeeze test will be conducted on an individual, product by product basis.

## 3.2.10 Wholesale products without retail equivalents

In document 05/88, ComReg proposed that commercial negotiation should be used in the first instance to determine the price of wholesale products not availed of by eircom's retail arm.

One respondent expressed its agreement with ComReg's position that the price for wholesale products should be set through commercial negotiation.

ComReg has set out the rationale for commercial negotiation being the "first port of call" and for its deviation from its original proposals in section 4.10 of document 05/67 and in section 3.2.10 of document 05/88. Those sections also provide the underlying rationale for ComReg's decision.

In view of the agreement with its amended proposal and having given further consideration to the issue, ComReg does not propose to alter its position again and will adopt the proposal as its final decision.

#### Decision No. 10.

Commercial negotiation will be used to determine the price for wholesale products that do not have an eircom retail equivalent. If such negotiation fails to determine a reasonable wholesale price, ComReg will intervene.

### 3.2.11 Retail prices

In document 05/88, ComReg proposed that sensitivity analysis should be performed around the rates of decrease of retail prices.

One respondent agreed that sensitivity analysis around the rates of decrease of retail prices was necessary.

ComReg has set out the underlying rationale for this decision in section 4.11 of document 05/67 and section 3.2.11 of document 05/88.

In view of the agreement with its proposal and having given further consideration to the issue, ComReg will adopt this proposal as its final decision.

#### Decision No. 11.

ComReg will perform sensitivity analysis around rates of decrease of retail prices in doing its discounted cash flow evaluation of margins.

#### 3.2.12 How should we approach costs?

In document 05/88 ComReg proposed a forward looking approach to assessing costs.

One respondent reiterated its view that a forward looking approach using future costs and revenues was consistent with Decision No.3 of this Decision Notice.

ComReg has detailed its underlying rationale for this decision in section 4.12 of document 05/67 and section 3.2.12 of document 05/88.

In view of the agreement with its proposal ComReg does not propose to alter its position and will therefore adopt the proposal as its final decision.

#### Decision No. 12.

ComReg will use a forward looking approach in assessing costs.

#### 3.2.13 Whose costs?

In document 05/88 ComReg proposed to use eircom's costs as a starting point, adjusted for the unavoidable costs incurred by a similarly efficient new entrant.

One respondent reiterated its agreement that eircom's costs should be used as the basis for establishing the costs of a similarly efficient operator.

ComReg has elaborated upon its reasoning for this decision in section 4.13 of document 05/67 and section 3.2.13 of document 05/88.

In view of the agreement with its proposal, ComReg does not propose to alter its position and will therefore adopt the proposal as its final decision.

#### Decision No. 13.

eircom's costs should be used as the basis for establishing the costs of a similarly efficient operator.

#### 3.2.14 Assessment of compliance - existing products

In document 05/88, ComReg proposed that price changes to existing products should be assessed through a statement of compliance with the published price control that

will be issued to ComReg by eircom. It was proposed that eircom would simultaneously notify these wholesale prices to other operators.

One respondent reiterated its agreement that when changing prices to existing products, eircom should provide ComReg with a statement of compliance and simultaneously notify other operators of the wholesale prices.

A second respondent reiterated its support for the notification period.

This respondent also stated that if the price change was complex and the compliance note was not clear, then ComReg may require at least 15 days in order to carry out the appropriate analysis. In such a case, this respondent submitted, some form of communication should be made to industry at around day 10, so as to indicate a completion date for ComReg's analysis.

Finally, this respondent suggested that eircom and ComReg assign extra time for the first 2 or 3 occurrences and once the models had been refined, future testing could then be more methodical and timely.

ComReg has provided detailed reasoning for its decision in section 5.1 of document 05/67. It has also responded to the views of respondents in document 05/88 in sections 3.2.14, as well as setting out the detailed provisions regarding notifications in the draft decision instrument (attached at Appendix B in 05/88).

In relation to the new points raised by the second respondent, ComReg would respond as follows. This respondent's points are addressed in the decision instrument in section 4 of this document which stipulates that eircom must provide the required level of detail to demonstrate its compliance. ComReg and eircom have in fact already examined potential scenarios, price changes and product changes in anticipation of these issues. Therefore, ComReg has concluded that it will not be necessary to have a different procedure for the first 2 or 3 instances.

While ComReg's deliberations in relation to this point have been ongoing, ComReg has nonetheless not been convinced by the alternative proposed by the respondent and has thus found no compelling reason to change its position as a consequence of the views expressed.

#### Decision No. 14.

See section 4 – the decision instrument.

#### 3.2.15 New products

In document 05/88, ComReg proposed that in relation to any new products or services, eircom should formally submit the price details to ComReg. ComReg would then have a 5 working day period where it would review the new product and then could do one or more things, including issuing an opinion in relation to eircom's compliance.

One respondent agreed that the price control should be applied to the equivalent offering of any new product, as described in section 6.2 - 6.5 of Appendix B of document 0.5/88.

A second respondent pointed out that if eircom introduced a completely new pricing methodology (as emerged in the UK over recent years), then 15 days would not be sufficient to fully comprehend the implications of such changes. This respondent stated that such a fundamental change would require a notification period of not less than 3 months

ComReg has provided detailed rationale for its decision in section 5.1 of document 05/67. It has also responded to the views of respondents in document 05/88 in sections 3.2.15 as well as setting out detailed provisions in its draft decision instrument in Appendix B of document 05/88.

In relation to the second respondent's point, ComReg has already addressed this point in decision 7 (i.e. section 3.2.7 of this decision notice) which states that in the event of significant unforeseen changes in the market place, ComReg will review the price control and may issue amendments.

ComReg would also reiterate that if eircom were to make significant changes to the basis of charging, eircom also has other SMP obligations imposed through document 05/11r, including the obligation of non-discrimination which must be complied with.

While ComReg's deliberations in relation to this matter have been ongoing, ComReg is not convinced by the alternative proposed by the respondent and has found no compelling reason to change its position as a consequence of the views expressed. ComReg will therefore adopt its proposal as its final decision.

#### Decision No. 15.

See section 4 – the decision instrument.

#### 3.2.16 Price control on connection fee

In document 05/88, ComReg proposed that the wholesale connection fee and port transfer charge should be cost oriented and reduced to €30.

One respondent agreed with the €30 charge for wholesale connection and port transfer.

ComReg detailed its original proposals in section 5.2.1 of document 05/67. In section 3.2.16 of document 05/88, ComReg elaborated upon its rationale for deviating from its original proposals while also responding to views expressed by respondents. The sections referred to set out the underlying rationale for ComReg's decision.

In view of the agreement with its proposal, ComReg does not propose to alter its position and will therefore adopt its proposal as its final decision

#### Decision No. 16.

The prices for wholesale connection and port transfer shall be cost oriented. From 1 January 2006, the price for wholesale connection and port transfer shall be €30. eircom shall not increase this charge without the prior approval of ComReg.

#### 3.2.17 Price control on wholesale monthly rental

In document 05/88, ComReg proposed a formula for the retail minus price control, such that wholesale rental price = retail rental price \* (1 - fixed percentage < 100%) – fixed monetary value.

One respondent reiterated its agreement that for products with usage charges the average revenue achieved should be used as the basis for the margin test.

ComReg has set out the rationale for its decision in section 5.2 of document 05/67. It has also responded to the views of respondents in document 05/88 in sections 3.2.17 as well as detailing the draft decision instrument in Appendix B of 05/88 of 05/88 of 05/88 of 05/88.

In view of the agreement with its proposal and having given further thought to the issue, ComReg does not propose to alter its position. ComReg will therefore adopt its proposal as its final decision.

#### Decision No. 17.

See section 4 – the decision instrument.

#### 3.2.18 Retail discount schemes

In document 05/88, ComReg proposed that if eircom wished to amend or to introduce a new discount scheme, it would require a submission prior to its launch in line with the procedure for the introduction of new products or services. Where a discount scheme could be justified by savings at the retail level, which would be equally available to OAOs, there would not be a need for wholesale price reductions<sup>8</sup>. However, where a discount could not be objectively justified by savings at the retail level, ComReg would regard this as a change in the retail price and would require corresponding wholesale amendments. In such circumstances, compliance with the retail minus control would be assessed on the weighted average of discounted and undiscounted lines.

One respondent agreed that where retail discounts are based on savings at the retail level, there should be no adjustment to wholesale prices.

<sup>&</sup>lt;sup>8</sup> Currently, eircom offer one discount scheme – the bulk order. This discount is justified by the savings at the retail level associated with savings from the ordering process.

ComReg has provided detailed rationale for its decision in section 5.2.3 of document 05/67. It has also responded to the views of respondents in document 05/88 in sections 3.2.18 as well as detailing the draft decision instrument in Appendix B of that document

In view of the agreement with its proposal and having given further thought to the issue, ComReg does not propose to alter its position. ComReg will therefore adopt its proposal as its final decision.

#### Decision No. 18.

See section 4 – the decision instrument.

#### 3.2.19 Retail promotions

In document 05/88, ComReg proposed that where eircom introduces retail promotions, such promotions must be matched at the wholesale level. ComReg also proposed that eircom would issue ComReg with a statement of compliance and simultaneously notify other operators of the imminent amendment of the wholesale terms and conditions.

One respondent agreed that it was appropriate that when introducing or amending promotions on existing products eircom must provide ComReg with a statement of compliance with the price control obligation.

A second respondent stated that the timescales for rebates needed to be more stringently defined as it stated there was an incentive for eircom to delay such payments. It suggested that any rebate payment should be made within 3 months of the start of the promotion to ensure that the full costs are known to eircom.

ComReg has set out the reasoning for its decision in section 5.2.4 of document 05/67. It has also responded to the views of respondents in document 05/88 in sections 3.2.19 as well as detailing the draft decision instrument in Appendix B.

In relation to the second respondent's point (which had previously been expressed by that respondent), ComReg has already indicated in document 05/88 that the rebate must be paid within "a reasonable period of time". It would be inappropriate to stipulate a 3 month deadline at this stage given that there has been no evidence to date of eircom unduly delaying payments of rebates. If such evidence is presented to ComReg, it will reassess the timescale for rebate payments and amend this decision accordingly.

While ComReg's deliberations in relation to this point have therefore been ongoing, ComReg has nonetheless not been convinced by the alternative proposed by the respondent and has thus found no compelling reason to change its position as a consequence of the views expressed.

#### Decision No. 19.

#### See section 4 – the decision instrument.

## 3.2.20 Other price controls – service establishment, cessation charge and bitstream connection service

In document 05/88, ComReg proposed that the existing arrangements in relation to the cessation charge should remain in place. However, ComReg also indicated that in the future new products will be developed and reasonable prices should be achieved through commercial negotiation.

The service establishment charge will be reviewed at the start of each new price control period. Any change in the service establishment or the cessation charge would require prior approval from ComReg.

In relation to the bitstream connection, ComReg proposed that access should be granted at reasonable prices. In considering what is reasonable ComReg will have regard, inter alia, to the cost of providing connection and the price of other comparable services.

One respondent stated its agreement with ComReg's proposed decisions 20, 21 and 22 in 05/88.

ComReg originally detailed its reasoning in section 5.2.5 of document 05/67. ComReg then evolved its position, where relevant, in section 3.2.20 of document 05/88. Those sections provide the rationale for ComReg's decisions.

In view of the agreement with its proposal ComReg does not propose to alter its position. ComReg will therefore adopt its proposal as its final decision.

#### Decision No. 20.

The current arrangements, where eircom offer two contractual alternatives - one requiring a commitment to a six month minimum term but no cessation fee, the other requiring no commitment to minimum term but requiring a cessation fee - are reasonable and amending this would be inappropriate. Over time new products, more tailored to individual operators' requirements, will be developed and reasonable prices should be achievable through commercial negotiation. Should this not happen ComReg would intervene to ensure that prices are reasonable.

#### Decision No. 21.

The service establishment charge will be reviewed at the commencement of each new price control period. In addition, a change in price or terms and conditions for the service establishment charge or the cessation charge requires prior approval from ComReg. In order to grant such approval, ComReg requires that eircom submit a cost based justification.

#### Decision No. 22.

The prices for bitstream connection services will not have specific obligations other than that access should be at reasonable prices.

#### 3.2.21 Publication / notification to the wholesale market

In document 05/88, ComReg proposed that changes to the wholesale prices should be notified to the wholesale market 15 days before coming into effect. This means that eircom wholesale cannot accept any orders in that 15 day notification period, either from its own retail arm or from OAOs.

One respondent reiterated its preference for a shorter notification period for wholesale price changes as it can cause confusion at the point of order when a new price is lower than an existing price.

ComReg detailed its reasoning in section 5.3 of document 05/67 and section 3.2.21 of its response to consultation. Those sections provide the rationale for ComReg's decision.

While ComReg's deliberations in relation to this point have been ongoing, ComReg has nonetheless not been convinced by the alternative proposed by the respondent and has thus found no compelling reason to change its position as a consequence of the views expressed.

#### Decision No. 23.

See section 4 – the decision instrument.

#### 3.2.22 Changes to the decision instrument

Apart from punctuation or grammatical amendments the only amendment to the decision instrument is to Table 1. It was indicated in document 05/88 that Table 1 might require updating should eircom amend the technical characteristics of the wholesale products between the date that the draft decision instrument (as contained in document 05/88) was notified to the EU Commission and the adoption of the decision instrument in ComReg's final decision. It should be noted that Table 1 has now been updated in this document to take account of amendments that have actually occurred.

#### 4 Decision Instrument

## 4.1 Statutory Powers Giving Rise to this Decision instrument

This decision instrument relates to the market for wholesale broadband access ("WBA") for the purpose of delivering broadband content to end users<sup>9</sup> and supersedes Section 9 of document  $05/11r^{10}$ , being the section entitled "Price Control". This decision instrument is made by the Commission for Communications Regulation ("ComReg"):

- I. Having had regard to sections 10 and 12 of the Communications Regulations Act 2002;
- II. Having taken account of its functions under Regulation 6 (1) of the European Communities (Electronic Communications Networks and Services) (Access) Regulations 2003<sup>11</sup>;
- III. Having (where appropriate) complied with the Policy Directions made by the Minister<sup>12</sup>;
- IV. Having taken the utmost account of the EU Commission's Recommendation<sup>13</sup> and the SMP Guidelines<sup>14</sup>;
- V. Having had regard to the market definition, market analysis and reasoning conducted by ComReg in document 05/11r, the analysis and reasoning set out in document 05/67<sup>15</sup> and the reasoning and individual decisions set out

<sup>&</sup>lt;sup>9</sup> As referred to in EU Commission Recommendation of 11 February, 2003 on Relevant Product and Service Markets within the electronic communications sector susceptible to ex ante regulation in accordance with Directive 2002/21/EC of the European Parliament and the Council of 7 March 2002 on a common regulatory framework for electronic communications networks and services.

<sup>&</sup>lt;sup>10</sup> Document No 05/11r; published on 17/02/05 - Market Analysis – Wholesale Broadband Access (Decision Notice - Designation of SMP and Related Remedies). Decision Notice D3/05.

<sup>&</sup>lt;sup>11</sup> S.I. No. 305 of 2003 the European Communities (Electronic Communications Networks and Services) (Access) Regulations 2003 which transposes Directive 2002/19/EC of the European Parliament and the Council of 7 March 2002 on access to, and interconnection of, electronic communications networks and associated facilities.

<sup>&</sup>lt;sup>12</sup> Policy Directions made by Dermot Ahern T.D. (the then) Minister for Communications, Marine and Natural Resources on 21 February 2003 and 26 March 2004.

<sup>&</sup>lt;sup>13</sup> EU Commission Recommendation of 11 February, 2003 on Relevant Product and Service Markets within the electronic communications sector susceptible to ex ante regulation in accordance with Directive 2002/21/EC of the European Parliament and the Council of 7 March 2002 on a common regulatory framework for electronic communications networks and services.

<sup>&</sup>lt;sup>14</sup> Commission Guidelines on market analysis and the assessment of significant market power under the Community regulatory framework for electronic communications networks and services.

 $<sup>^{15}</sup>$  Document No 05/67; published on 19/08/2005 – consultation on retail minus wholesale price control for the WBA market.

- previously in this decision notice, each of which form part of and shall be construed with this decision instrument;
- VI. Having taken account of the submissions received in relation to document 05/67 and document 05/88<sup>16</sup>; and
- VII. Pursuant to Regulations 27 (4) of the European Communities (Electronic Communications Networks and Services) (Framework) Regulations 2003<sup>17</sup> and Regulations 9 and 14 of the European Communities (Electronic Communications Networks and Services) (Access) Regulations 2003.

#### 4.2 **DEFINITIONS**

In this decision instrument, unless the context otherwise suggests:

- "Access Regulations" means the European Communities (Electronic Communications Networks and Services) (Access) Regulations 2003;
- "amendment" means, in respect of a tariff for any existing product, a change, adjustment, modification and any other cognate word or expression;
- **"bundle"** means a package, consisting of both a product and one or more goods and / or services, which is on offer or on sale to end users;
- "decision instrument" means this decision instrument;

- "discount" means an offer or sale of a product at less than its standard price. Examples shall include: a price reduction (including a volume related price reduction) a rebate, a reimbursement, a refund, a set-off and any other cognate words or expressions;
- **"end user"** has the same meaning as in Regulation 2 (1) of the European Communities (Electronic Communications Networks and Services) (Framework) Regulations 2003);
- "existing product" means any product on offer or on sale to end users, on or prior to the effective date of this decision instrument and any new product, in respect of which ComReg has issued an opinion and a confirmation, as referred to in section 4.6.5 of this decision instrument;

<sup>&</sup>lt;sup>16</sup> Document No 05/88; published on 23/11/2005 – Response to consultation and draft direction on retail minus wholesale price control for the WBA market.

<sup>&</sup>lt;sup>17</sup> S.I. No. 307 of 2003 the European Communities (Electronic Communications Networks and Services) (Framework) Regulations 2003 which transposes Directive 2002/21/EC of the European Parliament and the Council of 7 March 2002 on a common regulatory framework for electronic communications networks and services.

"new product" means any product on offer or on sale subsequent to the effective date of this decision instrument, which has different functional and / or technical characteristics and / or a different tariff structure to existing products;

"product" means any *eircom* retail broadband product on offer or on sale to end users which uses *eircom's* copper network equipment to transmit data signals and shall include existing products and new products;

**"OAO"** means other authorised operator, being an authorised undertaking for the purposes of the European Communities (Electronic Communications Networks and Services) (Authorisation) Regulations 2003;

"promotion" means an offer in respect of a product which is available for a finite period of time and which offers a tariff reduction but does not include incentives provided for within normal sales costs;

"SMP" means significant market power, which has the same meaning as that contained in Regulation 25 of the European Communities (Electronic Communications Networks and Services) (Framework) Regulations 2003; and

"working day" means a day other than Saturday, Sunday a bank holiday or a public holiday.

#### 4.3 SMP OBLIGATIONS IN FORCE

- 4.3.1 By virtue of document 05/11r published on 17 February, 2005, *eircom* had imposed upon it the SMP obligations of access, non-discrimination, transparency, price control and cost accounting.
- 4.3.2 *eircom* notified an appeal to the Electronic Communications Appeals Panel ("the ECAP") on 16 March, 2005 in respect of the price control SMP obligation. On the 24 June, 2005 the Minister for Communications, Marine and Natural resources appointed the ECAP to hear the notified appeal.
- 4.3.3 The other SMP obligations contained in document 05/11r are not the subject of any appeal to the ECAP or any judicial proceedings by any OAO or any other party. Accordingly, those SMP obligations remain in force in their entirety and shall where necessary be construed together with this decision instrument.

## 4.4 PRICE CONTROL OBLIGATION: GENERAL

- 4.4.1 *eircom* shall, pursuant to Regulation 14 of the Access Regulations, have SMP obligations relating to wholesale price control in the market for WBA, in the form of a retail minus price control, for the following purposes:
  - I. Establishing a wholesale price for products, promotions, discounts and bundles in the market for WBA;

- II. Preventing the potential application by *eircom* of a price or margin squeeze in the market for WBA. In this regard, *eircom* shall not create a margin squeeze in the market for WBA; and
- III. Preventing *eircom* from sustaining prices at an excessively high level in the market for WBA

#### 4.5 AMENDMENTS TO EXISTING PRODUCTS

- 4.5.1 Sections 4.5.2 4.5.6 apply to any amendment.
- 4.5.2 In this section, "retail rental price" means the retail price where a single product is supported by a single wholesale offering or, the weighted average (by number of subscribers) of the retail products' individual prices where more than one product is supported by a single wholesale offering. The retail minus price control applying to existing products shall be established by reference to the following formula:

Wholesale Rental Price = retail rental price \* (1 - fixed percentage < 100%) - fixed monetary value ("the price control")<sup>18</sup>.

- 4.5.3 *eircom* shall ensure that it applies the price control to the equivalent wholesale offering of any existing product. Amendments shall be in accordance with the price control and as specified in Table 1 below. Table 1 may from time to time be amended by ComReg for any of the following reasons:
  - I. When an equivalent wholesale offering of any new product is introduced;
  - II. When ComReg conducts a review of the price control; or
  - III. If ComReg decides to amend the price control or the obligations referred to in this decision instrument as a result of any material unforeseen changes in market conditions that in ComReg's opinion have a material impact on the market.

<sup>&</sup>lt;sup>18</sup> The following is an example, for illustrative purposes, of an application of the price control in practice: if the retail price for the "1024/128 @ 48:1 (IP)" product was to be €50, the wholesale price could be no more than €50 \* (1-29%) - €4.37 = €31.13.

Table 1

Wholesale Product	(	Control
Description <sup>19</sup>	Percentage (%)	Fixed Monetary Value (€)
1024/128 @ 48:1 (IP) plus usage charge <sup>20</sup>	30%	€3.70
1024/128 @ 48:1 (IP)	29%	€4.37
2048/256 @ 48:1 (IP)	28%	€4.70
3072/384 @ 24:1 (IP)	26%	€6.75
4096/384 @ 24:1 (IP)	24%	€12.64
5120/512 @ 24:1 (IP)	24%	€28.44

4.5.4 No later than fifteen working days prior to the date that an amendment is to become operative, *eircom* shall furnish to ComReg a detailed written statement of compliance demonstrating *eircom's* compliance with the price control and the obligations referred to in this decision instrument. The statement of compliance shall make full and true disclosure of all material facts for the purpose of demonstrating compliance with the price control and the obligations referred to in this decision instrument and shall state precisely and in all respects how *eircom* is in compliance with the price control and the obligations referred to in this decision instrument. The statement of compliance shall also have appended to it, all relevant supporting documentation for the purpose of demonstrating compliance with the price control and the obligations referred to in this decision instrument. The statement of compliance shall demonstrate how any adjustments to the price of the equivalent wholesale offering of an existing product are and will be in compliance with the price control and the obligations referred to in this decision instrument.

<sup>&</sup>lt;sup>19</sup> It was indicated in document 05/88 that Table 1 might require updating should *eircom* amend the technical characteristics of the wholesale products between the date that the draft decision instrument (as contained in document 05/88) was notified to the EU Commission and the adoption of the decision instrument in ComReg's final decision. It should be noted that Table 1 has now been updated in this document to take account of amendments that have actually occurred.

<sup>&</sup>lt;sup>20</sup> Where usage charges make up a significant portion of retail or wholesale prices, these charges will be converted into average monthly revenues and the price control will be applied accordingly.

- 4.5.5 At the same time that *eircom* furnishes the statement of compliance referred to in section 4.5.4 to ComReg, it shall furnish all OAOs availing of or intending to avail of existing products, with written notification of corresponding adjustments to an equivalent wholesale offering of any existing product. *eircom* shall furnish ComReg with a copy of the notification and written confirmation that all OAOs have been furnished with the notification.
- 4.5.6 Upon receipt of the statement of compliance and the notification referred to in sections 4.5.4 and 4.5.5 respectively, ComReg shall review the statement of compliance. Within the fifteen working day period referred to in section 4.5.4, ComReg may do one or more of the following things:
  - I. Provide *eircom* with both (a) an appropriate written opinion in relation to the statement of compliance referred to in section 4.5.4 and (b) written confirmation that the making available or offering for sale of the existing product is conditional only upon *eircom* being in compliance with its obligation in section 4.5.3;
  - II. Request any further information from *eircom* and set a deadline by which such information shall be provided. *eircom* shall provide the requested information by the deadline and in such format and to the level of detail as stipulated by ComReg. Upon receipt of the requested information from *eircom* and within the fifteen working day period referred to in section 4.5.4, ComReg may do one or more of the things referred to in sub-sections I, III or IV of this section;
  - III. Inform *eircom* in writing that the amendment(s) would in ComReg's opinion, not be in compliance with the price control and the obligations referred to in this decision instrument, giving reasons therefor and also inform *eircom* that the amendment if made operative will or could result in the issuing of a notification of non-compliance under Regulation 18 (1) of the Access Regulations; or
  - IV. For the purpose of further specifying requirements to be complied with by *eircom* relating to the price control and the obligations referred to in this decision instrument, issue a direction or directions to *eircom* under Regulation 17 of the Access Regulations, to refrain from making operative the corresponding adjustment(s) to the equivalent wholesale offering of any existing product.

## 4.6 NEW PRODUCTS

- 4.6.1 Sections 4.6.2 4.6.5 apply to any new product.
- 4.6.2 *eircom* shall not make available or offer for sale, the equivalent wholesale offering of any new product, until such time as ComReg does that which is specified in section 4.6.5, sub-section I (either within the five working day period referred to in

- section 4.6.5 or, upon completion of the review by ComReg referred to in section 4.6.5, sub-section II).
- 4.6.3 *eircom* shall apply the price control to the equivalent wholesale offering of any new product.
- 4.6.4 Prior to the date that a new product is to be made available or offered for sale, *eircom* shall furnish to ComReg a detailed written submission demonstrating *eircom's* proposed compliance with the price control and the obligations referred to in this decision instrument. The submission shall make full and true disclosure of all material facts for the purpose of demonstrating proposed compliance with the price control and the obligations referred to in this decision instrument and shall state precisely and in all respects how *eircom* would be in compliance with the price control and the obligations referred to in this decision instrument. The submission referred to in this section, shall also have appended to it, all relevant supporting documentation for the purpose of demonstrating *eircom's* proposed compliance with the price control and the obligations referred to in this decision instrument. The submission shall demonstrate how any adjustments to the price of the equivalent wholesale offering of a new product would be in compliance with the price control and the obligations referred to in this decision instrument.
- 4.6.5 Upon receipt of the submission referred to in section 4.6.4, ComReg shall review same. Within five working days, ComReg may do one or more of the following things:
  - I. Provide *eircom* with both (a) an appropriate written opinion in relation to the submission referred to in section 4.6.4 and (b) written confirmation that the making available or offering for sale of the new product is conditional only upon *eircom* being in compliance with its obligation in section 4.6.3. Once ComReg provides *eircom* with the written opinion and confirmation referred to in this sub-section, *eircom* shall provide OAOs that are availing of any existing products (or intending to avail of any existing products) with fifteen working days' prior written notification of corresponding adjustments to the equivalent wholesale offerings of any new products before making such adjustments. *eircom* shall at the same time furnish ComReg with written confirmation that all OAOs have been furnished with the said notification;
  - II. Inform *eircom* in writing that further analysis of the submission referred to in section 4.6.4 is required by ComReg, in which case ComReg may request any further information from *eircom* and set a deadline by which such information shall be provided. *eircom* shall provide any information that is requested by the deadline, in such format and to the level of detail as stipulated by ComReg. Upon receipt of the requested information, ComReg shall furnish *eircom* with an indicative timetable in writing for the completion of its further analysis. Upon completion of this analysis, ComReg may do one or more of the things referred to in sub-sections I, III, IV or V of this section;

- III. Inform *eircom* in writing that the new product(s) would in ComReg's opinion, not be in compliance with the price control and the obligations referred to in this decision instrument, giving reasons therefor and also inform *eircom* that the new product(s) if made available or offered for sale, will or could result in the issuing of a notification of non-compliance under Regulation 18 (1) of the Access Regulations;
- IV. For the purpose of further specifying requirements to be complied with by *eircom* relating to the price control and the obligations referred to in this decision instrument, issue a direction or directions to *eircom* under Regulation 17 of the Access Regulations, requiring that *eircom* makes specified corresponding adjustments to the price of the equivalent wholesale offering of any new product; or
- V. For the purpose of further specifying requirements to be complied with by *eircom* relating to the price control and the obligations referred to in this decision instrument, issue a direction or directions to *eircom* under Regulation 17 of the Access Regulations, to refrain from making available or offering for sale, the equivalent wholesale offering of any new product.

#### 4.7 PROMOTIONS

4.7.1 Sections 4.5.2 - 4.5.6 (excluding Table 1) in relation to amendments to existing products shall apply in like manner to promotions.

#### 4.8 DISCOUNTS AND BUNDLES

4.8.1 Sections 4.6.2 – 4.6.5 in relation to new products shall apply in like manner to discounts and bundles and shall apply to an adjustment to a discount or bundle available, or on sale, on or prior to the effective date of this decision instrument and to a new discount or bundle that is made available or offered for sale after the effective date of this decision instrument

#### 4.9 STATUTORY POWERS NOT AFFECTED

4.9.1 Nothing in this decision instrument shall operate to limit ComReg in the exercise and performance of its statutory functions, powers and duties under any primary or secondary legislation (in force prior to or after the effective date of this decision instrument) from time to time as the occasion may require.

#### **4.10 EFFECTIVE DATE**

4.10.1 This decision instrument shall be effective from the 13 day of January 2006 until further notice by ComReg.

ISOLDE GOGGIN CHAIRPERSON THE COMMISSION FOR COMMUNICATIONS REGULATION THE 13 DAY OF January 2006

## Appendix A – List of Directions and Decisions

## **List of Decisions**

Decision No. 1.	9
An ex ante retail minus test will be applied.	9
Decision No. 2.	9
A "similarly efficient operator" constitutes an appropriate benchmark for a	
margin squeeze analysis.	9
Decision No. 3.	9
Future costs and revenues will be used in ComReg's analysis.	9
Decision No. 4.	10
A discounted cash flow model will be used to assess margins.	10
Decision No. 5.	11
The discounted cash flow analysis will be carried out over five years.	11
Decision No. 6.	11
ComReg will apply a terminal value. The terminal value will be calculated by	
using the net cash flow of the fifth year of the discounted cash flow analysis,	
carried forward for three subsequent years.	11
Decision No. 7.	12
The margin will be subject to annual review. In the event of significant	
unforeseen changes in the marketplace, ComReg will review the price control	
and may issue amendments.	12
Decision No. 8.	12
eircom's cost of capital will be used as the basis for calculating the discount	
factor in the discounted cash flow analysis.	12
Decision No. 9.	13
The margin squeeze test will be conducted on an individual, product by product	
basis.	13
Decision No. 10.	13
Commercial negotiation will be used to determine the price for wholesale	
products that do not have an eircom retail equivalent. If such negotiation fails	;
to determine a reasonable wholesale price, ComReg will intervene.	13
Decision No. 11.	14
ComReg will perform sensitivity analysis around rates of decrease of retail price	es
in doing its discounted cash flow evaluation of margins.	14
Decision No. 12.	14
ComReg will use a forward looking approach in assessing costs.	14
Decision No. 13.	14
eircom's costs should be used as the basis for establishing the costs of a	
similarly efficient operator.	14
Decision No. 14.	15
See section 4 – the decision instrument.	15
Decision No. 15.	16
See section 4 – the decision instrument.	16
Decision No. 16.	17
The prices for wholesale connection and port transfer shall be cost oriented.	
From 1 January 2006, the price for wholesale connection and port transfer sha	П
be €30. eircom shall not increase this charge without the prior approval of	
ComReg.	17
Decision No. 17.	17
See section 4 – the decision instrument.	17

Decision No. 18.	18
See section 4 – the decision instrument.	18
Decision No. 19.	18
See section 4 – the decision instrument.	19
Decision No. 20.	19
The current arrangements, where eircom offer two contractual alternatives - o requiring a commitment to a six month minimum term but no cessation fee, the other requiring no commitment to minimum term but requiring a cessation fee are reasonable and amending this would be inappropriate. Over time new products, more tailored to individual operators' requirements, will be develope and reasonable prices should be achievable through commercial negotiation. Should this not happen ComReg would intervene to ensure that prices are	ne e -
reasonable.	19
Decision No. 21.	19
The service establishment charge will be reviewed at the commencement of each new price control period. In addition, a change in price or terms and conditions for the service establishment charge or the cessation charge require prior approval from ComReg. In order to grant such approval, ComReg require that eircom submit a cost based justification.  Decision No. 22.	
The prices for bitstream connection services will not have specific obligations other than that access should be at reasonable prices.  Decision No. 23.  See section 4 – the decision instrument.	20 20 20

## Appendix B - Regulatory Impact Assessment

#### Consultation issue

In the consultation, ComReg examined the impact of elements of the retail minus price control on affected parties.

It was noted that the regulatory impact of imposing a retail minus price control had been examined and consulted on as part of the process of reviewing the WBA market, and was therefore outside the scope of this current consultation. The focus at this stage was on the relative impacts of the different choices relating to the implementation of the retail minus control.

ComReg identified the core regulatory issues as follows:

- 1. The control is in the form of an ex ante retail minus formula.
- 2. The margin squeeze test will be reassessed annually to ensure that the assumptions made in the DCF analysis are accurate, and that the coefficients of the formula are amended if necessary.
- 3. Each wholesale product will have an individual price control formula.
- 4. Where eircom wish to introduce a new retail product it will be required to submit a proposal to ComReg for an assessment of a margin test prior to notifying OAOs of the impending wholesale change.
- 5. Before implementing any wholesale price changes, including allowing eircom retail to avail of the product, eircom must notify OAOs 15 working days in advance. This notification should be restricted to those OAOs with bitstream contracts and should not be publicly available.

For each core regulatory issue, ComReg outlined alternative proposals, and examined their likely impact on eircom, OAOs and consumers.

ComReg believes that enhancing competition in the WBA market will be of benefit to all competitors. It was not ComReg's intention to infer that intervention in one market would be used to address problems identified in another. However, ComReg does believe that longer term benefits to do with lowering costs and enhancing innovation will best be achieved with a price control which does not limit incentives to invest in infrastructure

In relation to the 5 core regulatory implications identified, ComReg has concluded the following:

- 1. The control is in the form of an ex ante retail minus formula. The benefits to OAOs and consumers of using an ex ante approach justify the greater regulatory impact upon eircom compared to using an ex post approach for the reasons set out in consultation 05/67 and in Section 3.2.1 in the response to consultation 05/88.
- 2. The margin squeeze test will be reassessed annually to ensure that the assumptions made in the DCF analysis are accurate, and that the coefficients of the formula are amended if necessary. The benefits to OAOs and potentially eircom of reassessing the margin squeeze test annually to ensure that the assumptions made in the DCF analysis are accurate justifies the greater regulatory impact upon eircom compared to less frequent reassessments.
- 3. Each wholesale product will have an individual price control formula. The benefits to OAOs and consumers of each wholesale product having an individual price control formula justifies the greater regulatory impact upon eircom compared to the portfolio approach. This will promote competition for all products to the benefit of consumers.
- 4. Where eircom wishes to introduce a new retail product, it will be required to submit a proposal to ComReg for an assessment of a margin test prior to notifying OAOs of the impending wholesale change. The benefits to OAOs and consumers of assessing the compliance of new products in advance of wholesale notification justifies the greater regulatory impact upon eircom compared to a statement of compliance for new products with simultaneous wholesale notification. This will reduce the risk of foreclosure and promote competition to the benefit of consumers.
- 5. Before implementing any wholesale price changes, including allowing eircom retail to avail of the product, eircom must notify OAOs 15 working days in advance. This notification should be restricted to those OAOs with bitstream contracts and should not be publicly available. The benefits to OAOs and consumers of an advance notification period of 15 working days prior to the implementation of any wholesale price changes justifies the greater regulatory impact upon eircom compared to no notification period. ComReg believes that a period greater than 15 days would be disproportionate and burdensome on eircom and would negatively affect consumers by unnecessarily delaying innovation.

Appendix C – Consultation on retail minus wholesale price control for the WBA market – Document No 05/67



## **Consultation Paper**

# Consultation on retail minus wholesale price control for the WBA market

Document No:	05/67
Date:	19 August 2005

All responses to this consultation should be clearly marked: "Reference: Submission re ComReg 05/67" as indicated above, and sent by post, facsimile, e-mail or on-line at <a href="www.comreg.ie">www.comreg.ie</a> (current consultations), to arrive on or before 17h30 on 16 September 2005 to:

Ms. Sharon Ward Commission for Communications Regulation Irish Life Centre Abbey Street Freepost Dublin 1 Ireland

Ph: +353-1-8049600 Fax: +353-1-804 9680 Email: financeconsult@comreg.ie

Please note ComReg will publish all respondents submissions with the Response to this Consultation, subject to the provisions of ComReg's guidelines on the treatment of confidential information – ComReg 05/24

# **Contents**

1 Ex	ecutive Summary	2	
2 Int	troduction	4	
2.1 2.2 2.3 2.4	Purpose of the consultation	4 5	
3 AP	PROACH	8	
3.1 3.2 3.3 3.4 3.5 3.6	Scope of Review  Objective of Retail Minus: Counteracting Margin Squeezes  Illustrating the concept of Margin Squeeze  Incentives for a vertically integrated operator to engage in a Margin Squeeze  Principles of ex ante and ex post application of a Margin Squeeze test  Conclusion	8 9 10 12	
4 Ca	lculating the required margin	16	
4.1 4.2 4.3 4.4 4.5 4.6 4.7 4.8 4.9 4.10 4.11	RETAIL PRICES (REVENUES)	16 17 18 19 20 21 21 22 22	
4.12 4.13			
5 Im	plementation of the control		
5.2 5.2 5.2 5.2	ASSESSMENT OF COMPLIANCE  FORM OF RETAIL MINUS PRICE CONTROL MECHANISM  2.1 Price control on Connection fee  2.2 Price control on wholesale monthly rental  2.3 Retail Discount Schemes  2.4 Retail Promotions  2.5 Other Price Controls  PUBLICATION / NOTIFICATION TO THE WHOLESALE MARKET	27 28 28 30 30 31	
6 Re	gulatory Impact Assessment ('RIA')	33	
6.1	THE EFFECT OF KEY PROPOSALS FROM THE CONSULTATION ON THE INTERESTED PARTIES		
7 Su	bmitting Comments	37	
Appen	dix A – Legislation	38	
Appendix B – Consultation Questions			

# 1 Executive Summary

The Commission for Communications Regulation ('ComReg') is initiating this consultation as an amendment of obligations which were imposed on an interim basis following the market review of the wholesale broadband access (WBA) market in Ireland (the 'market'). As part of its analysis of the Market ComReg identified competition problems, in particular the possible leverage of market power by eircom Limited ('eircom') in the Market into the downstream prospectively competitive, retail market for broadband services, by way of a margin squeeze (also known as a price squeeze).

ComReg proposes to address the competition problems identified in the Market by applying a retail minus price control. The objective of this consultation is to propose an amendment to the existing interim price control framework.

ComReg's aim is to develop a methodology for calculating retail minus which will be transparent in its methodology and operation, and which will offer greater predictability to the market. ComReg notes that any margin set should neither distort incentives to invest in infrastructure, nor encourage inefficient market entry.

In this consultation, ComReg considers whether regulatory objectives would better be achieved by applying a retail minus control ex post or ex ante. Given the circumstances prevalent in the Market (as identified by ComReg in its analysis of the Market), ComReg is currently of the view that it would not yet be sufficient to rely on an ex post margin squeeze test in order for ComReg to achieve its objectives described in section 12 of the Communications Regulation Act 2002. In that regard, it is ComReg's view that if the Market were more mature and closer to effective competition, then an ex post test would be the appropriate way in which to test for a margin squeeze. An ex ante control would preclude potential margin squeeze, and would offer greater predictability for investment, and transparency in implementation and operation.

The key principles of the proposed control can be summarised as follows:

- the overall approach should be forward-looking, assessing economic costs and revenues over time, and building in judgements about the likely future value of variables
- eircom's costs and revenues should be used as the basis for establishing those of a similarly efficient operator, and these should be modified to take account of differences in scale, costs which would be borne by efficient new entrants but not by eircom, and of the evolution of prices
- a discounted cash flow (DCF) should be adopted, with a DCF analysis carried out over a 5 year period, and truncated with a terminal value
- the margin squeeze test should be applied on a product by product basis with a separate control for each wholesale and retail product pair

- the margin should be reviewed annually, unless circumstances in the market change significantly
- where eircom changes prices of existing products eircom will not be required to submit the proposed changes to ComReg prior to notifying the wholesale market.
- where eircom introduces new products it will be required to submit any proposals to ComReg in advance, and will be required to provide the information necessary to assess whether or not there is a potential margin squeeze.
- the wholesale market should be notified 15 working days in advance of any proposed change to wholesale prices.

ComReg welcomes comments on this consultation. The consultation period will run until 16 September 2005.

#### 2 Introduction

#### 2.1 Purpose of the consultation

ComReg is initiating this consultation to enable the public and interested parties to make submissions in respect of ComReg's proposals to amend, in accordance with Regulation 15 of the European Communities (Electronic Communications Networks and Services) (Access) Regulations 2003, certain of the obligations imposed on eircom in relation to the Market. In the course of the process leading up to the definition and analysis of the Market, ComReg consulted on a number of issues underpinning retail minus as a means of price control, and received preliminary comments from industry.

This consultation refers to wholesale bitstream products that are used as inputs by providers of retail broadband services to offer broadband internet access services to consumers and businesses. Currently, eircom is the main provider of wholesale broadband access products, and provides these products and services service both to its own retail arm and to competing operators.

The proposals for the implementation of an amended retail minus price control mechanism are designed to replace the existing interim price control and to ensure that potential competition problems identified in the analysis of the Market, conducted by ComReg, continue to be adequately addressed. In particular, ComReg is concerned about the possible leverage of market power by eircom in the Market into the downstream prospectively competitive retail market for providers of retail broadband services. The amended retail minus scheme is intended to ensure that the potential competition problems in the Market continue to be addressed and also to facilitate the emergence of effective competition in the retail market for broadband services and prevent anti-competitive practices without compromising desirable investments at a network level.

# 2.2 Regulatory objectives under the Communications Regulation Act, 2002

Section 12 of the Communications Regulation Act, 2002 outlines the objectives of ComReg in exercising its functions. In relation to the provision of electronic communications networks, electronic communications services and associated facilities these objectives are to:

- promote competition
- contribute to the development of the internal market, and
- promote the interests of users within the European Union.

It is ComReg's view that these proposals are in line with the objectives set out in the Communications Regulation Act, 2002. The purpose of the proposals is to seek to promote competition amongst operators to ensure that end-users derive the maximum benefit in terms of price, choice and quality.

#### 2.3 Regulatory Framework

Four sets of Regulations,<sup>1</sup> which transpose into Irish law four European Community directives on electronic communications networks and services,<sup>2</sup> entered into force in Ireland on 25 July 2003. The final element of the European electronic communications regulatory package, the *Privacy and Electronic Communications Directive*, was transposed into Irish law on 6 November 2003.

The new communications regulatory framework requires that ComReg define relevant markets appropriate to national circumstances, in particular relevant geographic markets within its territory, in accordance with the market definition procedure outlined in the *Framework Regulations*<sup>3</sup>.

In its Recommendation on relevant product and service markets, the European Commission identified the market for wholesale broadband access as being a market which could be susceptible to ex ante regulation. ComReg initiated a national consultation on this market on 5 March 2004 (ComReg Document 04/25). ComReg received and analysed responses from industry, and responded to the consultation with a proposed draft measure on 29 July 2004 (ComReg Document 04/83). In Decision Notice 03/05 (Document Number 05/11r) ComReg designated eircom with Significant Market Power (SMP) in the market for wholesale broadband access and imposed a price control obligation.

ComReg's conclusion from its analysis of the Market was that a retail minus price control was necessary in the Market for two main reasons, namely:

<sup>3</sup> Framework Regulation 26.

<sup>&</sup>lt;sup>1</sup> Namely, the European Communities (Electronic Communications Networks and Services) (Framework) Regulations 2003 (S.I. No. 307 of 2003), ("the Framework Regulations"); the European Communities (Electronic Communications Networks and Services) (Authorisation) Regulations, 2003 (S.I. No. 306 of 2003), ("the Authorisation Regulations"); the European Communities (Electronic Communications Networks and Services) (Access) Regulations 2003 (S.I. No. 305 of 2003), ("the Access Regulations"); the European Communities (European Communications) (Universal Service and Users' Rights) Regulations 2003 (S.I. No. 308 of 2003), ("the Universal Service Regulations").

The new regulatory framework for electronic communications networks and services, comprising of Directive 2002/21/EC of the European Parliament and of the Council on a common regulatory framework for electronic communications networks and services, ("the Framework Directive"), OJ 2002 L 108/33, and four other Directives (collectively referred to as "the Specific Directives"), namely: Directive 2002/20/EC of the European Parliament and of the Council on the authorisation of electronic communications networks and services, ("the Authorisation Directive"), OJ 2002 L 108/21; Directive 2002/19/EC of the European Parliament and of the Council on access to, and interconnection of, electronic communications networks and services, ("the Access Directive"), OJ 2002 L 108/7; Directive 2002/22/EC of the European Parliament and of the Council on universal service and users' rights relating to electronic communications networks and services, ("the Universal Service Directive"), OJ 2002 L 108/51; and the Directive 2002/58/EC of the European Parliament and of the Council concerning the processing of personal data and the protection of privacy in the electronic communications sector, ("the Privacy and Electronic Communications Directive"), OJ 2002 L 201/37.

- firstly the application of a retail minus scheme would ensure that an operator identified as having SMP in the wholesale bitstream market could not leverage its significant market power in the Market into the prospectively competitive downstream retail market for providers of retail broadband services by applying *margin squeezes*.
- secondly, a suitably designed retail minus measure affords an opportunity for ComReg together with industry to establish an environment in which there would be greater transparency and more certainty.

ComReg has stated that a retail minus price control in the Market could be applied on either an ex ante or ex post basis. ComReg, in ComReg Document 05/11r<sup>4</sup> (following on from its definition of the Market and its designation of eircom as having SMP in the Market), imposed an interim ex ante retail minus price control on eircom. ComReg indicated that this would be an interim measure that would apply until a subsequent consultation on a replacement price control was completed and ComReg issued a new direction on the application of a retail minus control in the Market. In this paper ComReg elaborates on its thinking in relation to ex ante and ex post application of retail minus price control. As set out in section 3 ComReg is currently of the view that an ex ante retail minus price control is appropriate at this stage in the development of the Market in Ireland. ComReg notes that the same conclusion was reached in the UK in its review of the WBA market.<sup>5</sup>

ComReg noted in ComReg Document 04/83<sup>6</sup> that the implementation of retail minus was complex, and that a number of issues would need to be resolved and as a result imposed an interim price control. In this consultation, ComReg addresses these issues in further detail and at the conclusion of this consultation process ComReg will impose (following notification to the European Commission) a price control as envisaged in the WBA market review.

#### 2.4 Structure of Consultation Document

The remainder of this consultation document is structured as follows:

- Section 3 outlines ComReg's approach to retail minus.
- Section 4 examines the treatment of costs, revenues and profitability.
- Section 5 discusses the implementation of the control
- Section 6 outlines the regulatory impact assessment

6

<sup>&</sup>lt;sup>4</sup> Document No 05/11r; published on 17/02/05 - Market Analysis - Wholesale Broadband Access (Decision Notice - Designation of SMP and Related Remedies)

<sup>&</sup>lt;sup>5</sup> Wholesale Broadband Access Market: Explanatory Statement and Notification, Oftel, 16 December 2003.

<sup>&</sup>lt;sup>6</sup> Document Number 04/83 - Market Analysis: Wholesale Broadband Access (Response to Consultation Document 04/25 and Draft Decision)

- Section 7 provides details with regard to the submission of comments
- Annex A provides the legislative basis
- Annex B contains a list of the consultation questions

#### 3 APPROACH

#### 3.1 Scope of review

As part of the process of defining and analysing the Market and identifying appropriate remedies to impose on eircom, ComReg consulted on a range of broad principles applicable to retail minus.<sup>7</sup> This consultation invites comment on ComReg's current position with regard to the application of a retail minus measure that would encourage competition in the Market and optimise the incentives to all operators to invest. ComReg believes that the measure adopted must meet the criterion of transparency in its implementation, and should offer enhanced certainty for all operators.

This consultation exercise is intended to explain to interested parties ComReg's preferred methodology for the application of a retail minus scheme in the future. This section sets out ComReg's proposals for how a margin squeeze should be assessed. This will be dealt with in a draft Direction to be published having considered responses to this consultation.

# 3.2 Objective of retail minus: counteracting margin squeezes

The primary objective of the wholesale price control is to ensure that an operator holding SMP in the Market does not leverage market power into the closely related downstream market for the provision of retail broadband services by engaging in what is known as a margin squeeze, or price squeeze. The terms are generally used interchangeably. During the market review ComReg considered a wholesale price control mandating either cost oriented prices or a retail minus control. ComReg concluded that either approach would achieve the regulatory objectives.

ComReg imposed the price control obligation using the retail minus approach as this is the less burdensome of the two options. Retail minus pricing also has the added characteristic that it does not seek to control the absolute level of prices but only the margin between retail and wholesale. ComReg is of the view that this control is the most appropriate for a fast growing market where demand profiles are uncertain and volume forecasts more difficult than for more established products.

The European Commission first raised the issue of potential anti-competitive problems in relation to margin squeezes in telecommunications markets in its Access Notice in 1998:

"A price squeeze could be demonstrated by showing that the dominant company's own downstream operations could not trade profitably on the basis of the upstream price charged to its competitors by the upstream operating arm of the dominant company.... In appropriate circumstances, a price squeeze could also be demonstrated by showing that the margin

<sup>&</sup>lt;sup>7</sup> In particular see paragraphs 6.114-184 in Response to Consultation and Consultation on Draft Decision Market Analysis: Wholesale Broadband Access (Response to Consultation Document 04/25 and Draft Decision) Document No: 04/83, 29 July 2004.

between the price charged to competitors on the downstream market (including the dominant company's own downstream operations, if any) for access and the price which the network operator charges in the downstream market is insufficient to allow a reasonably efficient service provider in the downstream market to obtain a normal profit (unless the dominant company can show that its downstream operation is exceptionally efficient)."8

A reasonably efficient service provider in the downstream market is understood to mean a firm that is as efficient as the downstream affiliate of the vertically integrated access provider. Hence, this interpretation of a margin squeeze states that a 'squeeze' occurs when a downstream firm as efficient as the vertically integrated firm's downstream affiliate is unable to achieve a normal profit because the margin between the retail price and wholesale charge is too small. ComReg proposes to apply the "similarly efficient operator" standard.

A margin squeeze applied by a vertically integrated operator having SMP in the WBA market is also a form of price discrimination, as noted in the European Regulators Group common position on the application of remedies:

"Price discrimination can be used by a vertically integrated undertaking with SMP on the wholesale market to raise its rivals' costs downstream and induce a margin squeeze. This is achieved by charging a higher price (which usually is above costs) to downstream competitors than implicitly charged to the own retail affiliate, i.e. discrimination between internal and external provision." <sup>10</sup>

ComReg notes that the attraction of a retail minus control is that it avoids the possibility of a margin squeeze in a transparent and predictable manner without requiring the regulator to take a view of the appropriate absolute wholesale price level.

#### 3.3 Illustrating the concept of margin squeeze

The main reason for a vertically integrated firm to engage in a margin squeeze is to gain market power in the downstream retail market. By raising the price of the wholesale input it sells to its downstream competitors, it can ensure they do not make a normal profit and, ultimately, exit the market. This will increase the vertically integrated firm's market power in the retail market, permitting it to charge a higher retail price and earn higher profits. Raising the price of the wholesale input not only puts pressure on firms currently in the retail market, but also acts as an entry deterrent to potential entrants, who realise that entry will not be profitable. The overall effect of a successful margin squeeze is to drive out current retail competitors and to deter new entry, thus ensuring no effective competition in the retail market.

-

<sup>&</sup>lt;sup>8</sup> EC Access Notice para 117-118, Notice on the application of the competition rules to access agreements in the telecommunications sector, OJ 1998 C 265/2.

9 Using this standard when applying retail minus ensures that firms that are less efficient than eircom are not encouraged to enter the market.

<sup>&</sup>lt;sup>10</sup> Page 35, ERG Common Position on the approach to Appropriate remedies in the new regulatory framework, ERG (03) 30rev1, April 2004.

The key aspect of assessing whether a margin squeeze is being applied is to consider the difference between the retail price in the down stream market and the wholesale price of the input supplied. First, assume the downstream competitor is as efficient in the downstream sector as the vertically integrated firm. If the vertically integrated firm raises the cost of the wholesale input to the point where the total costs of its competitor (consisting of the price of the wholesale input plus its downstream retail costs) exceed the market price, then the competitor will be unable to make a profit. The key issue to assess is the difference between the wholesale input's price and the price of the final good – the difference, or margin, between them is being squeezed by the vertically integrated firm.

The vertically integrated firm itself can earn profits, as its own input costs are less than the price it charges its downstream competitor. The net effect is to ensure the downstream competitor earns a loss despite being as efficient as the vertically integrated firm. This should mean that downstream rivals leave the market, while potential entrants to the downstream market are deterred from coming in. This consolidates the vertically integrated firm's position in the downstream market and allows it to charge higher retail prices. By limiting entry at the retail level, it may also help prevent future wholesale entry, as wholesale entry may be easier for firms that have established a strong retail presence.<sup>12</sup>

To summarise, the main criteria assessed when applying a margin squeeze test are as follows:

- There is a vertically integrated firm with SMP in the upstream market
- The margin between the retail price and the price charged for the wholesale element is insufficient to cover the vertically integrated firm's downstream costs (by implication therefore an equally efficient downstream firm would not be profitable)
- The vertically integrated firm does not incur a loss overall.

# 3.4 Incentives for a vertically integrated operator to engage in a margin squeeze

Where a vertically integrated operator has SMP in the upstream market the wholesale price of the upstream element is determined by a vertically integrated firm absent regulation. It is likely this firm would also have influence over the retail price in the final downstream market – this depends upon whether it currently has market power in the downstream market. Hence an unregulated vertically integrated firm with SMP in the upstream market can apply a margin squeeze by affecting the relative values of the wholesale and retail prices.

\_

<sup>&</sup>lt;sup>11</sup> Note that at this wholesale input price, the vertically integrated firm's downstream arm also would not make a normal profit.

<sup>&</sup>lt;sup>12</sup> This is an illustration of the "ladder of investment" theory.

A vertically integrated firm could practise a margin squeeze by setting a relatively high price for the upstream element and/or a relatively low retail price, such that it leaves itself insufficient margin to cover its own downstream costs.

At this stage it is worth asking what incentive a vertically integrated firm has to apply a margin squeeze. There are two main incentives:

- 1. A successful margin squeeze, by ensuring downstream competitors will not make a normal profit, will force out such competitors as well as ensuring potential new entrants do not come into the downstream market. This will enable the vertically integrated firm to strengthen its position in the downstream market and charge a higher price than would result from a competitive market to the detriment of consumers.
- 2. High barriers to entry in the upstream market may be lowered by successful entry into the downstream market. This gives a further incentive to the vertically integrated firm: by ensuring a dominant position in the downstream market it not only obtains greater profits in that market, but helps preserve its dominant position in the upstream market, to the further detriment of endusers.<sup>13</sup>

One critique of margin squeezes suggests that a vertically integrated firm would not find it rational to apply them. <sup>14</sup> If a competing downstream firm is known by the vertically integrated firm to be more efficient at retailing a service, the vertically integrated firm would do better by selling the upstream input at a price that enables it to appropriate profits from this more efficient downstream firm. In this setting the vertically integrated firm's downstream costs exceed the costs of the entrant. If the vertically integrated firm applies a margin squeeze to exclude the entrants, it will earn lower profits than it would have if it had sold the input to the efficient downstream firm. Therefore, it follows that the latter 'accommodating strategy' is superior. However, this argument is incomplete, as a successful margin squeeze will give the vertically integrated firm market power in the downstream market thus allowing it to raise the level of the retail price in which case a margin squeeze may well be more attractive than accommodation. And it ignores the second incentive, where a benefit from margin squeezes may be preventing the entry of rivals into the wholesale market.

Overall, ComReg analysis shows that a vertically integrated firm with SMP in the Market and which offers downstream retail broadband services faces incentives to apply a margin squeeze. ComReg notes that the more efficient the market entrant, the less incentive there would be to squeeze the margin, but that it may still occur. ComReg therefore proposes to apply an appropriate test to ensure that a margin squeeze is not applied in the Market.

-

<sup>&</sup>lt;sup>13</sup> This is related to the "ladder of investment" theory.

<sup>&</sup>lt;sup>14</sup> This is known as the Chicago critique in the economics literature.

# 3.5 Principles of ex ante and ex post application of a margin squeeze test

The application of retail minus pricing obligations by National Regulatory Authorities is therefore predicated on ensuring fair competition and the avoidance of margin squeezes. Competition and Regulatory Authorities typically assess the value of margin between wholesale and retail prices and compare this value against downstream costs incurred in providing the retail service. This test is known as an *imputation test*. If the downstream retail costs exceed the margin between retail and wholesale prices it is presumed that a margin squeeze is being applied, as an efficient downstream firm could not enjoy a normal profit.

ComReg is seeking to apply retail minus in this market consistent with the objectives of promoting predictability and enhanced transparency. The control that will be applied by ComReg could apply the imputation test in either an ex ante or ex post manner. In ComReg document 04/83, which was published as part of the process of defining and analysing the Market and identifying appropriate remedies to impose on eircom, it was stated that the question of an ex ante versus ex post application of a retail minus price control would be a key feature of this consultation.

Before addressing this question in further detail it is instructive to look at the experiences to date of competition and regulatory cases where the retail minus methodology has featured.

There are several examples of the European Commission's legal approach to margin squeeze<sup>16</sup> via the application of Article 82 of the EU Treaty. Some commentators<sup>17</sup> have noted that Article 82 has been used not just to address perceived anticompetitive behaviour, but to achieve broader liberalisation objectives. For the purposes of this consultation, it can be noted that cases have included a focus on OAOs, and a focus on new market entrants. This means that in some cases, showing that a rival would (or would not) be excluded from the market was decisive in finding an unlawful margin squeeze. In other cases (notably Deutsche Telekom) the focus was on the potential exclusion of new market entrants.

ComReg has noted that a margin squeeze test can be carried out ex ante, typically before the launch of a new product or service, or ex post, where the test is applied

\_

<sup>&</sup>lt;sup>15</sup> Such tests have been applied in an abuse of dominance case involving BskyB in the United Kingdom (see Office of Fair Trading, 2002, BskyB: The Outcome of the OFT's Competition Act Investigation, December) and in the Deutsche Telekom case before the European Commission OJ 2003 L 263/9.

<sup>&</sup>lt;sup>16</sup> See for example National Carbonising (Case 107-75 R, National Carbonising V. Commission, [1975] ECR 1193), Napier Brown/British Sugar ([1988] O.J.L 284/41), IPS vs Commission (Case T-5/97, [2000] ECR II-3755) and most recently Deutsche Telekom ([2003] O.J. L 263/9).

<sup>&</sup>lt;sup>17</sup> See for instance John Kallaugher "The "Margin Squeeze" under Article 82: Searching for Limiting Principles" presented at the conference organised by the Global Competition Law Centre (GCLC) in association with British Telecommunications plc, BT Centre, London, 10 December 2004.

usually following a complaint. The table below summarises the advantages and disadvantages of each approach when applied to retail minus.

Application of retail minus:	Pros	Cons
Ex ante	Can provide greater predictability for investment  Transparency in implementation and operation  Precludes potential margin squeeze	Places greater burden on regulated firm as it needs to ensure compliance  Potentially greater burden on regulator  Could result in delays in product releases due to compliance requirements  May reduce price or product innovation
Ex post	Provides for much greater flexibility  Less burdensome for regulated firm and for regulator	Less predictability  Less transparent  Late detection of a price squeeze could result in vertically integrated firm with market power upstream successfully deterring entry e.g. by acquiring a substantial share of new markets, which could reinforce its market power due to tipping

ComReg's analysis in the WBA market review concluded that the market is not effectively competitive, and that regulatory intervention was required to address this. Thus, it is unlikely that ex post regulation would be sufficient to ensure that eircom does not leverage its SMP in the wholesale market into the potentially competitive downstream market. This is because ex post measures generally address the abuse of a dominant position, rather than the holding of a dominant position. Specifically in this market, this means that by the time problems associated with margin squeeze were recognised and reported, and an ex post test was carried out, market share could already be secured by a vertically integrated firm, and alternative market entry prevented. In determining that a market should be subject to ex ante regulation, the

European Commission in its Recommendation on Relevant Markets<sup>18</sup> has already concluded that Competition Law is insufficient to address the competition failures identified

In considering the relative merits of ex ante and ex post application, ComReg returns to its overall regulatory objectives. Given the current situation of the broadband market in Ireland, ComReg believes that it would not yet be sufficient to rely on an ex post margin squeeze test to control market power. Were the market more mature and closer to effective competition, then an ex post test would be the appropriate way in which to test for a margin squeeze.

An ex ante approach would offer greater transparency in the market, because any proposed changes to price would be assessed for potential margin squeeze before they came into effect, using a method and a process which had already been agreed with industry. This would effectively preclude the possibility of margin squeeze. The benefit for the regulated firm is that it would know what margin squeeze test would be applied and how, and would therefore be able to ensure compliance. The benefit for OAOs would be that potential margin squeeze would be excluded, and would be seen to be excluded.

A second benefit of an ex ante approach is that it would offer greater predictability in the market. By agreeing a retail minus price control which would run for a specified period, all operators would be able to plan their product offerings and business strategies with more secure financial information

An ex ante retail minus approach would work in principle as follows. ComReg would use information about the downstream costs, possibly those associated with the vertically integrated firm, to determine the margin that would be required between revenues (prices) and the upstream costs. The vertically integrated firm would be free to choose its retail price but the maximum wholesale price would be determined by reference to a retail minus formula such that a margin squeeze would be avoided.

It is ComReg's view that ex ante measures are required to reduce the level of market power in the WBA market, and to ensure that competition becomes established. ComReg notes that the principles of transparency and predictability are paramount, and that these principles are best achieved by implementing an ex ante approach.

#### 3.6 Conclusion

This section has detailed ComReg's approach to retail minus, the conclusions of which can be summarised as follows:

,

<sup>&</sup>lt;sup>18</sup> Official Journal of the European Union, Commission Recommendation of 11 February 2003(2003/311/EC).

- The objective of implementing a retail minus price control is to ensure that an operator with SMP in the WBA market does not leverage market power into the downstream ISP market via a margin squeeze
- The application of a retail minus mechanism should be transparent
- The application of retail minus should offer all operators in the market greater predictability in terms of their investment and planning
- While a margin squeeze test can be carried out on an ex ante or ex post basis, ComReg believes that in the current market situation, regulatory objectives will be best achieved by applying an ex ante margin rule
- Q. 1. Do you agree with the application by ComReg of an ex ante retail minus test?

# 4 Calculating the required margin

#### 4.1 Introduction

ComReg proposes to adopt an ex ante form of retail minus. In order to determine a retail minus control that prevents the application of a margin squeeze ComReg must analyse three elements:

- The wholesale price
- The retail price
- The downstream costs of the vertically integrated firm (adjusted to take account of a similarly efficient operator).

In this section the calculation of each of these is discussed in greater detail, and the rationale for their adoption is also discussed.

Before addressing the main components of the ex ante retail minus control, a discussion of what constitutes a similarly efficient operator and the role of dynamic issues in the application of a margin squeeze test is necessary.

#### 4.2 Similarly efficient operator

ComReg proposes to use as the benchmark the concept of a 'similarly efficient operator' i.e. one which shares the same cost function as eircom's own downstream businesses but which does not yet necessarily enjoy the same economies of scale and scope as eircom's overall business currently does.

ComReg considers that the appropriate conceptual approach for the margin squeeze analysis in this situation would be to establish a margin which would allow a similarly efficient operator to enter the market today; to incur the relevant start-up costs, initial losses etc and still expect to be able to recover their costs over a reasonable period of time and to compete effectively with eircom in the broadband services market going forward. The adjustments which ComReg may make to take account of costs which are unavoidable for competitors are discussed below.

ComReg recognises that in making adjustments to costs and applying the margin squeeze framework to a similarly efficient operator, it needs to avoid promoting inefficient entry. In the longer term ComReg expects that local loop unbundling (LLU) and direct connections are likely to provide the more important basis for the promotion of competition in broadband, and is therefore concerned not to set a margin which could distort incentives to invest in infrastructure, or encourage inefficient entry.

In making the necessary judgements, ComReg will therefore take account of the effects on infrastructure investment.

# Q. 2. Do you agree that a "similarly efficient operator" constitutes an appropriate benchmark for a market squeeze analysis?

### 4.3 Dynamic issues

In a previous section dynamic considerations were raised in relation to the incentives to practice a price squeeze. In this section dynamic issues related to the application of the test proposed by ComReg and the measurement of variables used in the test is addressed.

In a relatively new market such as broadband, unit costs are likely to change as the number of subscribers increases. ComReg believes that due to economies of scale and scope, over time unit costs should decline in real terms in this market. It is also likely to be the case that as firms acquire a better understanding of the market, unit costs will fall to reflect this (what is sometimes termed learning-by-doing).<sup>19</sup>

Prices may also vary over time. It is often the case, following the introduction of new products, that firms discount prices in an attempt to foster demand, partly to encourage take-up and partly to enable lower unit costs.

In applying a margin squeeze test in this market ComReg could rely on historical or current data. The benefit of doing this is certainty: the data are known. The downside of doing this is the possibility that recent data are a poor indicator for the values of variables in the future. Where such data are employed they might suggest that the ex ante retail minus test would require that the vertically integrated firm's downstream unit cost is less than or equal to the retail price minus the wholesale price. This margin could be large and may overstate current and future downstream costs and as a result would impose an unfair burden on the regulated firm. Alternatively, by using the costs of the regulated firm the value of the vertically integrated firm's downstream unit cost may be low (as this firm may already enjoy substantial economies of scale due to first mover advantages). In this case the margin could be too small and would impose an unfair burden on entrants.

Another option would be for ComReg to rely on current data and judgements about the values of variables in the future. The benefit of doing this is that the effects of scale and learning economies may be factored in, along with any effects associated with introductory offers. The costs of adopting this approach would derive from errors in the judgments formed.

As the broadband products sold in this market have a customer life-cycle of many months (typically more than twelve and possibly up to forty eight), to ensure that the setting of margins adequately performs in an ex ante fashion the margin squeeze test, it will be necessary to form judgments about the value of future costs and prices (or revenues). In addition the prevalence of scale and learning economies also favours the application of current data and judgements about the values of the relevant variables.

-

<sup>&</sup>lt;sup>19</sup> See Kenneth J. Arrow (1962) "The economic implication of learning by doing", Review of Economic Studies, vol. 29, pp. 155-73.

ComReg believes that it is appropriate in this market to apply retail minus based on current data and judgements about future variables. However, ComReg notes that this approach may impact on the key objective of predictability.

Q. 3. Do you agree that in this market it is more appropriate to adopt an approach which can consider future costs and revenues rather than relying on historical costs and revenues?

### 4.4 Discounted cash flow analysis

By taking account of the market dynamics and adopting a forward looking approach to assessing costs and revenues (prices), ComReg believes that a discounted cash flow (DCF) method can be adopted to assess appropriate margins. A DCF approach is often used to assess business investment decisions. It uses the future value of cash flow projections and discounts them to arrive at a present value. An investment is worthwhile if the sum of the DCFs is non negative.

To apply retail minus ex ante using DCF requires an assessment of margins over time, which may be set as time varying or time invariant. Having established an appropriate margin between retail and wholesale prices, that produces non negative returns from the DCF analysis, an ex ante retail minus control is formulated that will preclude the application of a margin squeeze.

The application of the DCF method would be undertaken by ComReg by assessing the profitability of a similarly efficient operator to eircom, using data from eircom but making appropriate adjustments to account for the evolution of future prices and costs specific to a new entrant. The DCF approach will be employed in such a way so that the similarly efficient operator obtains a non-negative return over some defined period. Having ensured this outcome, the model will inform ComReg of the appropriate margins.

Applying a DCF method using forward looking assessments for the relevant variables requires judgments to be made about costs and revenues, as well as about the appropriate cost of capital.

A DCF approach seems to be the most appropriate tool in the circumstances and has the virtue that it is widely understood and used in business planning for analysing capital investment projects where the returns are realised over a number of time periods.

ComReg is proposing to apply DCF analysis by considering the position of an entrant today. In applying this approach, ComReg proposes to use eircom's costs – both historic and forecast data – as a data source. However, ComReg proposes to apply adjustments to these data, as discussed in more detail below.

A complication of this approach arises in terms of how best to determine the appropriate margin. In the following sections both the treatment of variables over time and the setting of the margin is discussed.

# Q. 4. Do you agree that a DCF method should be used to assess margins?

#### 4.5 What time horizon?

In assessing the profitability of a similarly efficient operator, there are a number of approaches for dealing with assessing profitability over time. For example, the profitability of the key current investments would imply using the economic life of those assets. Alternatively a very long-run approach which would consider profitability over the whole lifetime of the business or at least over multiple investment cycles could be employed.

Inevitably there is a trade-off between reliability and the horizon chosen. The longer the horizon, the more demanding the forecasting exercise and the more unreliable the forecasts – particularly as it would require the formation of judgments about future technologies.

Given the considerable difficulties associated with taking a very-long run approach with services which are still developing, as is the case here, ComReg proposes to adopt an approach used by other NRAs which focuses on the nature of the underlying investments and an appropriate period for the recovery of those investments.

ComReg considers that an appropriate timescale would be one under which the current investments were profitable and where the firm did not rely on earning supernormal profits on future investments. On this basis, ComReg considers that it would be more reasonable to specify a time period that was related to the economic life of the underlying assets rather than the very long-run.

ComReg could choose a shorter time period but this would run the risk that it would not allow sufficient time for an entrant to recover early year losses, resulting in an inflated margin in order to pass the margin squeeze test, and thus encouraging inefficient entry. Conversely, a longer time period places reliance on forecasts for an extended period into the future, increasing the risk of forecast error and the likelihood of generating unreliable results.

Mindful of the need to balance these two concerns, ComReg considers that it is reasonable to carry out the DCF analysis over a five year period and truncate the analysis at that point with the inclusion of terminal values. ComReg believes that this provides sufficient time for an entrant to recover any legitimate initial losses whilst taking account of issues around reliability of forecasts in that the analysis uses both data from eircom's actual experience as well as eircom's updated forecast data.

Q. 5. ComReg proposes to carry out the DCF analysis over five years. Do respondents agree?

#### 4.6 Terminal value

ComReg proposes to apply the DCF analysis for five years and truncate at that point instead of trying to project the cash flows to infinity. The standard approach for dealing with such truncation is to consider the terminal value associated with the activity at the time of truncation to reflect the fact that the business will continue beyond this time and that assets have an on-going economic value (or earning power).

The approach which ComReg proposes will allocate the costs of assets to be recovered (including cost of capital) between two periods of time: the explicitly-modelled period of five years from launch up to the point of truncation and the years beyond the point of truncation. The smaller the terminal value is, the greater the amount of cost to be recovered in the explicitly modelled period.

Q. 6. Do you agree that it is appropriate to apply a terminal value? If so, what do you believe would be an appropriate method for calculating this value?

#### 4.7 How often should the margin be reassessed

ComReg has considered whether the retail minus control should be invariant over a fixed period, and if so, how long. Further, ComReg has considered the circumstances which would trigger a review.

In considering the period over which the control should apply, ComReg has examined a range of options. One option would be to review the appropriate margin every time there is a retail price change. While this would ensure that the control was highly responsive to the market, and that any changes in product or service offering could be accommodated, ComReg considers that this approach could not provide the level of predictability which is required by all operators. As the provision of greater certainty as regards the margin is a key purpose of proposing this measure, ComReg does not believe that a margin review triggered each time there is a retail price change would achieve the regulatory objectives.

Another option would be to fix the margin for a multi-year period, perhaps by linking the margin review with the next market analysis of the WBA market. The advantage of this approach is that at this time ComReg will revisit the issues of market definition and its assessment of market power. ComReg will also have the opportunity to assess whether the remedies imposed as a result of the first market review have been effective and proportionate in addressing SMP in the WBA market, and to examine the level and nature of market entry.

However, while this approach would have appeal in a market which was mature and fairly predictable, ComReg is concerned that a fixed margin applied over a multi-year period may not be the best option in a dynamic and new market. ComReg notes that other regulatory interventions, such as Local Loop Unbundling (LLU) are likely to impact on the WBA market over the next few years. While a longer term fixed margin would deliver greater certainty for operators, the lack of responsiveness to change would be a major disadvantage.

In order to balance the provision of greater certainty with the need for flexibility, ComReg therefore proposes that the retail minus control should be subject to annual review to ensure that it is providing the appropriate margin. ComReg notes that an overall analysis of the principles underpinning the setting of a retail margin will form part of the next market review of the WBA market, and that this process may take precedence in the review timetable. Further, ComReg would wish to retain the flexibility to initiate a review of the margin should circumstances in the market change significantly.

Q. 7. Do you agree that the margin should be subject to annual review, unless triggered by significant change in the market?

### 4.8 What discount rate should be applied?

In the DCF exercise ComReg will need to apply a discount rate. ComReg proposes that it would be appropriate to use a measure of the weighted average cost of capital (WACC). This may be applied on a pre-tax or post-tax basis. Eircom currently applies a pre tax WACC of 11.5% in its separated accounts.

ComReg therefore proposes to use eircom's cost of capital as a basis for calculating the relevant discount factor in the DCF analysis.

Q. 8. Do you agree that eircom's cost of capital should be used as the basis for calculating the discount factor in the DCF analysis?

#### 4.9 Individual services or a portfolio?

21

It is possible to carry out a margin squeeze analysis at the whole business level, i.e. across a portfolio of related products and also at the level of individual products and services.

eircom currently provides a number of retail ADSL products. Each of these retail products has a corresponding wholesale bitstream product. Of course entrants are also likely to offer a portfolio of services and could choose to compete with eircom across a similar product portfolio. However, it seems premature for ComReg to make judgments about the nature of entry. ComReg's margin squeeze testing to date has been based on the price differential between each of the retail products and its equivalent wholesale product.

ComReg recognises that the main benefit of conducting a margin squeeze test across a portfolio of products is that it offers the operator greater flexibility in designing its offerings, and consequently could lead to greater innovation in the market. However, a portfolio approach is necessarily less transparent, and risks an operator squeezing the margin on one product while remaining within the overall balance of the portfolio.

ComReg proposes to apply the ex ante retail minus framework on a product by product basis rather than on a portfolio basis. Ensuring that there is no margin squeeze on any individual product should avoid an entrant having to replicate eircom's product mix in order to be viable. Conducting the margin squeeze at the level of the individual product would allow OAOs to enter the market and target specific retail segments. If the analysis was performed on a portfolio basis eircom could potentially cross subsidise between retail products such that addressing certain segments in isolation would be unattractive.

Q. 9. Do you agree that the margin squeeze test should be conducted on the basis of individual products?

# 4.10 Wholesale products without retail equivalents

In addition to the wholesale products that eircom uses to support its own retail products and services it also offers two wholesale products which its own retail arm does not avail of. ComReg document 05/11r imposes upon eircom an obligation to meet reasonable requests for wholesale bitstream products and therefore it is possible that additional products will be developed at the wholesale level which eircom will not use at the retail level. As eircom does not offer retail equivalents of these wholesale products, the application of a retail minus price control would clearly be impossible.

ComReg proposes that wholesale products, without an equivalent retail product offered by eircom, should be excluded from the margin squeeze analysis. In order to avoid distorting the market, ComReg believes that these products should be priced in a manner consistent with the nearest equivalent wholesale product (which eircom does avail of at the retail level) suitably adjusted to reflect differences in functional characteristics.

Q. 10. Do you believe that this is an appropriate way to treat wholesale products which eircom does not avail of at the retail level?

# 4.11 Retail Prices (Revenues)

A key element of the DCF analysis in an ex ante margin squeeze test is the prices of the downstream services, which generate the revenues. As the DCF exercise is forward-looking, ComReg will need to assess the likely path of costs and revenues

for a similarly efficient operator to eircom. It is important when doing this that ComReg does not include the effect on revenues of the application of market power.

If a similarly efficient operator were to generate positive returns in the future due in part to its ability to exercise market power, it would be incorrect to include this in the margin calculations. Therefore the assessment of revenues will be based on the presumption that the market will become effectively competitive in future years.

ComReg starts from the assumption that in a competitive market, a large fall in costs would generally be expected to lead to lower prices, though this may not occur immediately. However, it seems unreasonable to suppose that firms could maintain prices above costs indefinitely in the face of competition.

ComReg believes that real unit costs in this market are likely to fall, rather than increase, over time, and that competition among providers would reduce prices. ComReg proposes to conduct sensitivity analysis when evaluating margins to assess the impact of competition on prices.

Q. 11. Do you agree that ComReg should perform sensitivity analysis around rates of decrease of retail prices in doing its DCF evaluation of margins?

# 4.12 How should we approach costs?

ComReg has considered whether it is more appropriate to use historical costs or projected costs when assessing the cost of offering service. Generally, historical costs would look at actual costs over a specified period. Standard accounting techniques would be used to analyse costs and assess profits. This approach would normally be used where there is reliable time series data on which to base projections, and where the market is reasonably stable, so that past performance can provide a reasonable guide to future performance.

The WBA market in Ireland does not exhibit the characteristics which would tend towards adopting a historical costs approach. The market is relatively new, subject to potentially rapid change, and is not likely to develop stability in the short to medium term. An alternative approach to the accounting-based focus on historic costs is to adopt a forward looking approach which analyses economic costs and revenues over time. This approach would be standard practice in business planning, and ComReg considers that, given the characteristics of the WBA market in Ireland, the most appropriate approach is to utilise a forward looking assessment of economic costs and revenues.

Q. 12. Do you agree that it is more appropriate to adopt a forward-looking approach to the assessment of costs?

#### 4.13 Whose costs?

The key aim of an ex ante retail minus rule to apply a margin squeeze test is to ensure that similarly efficient operators to eircom are able to compete and build a profitable business.

ComReg notes that a competition analysis approach would use eircom's costs in calculating the margin. The analysis would assess the level of downstream profitability which eircom could expect assuming it paid the same wholesale price as its competitors. This means that the calculation would be of the additional costs which eircom incurs in order to offer its retail products. These are incremental costs. While this approach has some appeal – notably that eircom's costs are known and reported – the disadvantage is that it cannot take into account costs which a new entrant would incur, but which eircom would not.

An alternative approach would be to start from the costs which would be borne by a new entrant to the market, and to build a bottom-up model of a new entrant's costs. While this may appear to be a more realistic measure of the actual experience of a new market entrant, ComReg believes that the cost assessment is more reliable if built on actual costs rather than hypothetical costs, and that eircom's cost profile provides the best starting point for the analysis. ComReg notes that this is consistent with practice of other NRAs, and with recommendations from the EC and ERG.

ComReg's proposed approach is to set a margin which would allow an entrant of *similar efficiency* to eircom to enter the market and compete on a forward looking basis. In ComReg's view, this means that costs necessarily incurred by a new entrant but not by eircom should be incorporated into the analysis. However, it must be emphasised that new entrant costs must be considered carefully to ensure that there is no reward for inefficient entry.

ComReg's proposal is therefore that eircom's costs are used as a starting point for the assessment, and that they are modified to take account of costs borne by similarly efficient new entrants.

Q. 13. Do you agree that eircom's costs should be used as the basis for establishing the costs of a similarly efficient operator?

# 5 Implementation of the control

ComReg is proposing to implement an ex ante retail minus price control. In this section, ComReg puts forward its proposals for the practical application of a retail minus control. There are three elements that ComReg will consult upon in this section. These are:

- the process for assessing compliance of wholesale prices with a retail minus control,
- the form of the retail minus price control; and
- the process of notifying OAOs of wholesale price changes.

In the sections below ComReg proposes a number of new time lines associated with the introduction of new products and amendments to existing products. These timelines relate solely to the wholesale price control obligation. eircom has a number of other obligations, including the obligation to provide access and the obligation to behave in a non discriminatory manner. Compliance with these obligations may have implications for the timings around the introduction of new products or amendments to existing products above those set out in the price control.

# 5.1 Assessment of compliance

In order to assess compliance with a retail minus price control, ComReg will clearly require visibility of both retail and wholesale prices, and as ComReg is proposing an ex-ante control, ComReg will need to have visibility of changes in retail or wholesale prices prior to their implementation. ComReg has not defined a retail broadband market that is suitable for ex-ante regulation and is in no way seeking the power to regulate retail prices. Any notification to ComReg by eircom of retail price changes in advance of their implementation is solely for the purpose of assessing compliance with the wholesale price control. Should ComReg find eircom to be non compliant with the wholesale price control, any regulatory interventions by ComReg would relate only to wholesale prices.

ComReg believes it would be undesirable if a lengthy approvals process were to delay innovation in the market place. It is in the best interests of all operators and consumers that product and service innovation be encouraged and ComReg will seek to ensure that any mechanisms put in place cause the minimum delay. ComReg recognises that eircom should be encouraged to innovate and must be allowed to respond to price changes by other operators in the retail market. There is a need to strike a balance between protecting competitors in the downstream market from the possible application of a margin squeeze and enabling the vertically integrated SMP operator to compete fairly in the downstream market.

ComReg has considered whether eircom should be required to submit proposed retail and wholesale price changes to ComReg in advance of their implementation, so that ComReg can assess whether the proposed wholesale prices are compliant with the wholesale price control. Where eircom wishes to implement a change in the retail price of one of its existing products (including amendments to promotions), ComReg

believes that the application of a retail minus formula is a sufficiently transparent mechanism that compliance with the obligation should be readily determined.

Therefore, ComReg proposes that in advance of eircom making a retail price change, it should be obliged to issue ComReg with a statement of compliance setting out the precise nature of the retail price change. The statement should demonstrate the corresponding adjustments which will be made to the wholesale prices such that compliance with the obligation is achieved. Once eircom has submitted such a statement to ComReg, it would be free to immediately notify wholesale operators of the wholesale price change as set out in Section 5.3 below without awaiting formal approval from ComReg.

In these circumstances, ComReg would only intervene if it found that the proposed changes in retail and wholesale prices were not in compliance with the price control obligation. If the proposed changes were not in compliance, ComReg would either initiate enforcement proceedings or direct amendments to the wholesale prices.

Q. 14. Do you agree that in the case of amendments to prices of existing products (including amendments to promotions) it is sufficient for eircom to issue ComReg with a statement of compliance and simultaneously notify wholesale prices to other operators?

In the case where eircom introduces new retail products, whether by changing the technical specifications of the product or by changing the basis of pricing<sup>20</sup> of the products, the assessment of compliance with the obligations will necessarily be less clear cut than is the case of amendments to existing products. There is a tension between the desire for transparency in the wholesale price control and the need to retain flexibility so as not stifle innovation. ComReg believes that it would be impractical to try and create a wholesale price control that accommodated all potential innovations at the retail level in advance of their introduction. Therefore, ComReg believes that in order to prevent a potential margin squeeze it would not be sufficient to adopt the same assessment process as proposed for amendments to existing products in the case of the introduction of new products.

In selecting the assessment procedure associated with the wholesale price control, ComReg should seek to ensure that it does not stifle innovation and cause unnecessary delays to the introduction of new products. ComReg has considered whether it would be appropriate to set a fixed time limit in which it would consider the compliance of new products. However, ComReg sees two disadvantages to this approach. On the one hand, as it is impossible to anticipate all future developments, it may not be possible to complete all assessments within a defined period. On the

\_

<sup>&</sup>lt;sup>20</sup> For instance, ComReg regarded the introduction of eircom's time based retail product (with its associated wholesale product) as a new product because although it had the same technical specification as the pre existing flat rate product, it was a significantly different tariff structure.

other hand, a fixed time period may in fact slow the introduction of new products where the assessment turns out to be relatively straightforward.

ComReg has also considered whether it would be appropriate to make no commitment about the time taken to complete the assessment of new products. ComReg believes that this would create unnecessary and undesirable uncertainty in the market.

Therefore, ComReg proposes that prior to the introduction of new products, eircom submits to ComReg a proposal setting out precise details of the new retail product and of the corresponding wholesale product<sup>21</sup>. Within a period of five working days ComReg will revert to eircom, either confirming that the proposal is in compliance with the obligation in which case eircom may proceed immediately to notify wholesale operators as set out in Section 5.3, or informing eircom that further analysis to determine compliance of the proposal is required and setting out indicative timelines for that further analysis. During the five days subsequent to eircom making a proposal to ComReg or during a period of further analysis, eircom will be prevented from introducing new wholesale products.

ComReg would approve the new wholesale prices once it is satisfied that the margin between retail and wholesale prices is sufficient to avoid a margin squeeze. Having determined the appropriate relationship between retail and wholesale prices for the new products, ComReg would update the retail price mechanism such that any subsequent amendments to the prices of these products would be assessed using the mechanism set out for existing products.

Q. 15. Do you believe that this proposal for price changes to new products is an appropriate mechanism for assessing the compliance of new products?

#### 5.2 Form of retail minus price control mechanism

The wholesale bitstream service is composed of a number of elements, such as connection fee, monthly rental, usage fees, cessation, backhaul and service establishment. ComReg is proposing that the retail minus price control should only apply to the connection and monthly rental elements of the service.

In Section 4 of this document, ComReg has acknowledged that there are fixed and variable elements to the cost stacks associated with the retail bitstream products. It has advocated a DCF analysis for a defined period in order to take relevant account of the intricacies of the cost stacks and the product life cycle. Prior to receiving the responses to this consultation and completing this analytical work, ComReg believes

Where eircom introduces a new retail product that is not adequately supported by the existing wholesale product set, eircom will be required to introduce a new wholesale product providing equivalent functionality and or tariff structure.

that it would be premature to consult on the appropriate size of the margin required for the wholesale price control.

However, ComReg believes that in the interests of transparency it is appropriate to consult on the form of the price control and to seek interested parties' views on whether ComReg proposals satisfy the requirements for transparency, predictability and flexibility while avoiding the possibility of margin squeeze during the life of this control.

#### 5.2.1 Price control on Connection fee

ComReg believes that the wholesale connection fee should always be cheaper than the standard (i.e. excluding temporary promotions or discount schemes) retail connection fee. Therefore, ComReg proposes that the wholesale price control for the connection fee should take the following form:

# Wholesale connection fee = Standard retail connection fee \* fixed percentage(<100%)

The treatment of temporary promotions and discount schemes are addressed later in section 5.2.3 and 5.2.4 below.

# Q. 16. Do you believe that this is an appropriate form of price control for the connection fee?

#### 5.2.2 Price control on wholesale monthly rental

The primary purpose of imposing a retail minus price control is to avoid a margin squeeze. ComReg hopes that the price of bitstream will continue to reduce over time as the take-up of the service increases. The wholesale price control should not be an unreasonable impediment to the reduction of retail prices over time, and should have sufficient flexibility to accommodate significant reductions in the retail price. On the other hand, it is vital that as retail prices decline, adequate margins are maintained for competing operators.

In light of these objectives, ComReg has considered three different mechanisms for the wholesale monthly rental price control.

The first option that ComReg considered is where the margin is expressed as an absolute fixed monetary value. This means that the gap between retail prices and wholesale prices is a fixed monetary value. This method was employed by ComReg during the interim period.

While ComReg believes that this was the appropriate control for an interim period, prior to the completion of this consultation and its associated analysis, it did not and does not believe that this would be an appropriate basis for a permanent price control. While a price control of this form has the advantage of preventing any

decrease in the margins available to OAOs (which was the objective in the interim period), it does not seem appropriate going forward in the context of reducing retail prices and increasing volumes. By expressing the retail minus as an absolute value, there is a significant risk that an artificial price floor could be created at the wholesale level if the price were to remain above cost. This could potentially stifle expansion in this market at the retail level. Alternatively, the wholesale price could be forced below cost which would place an unfair burden on eircom and would distort the investment incentives for this product as well as other wholesale products such as unbundled local loops.

The second option that ComReg considered is to express the margin as a fixed percentage of the retail price. This scenario has the advantage of increased flexibility in terms of product pricing. However, there is a risk that if the retail prices are dramatically reduced the corresponding margin, while maintained in percentage terms, will be reduced in absolute monetary terms to such a degree that there is a margin squeeze. ComReg's preliminary analysis of the costs associated with providing retail service in this market suggest that a significant number of unit costs are unlikely to reduce significantly as prices decline and volumes grow. In order to ensure that sufficient margin is maintained, if prices were to fall significantly, there is a risk that margins for the current prices would have to be set too high.

Finally, the third option considered by ComReg is a combination of a fixed monetary value and a fixed percentage. ComReg advocates this third option which lies between the two extremes examined above. This approach while avoiding a margin squeeze also takes into account the evolving nature of this market. ComReg acknowledges that prices are set to decrease over time as the corresponding volumes increase. Using a hybrid formula the absolute monetary value of the margin will decline but the percentage margin will increase. Such a hybrid approach also balances flexibility with transparency and predictability for all market players. ComReg's proposed control would therefore be:

# Wholesale rental price = (Retail rental price \* fixed percentage <100%) – fixed monetary value

ComReg proposes to set retail minus controls, in accordance with the formula above for all current products. These controls would be applied by reference to the standard retail rental i.e. excluding promotions and discounts. The treatment of promotions and discounts is discussed below in sections 5.2.3 and 5.2.4.

If eircom were to change any of the characteristics of either its retail or its wholesale products during the period of the control, it would be required to follow the approval and notification processes set out in Sections 5.1 and 5.3.

Where usage charges make up a significant portion of retail or wholesale prices (such as eircom's 'time' and 'kronos') ComReg proposes to convert usage charges into average monthly revenues (at retail and wholesale levels) and then apply retail test to the average recurring revenues.

Q. 17. Do you think that is an appropriate way to apply the retail minus formula? If you feel that there is a superior formula, please provide your reasoning.

#### 5.2.3 Retail Discount Schemes

eircom currently has one retail discount scheme – the Bulk Connection Discount Scheme. This provides discounts on the self install connection charge for bulk orders for DSL service. It is available for eircom's retail DSL products other than eircom Broadband Home Starter. There are a number of discount levels depending on the size of the order.

It is proposed that if eircom makes any changes to this discount scheme, during the relevant control period, then ComReg may require corresponding changes to wholesale bitstream prices. ComReg proposals would be carried out in accordance with the assessment process proposed for amendments to existing products.

Where eircom wishes to amend or introduce a new discount scheme, ComReg proposes that it would be subject to the same assessment process as outlined for new products.

Q. 18. Do you think that this is an appropriate way to treat discounts?

#### 5.2.4 Retail Promotions

eircom may from time to time introduce temporary promotions. In such circumstances, ComReg believes that controls on promotions are necessary in order to avoid the damaging effects of a margin squeeze on competition in the provision of retail DSL services. For many new customers the margin on the standard product will be irrelevant because they have taken up DSL under a promotion. This is particularly important for this market because of the extensive use of promotions to sell DSL products where eircom adopts a policy of near continuous, but varying, promotions.

Given the frequency with which new promotions are introduced at the retail level, ComReg believes that it would be undesirable to amend the wholesale prices every time a new retail promotion was introduced or withdrawn. Therefore, ComReg proposes a continuation of the current arrangement whereby each retail promotion is matched by a corresponding wholesale promotion, the value of which is computed by assessing the revenue foregone at the retail level between the promoted prices and the standard prices. This wholesale promotion takes the form of a one off rebate payment.

ComReg will treat any retail promotions and their corresponding wholesale promotions relating to existing products (including extension of deadline or early

withdrawal) in the same way as it would for the assessment of price changes of existing products as set out in Section 5.1.

Q. 19. Do you believe that this is an appropriate way to treat promotions? If not, please state clearly the reasons for your disagreement.

#### 5.2.5 Other Price Controls

There are currently a range of other bitstream related wholesale charges and services that are not so amenable to a retail minus control, mainly because they do not have direct retail equivalents. The approach to controlling prices for each of these additional services is set out below

#### 5.2.5.1 Service Establishment Charge

eircom currently sets a service establishment charge of €8,035 per Access Seeker and aims to recover eircom's costs in providing authorisation and training in bitstream IT systems for Access Seeker staff. ComReg proposes that eircom make no increase in this charge, or changes to other terms or conditions, without prior approval from ComReg. ComReg would expect to give approval where there is an appropriate cost-based justification.

ComReg proposes that this charge will be reviewed prior to the commencement of each new price control period.

### 5.2.5.2 Cessation Charge

eircom currently sets a wholesale cessation charge of €60 per port. This charge does not apply to Access Seekers who agree to amendment of their contracts to reflect a minimum 6 month term for each bitstream subscriber port. ComReg proposes that eircom should make no increase in this charge, or changes to other terms or conditions, without prior approval from ComReg. ComReg would expect to give approval where there is an appropriate cost-based justification.

#### 5.2.5.3 Bitstream Connection Service

ComReg stated in the previous consultation that it believed that the prices for the bitstream connection services should be based on FL-LRIC costs. However, ComReg has reconsidered this position in light of the constructive engagement between eircom and access seekers in developing new backhaul products. Therefore, ComReg proposes that it would be appropriate to forbear from imposing specific obligations on these services other than that access to these services should be at reasonable prices.

Q. 20. Do you believe that this is an appropriate way of treating the service establishment charge, cessation charge and bitstream connection service?

# 5.3 Publication / notification to the wholesale market

ComReg believes there is the potential for a retail minus mechanism to give eircom's retail operation an unfair advantage in the market, in that they know that any change they make to retail prices will automatically trigger a change in the wholesale price. However, if a competing operator were to make a price reduction it would continue to pay the same wholesale price, and hence would have lower margins. To prevent this effect from distorting competition, ComReg believes that it is essential that prior to eircom making any retail price change that would cause a change in the wholesale price, eircom must notify customers of its wholesale bitstream services in advance of the imminent wholesale price change.

The purpose of this requirement is to give OAOs time to consider their retail offer, given changes in the wholesale prices. ComReg has to consider how long the period between notification and implementation of the wholesale prices should be. In reaching an assessment of an appropriate period, ComReg has balanced the need to give OAOs sufficient time to respond to changes of wholesale prices with changes to retail prices with the desire to avoid creating unnecessary impediments on eircom's flexibility. While it is reasonable that in the current state of the market there should be some advance notification of wholesale changes, ComReg does not propose changing the notification period from the current obligation of 15 working days and does not believe that a longer period is required for OAOs to develop products and services.

ComReg notes that the prime objective of advance publication of proposed wholesale price changes is to enable the purchasers of wholesale bitstream to adjust their retail offer, should they so wish. It is therefore a matter to be addressed in the wholesale market, and the wholesale market comprises eircom and OAOs who purchase bitstream services. Given this, ComReg considers that it may not be necessary to widely publish proposed changes to wholesale prices, and that the objective may better be served by direct communication between eircom and OAOs.

ComReg proposes that OAOs should be notified by eircom of proposed changes to wholesale prices 15 working days before they come into effect. There should no longer be a requirement for general publication of proposed changes.

Q. 21. Do you agree with ComReg's proposal that changes to wholesale prices should be notified to the wholesale market 15 working days before coming into effect?

# 6 Regulatory Impact Assessment ('RIA')

In deciding on the precise implementation of the retail minus price under consultation, ComReg will take careful consideration of the impact of its proposals on affected parties.

Previously, ComReg has undertaken a rigorous and detailed review of the WBA market. In that review ComReg designated eircom with significant market power and identified a number of competition problems, including the serious risk of eircom foreclosing the potentially competitive retail market by leveraging its market power in the wholesale market into the retail market. In order to remedy this competition problem ComReg imposed an obligation of a wholesale price control. When imposing the price control ComReg considered whether the obligation should be to have cost oriented wholesale prices or whether wholesale prices should be set by reference to a retail minus formula that precluded the possibility of a margin squeeze.

ComReg concluded that either obligation would achieve the regulatory objective of preventing eircom from leveraging its market power into the retail market, and selected the retail minus approach as it was the less burdensome of the available remedies. The obligation for a wholesale price control based on retail minus was extensively consulted upon prior to its imposition, and ComReg conducted a RIA as part of the market review process. Therefore, the regulatory impact of imposing a retail minus control is beyond the scope of this RIA exercise.

The RIA conducted here examines the relative impacts of the different choices relating to the implementation of the retail minus control discussed above. For instance, it discusses the costs and benefits of an ex ante control compared to an ex post one. It does not consider every possible choice of how to implement the control, but, in keeping with the principle of proportionality, concentrates on the more important aspects. Tabulated below is a framework that ComReg proposes to use to evaluate the impact on affected parties of the various choices ComReg will have to make in implementing the retail minus wholesale price control. ComReg seeks respondents' views on the proposals set out below. The effects on eircom, other operators, and consumers are considered. ComReg does wish to stress the substantial benefits it believes will accrue to competitors from measure than protect and enhance competition, both in terms of the immediate effect on prices, but particularly on the long-term benefits of lowering costs and enhancing innovation.

# **6.1** The effect of key proposals from the consultation on the interested parties

33

1. The control is in the form of an ex ante retail minus formula.

Proposal / Alternative	eircom	OAOs	Consumers
Ex ante	Higher cost of compliance	Greater certainty and predictability in the	No margin squeeze should ensure

	Possible lack of flexibility could limit innovative pricing	warket  Very low risk of being margin squeezed	competition and lower prices in retail market  Increased certainty and predictability should encourage investment and dynamic competition with long-run benefits for consumers
Ex post	Lower cost of compliance  More flexibility on the timing of the introduce new products	Less transparency and predictability  Some risk of margin squeeze being successful due to delay in implementing ex post control  Less surety as to future margins	Increased risk of successful margin squeeze may lower competition

2. The margin squeeze test will be reassessed annually to ensure that the assumptions made in the DCF analysis are accurate, and that the coefficients of the formula are amended if necessary.

Proposal /	Eircom	OAOs	Consumers
Alternative			
Reassess more	Greater cost of	Less certainty in the	Limited effect
frequently than	compliance	market	
annually			
	More likely to reflect changing market and cost conditions	More likely to reflect changing market and cost conditions	
Reassess less	Lower cost of	More certainty in the	Limited effect
frequently than	compliance	market	
annually			
	More likely to reflect	Danger of not	
	changing market and	accurately reflecting	
	cost conditions	changing market	
		conditions	

3. Each wholesale product will have an individual price control formula

Proposal /	eircom	OAOs	Consumers
Alternative			
Individual product	Higher cost of compliance	Greater certainty in the market	More likely to ensure no successful margin
	May limit flexibility pricing	More likely to ensure key products not subject to margin squeeze	squeeze and hence ensure competition
Portfolio of products	Lower cost of compliance  Greater flexibility of pricing	Less certainty in the market  Possibility that OAO could be squeezed on one or two key products	May limit flexibility of supply responses to changing consumer demand

4. Where eircom wish to introduce a new retail product it will be required to submit a proposal to ComReg for an assessment of a margin test prior to notifying OAOs of the impending wholesale change.

Proposal /	eircom	OAOs	Consumers
Alternative			
Assessment of	Grater cost of	Degree of certainty,	Benefits of
compliance in	compliance	transparency and	competition
advance of		predictability	
wholesale	May slow down		
notification	introduction of new	Competition	
	wholesale products	problems remedied	
Statement of	Lower cost of	Greater uncertainty,	Less competition
compliance with	compliance	less transparency	
simultaneous		and less	Quicker introduction
wholesale	Increase speed of	predictability	of new products
notification	introduction of new		
	wholesale products	Possible disruption if new wholesale	
	Possible disruption if new wholesale products had to be amended / withdrawn	products had to be amended / withdrawn	

5. Before implementing any wholesale price changes, including allowing eircom retail to avail of the product, eircom must notify OAOs 15 working days in advance. This notification should be restricted to those OAOs with bitstream contracts and should not be publicly available.

Proposal /	eircom	OAOs	Consumers
Alternative			
Advance	Loss of automatic first	Greater certainty and	By ensuring equality
notification period	mover advantage	predictability	of treatment should
of 15 working days			ensure greater
to OAOs	No advance public speculation (as only current wholesale	Greater chance to react to competitive change in market	competition in retail market
	customers are obliged to be informed)		May limit price flexibility and responsiveness
No notification	Greater flexibility in	Can place at	May limit
period	pricing	competitive	competition
		disadvantage	

#### 7 Submitting Comments

All comments are welcome, however it would make the task of analysing responses easier if comments were referenced to the relevant question numbers from this document.

The consultation period will run from 19 August 2005 to 16 September 2005 during which the Commission welcomes written comments on any of the issues raised in this paper.

Having analysed and considered the comments received, ComReg will review the subject matter of the consultation and the submissions and publish a report in October 2005 on the consultation which will, inter alia summarise the responses to the consultation.

In order to promote further openness and transparency ComReg will publish all respondents' submissions to this consultation, subject to the provisions of ComReg's guidelines on the treatment of confidential information – ComReg 05/24

#### Please note

ComReg appreciates that many of the issues raised in this paper may require respondents to provide confidential information if their comments are to be meaningful.

As it is ComReg's policy to make all responses available on its web-site and for inspection generally, respondents to consultations are requested to clearly identify confidential material and place confidential material in a separate annex to their response

Such Information will be treated subject to the provisions of ComReg's guidelines on the treatment of confidential information – ComReg 05/24

#### Appendix A - Legislation

European Communities (Electronic Communications Networks and Services) (Framework) Regulations 2003 (S.I. No. 307 of 2003)

European Communities (Electronic Communications Networks and Services) (Access) Regulations 2003 (S.I. No. 305 of 2003)

Directive 2002/21/EC of the European Parliament and of the Council on a common regulatory framework for electronic communications networks and services

Directive 2002/19/EC of the European Parliament and of the Council on access to, and interconnection of, electronic communications networks and services

Communications Regulation Act 2002

#### Appendix B - Consultation Questions

All Consultation Questions must be listed here. This can simply be done by positioning below and selecting F9 to update after you have completed your document. In order for questions to be automatically listed here they must be style Q.1. ComReg Question.

#### **List of Questions**

Q. 1. test?	Do you agree with the application by ComReg of an ex ante retail minus $15$
Q. 2. appropr	Do you agree that a "similarly efficient operator" constitutes an iate benchmark for a market squeeze analysis?
	Do you agree that in this market it is more appropriate to adopt an the character of the control
Q. 4.	Do you agree that a DCF method should be used to assess margins?19
Q. 5. respond	ComReg proposes to carry out the DCF analysis over five years. Do lents agree?
Q. 6. do you	Do you agree that it is appropriate to apply a terminal value? If so, what believe would be an appropriate method for calculating this value? 20
	Do you agree that the margin should be subject to annual review, unlessed by significant change in the market?21
	Do you agree that eircom's cost of capital should be used as the basis ulating the discount factor in the DCF analysis?21
	Do you agree that the margin squeeze test should be conducted on the individual products?22
	Do you believe that this is an appropriate way to treat wholesale s which eircom does not avail of at the retail level?
	Do you agree that ComReg should perform sensitivity analysis around decrease of retail prices in doing its DCF evaluation of margins?23
	Do you agree that it is more appropriate to adopt a forward-looking th to the assessment of costs?24
	Do you agree that eircom's costs should be used as the basis for hing the costs of a similarly efficient operator?24
product issue Co	Do you agree that in the case of amendments to prices of existing s (including amendments to promotions) it is sufficient for eircom to omReg with a statement of compliance and simultaneously notify ale prices to other operators?
	Do you believe that this proposal for price changes to new products is opriate mechanism for assessing the compliance of new products?27

Q. 16. Do you believe that this is an appropriate form of price control for the connection fee?	28
Q. 17. Do you think that is an appropriate way to apply the retail minus formula? If you feel that there is a superior formula, please provide your reasoning	30
Q. 18. Do you think that this is an appropriate way to treat discounts?	30
Q. 19. Do you believe that this is an appropriate way to treat promotions? If not, please state clearly the reasons for your disagreement	31
Q. 20. Do you believe that this is an appropriate way of treating the service establishment charge, cessation charge and bitstream connection service?	32
Q. 21. Do you agree with ComReg's proposal that changes to wholesale prices should be notified to the wholesale market 15 working days before coming into effect? 32	

Appendix D – Response to consultation and draft direction on retail minus wholesale price control for the WBA market – Document Number 05/88



#### Response to Consultation

# Response to consultation and draft direction on retail minus wholesale price control for the WBA market

Document No:	05/88
Date:	23 November 2005

#### **Contents**

1	Executive Summary	2
2	Introduction	4
	3.1 BACKGROUND	5 5
3	Decision Notice Issues	7
	3.1 CONTEXT	7 7
	<ul> <li>3.2.3 Dynamic issues – historic costs or forecasted future costs?</li></ul>	12 14 15
	<ul> <li>3.2.7 How often should the margin be reassessed?</li> <li>3.2.8 What discount rate should be applied?</li> <li>3.2.9 Individual services or a portfolio?</li> <li>3.2.10 Wholesale products without retail equivalents</li> </ul>	18 19
	3.2.11 Retail prices	22 23 24
	<ul> <li>3.2.14 Assessment of compliance - existing products</li> <li>3.2.15 New products</li> <li>3.2.16 Price control on connection fee</li> <li>3.2.17 Price control on wholesale monthly rental</li> </ul>	27 30
	3.2.18 Retail discount schemes	34 35
	3.2.21 Publication / notification to the wholesale market	
4	Regulatory Impact Assessment ('RIA')	43
A	appendix A – List of Draft Directions and Decisions	45
A	ppendix B – Draft Control Mechanism	47
A	appendix C - The Broadband 'New Entrant' Discounted Cash Flow (DCF) M	odel

#### 1 Executive Summary

The Commission for Communications Regulation ('ComReg') initiated this consultation as an amendment of obligations which were imposed on an interim basis following the market review of the wholesale broadband access (WBA) market in Ireland (the 'market'). As part of its analysis of the market ComReg identified competition problems, in particular the possible leverage of market power by eircom Limited ('eircom') in the upstream market into the downstream prospectively competitive, retail market for broadband services, by way of a margin squeeze (also known as a price squeeze).

ComReg proposed to address the competition problems identified in the market by applying a retail minus price control. The objective of the consultation was to develop a methodology for calculating retail minus which would be transparent in its methodology and operation, and which would offer greater predictability to the market. Applying this methodology, an amendment to the existing interim price control framework would be proposed<sup>1</sup>.

Responses to the consultation were received from the following organisations:

- Alto
- BT Ireland
- Digiweb
- Eircom
- Ireland Offline
- Netsource

ComReg thanks all respondents. Their views have been carefully considered, and have informed the conclusions and draft decision notice put forward in this document

ComReg's overall conclusions can be summarised as follows:

- the control will be applied ex ante rather than ex post. An ex ante control will
  preclude a potential margin squeeze, and will offer greater predictability for
  investment, and transparency in implementation and operation
- the overall approach will be forward-looking, assessing economic costs and revenues over time, and building in judgements about the likely future value of variables.

ComReg 05/88

<sup>&</sup>lt;sup>1</sup> Document No 05/11r; published on 17/02/05 - Market Analysis - Wholesale Broadband Access (Decision Notice - Designation of SMP and Related Remedies)

- eircom's costs and revenues will be used as the basis for establishing those of a similarly efficient operator<sup>2</sup>, and these will be modified to take account of differences in scale, costs which would be borne by efficient new entrants but not by eircom, and of the evolution of prices
- a discounted cash flow (DCF) analysis will be adopted, with a DCF analysis carried out over a 5 year period, and truncated with a terminal value
- the margin squeeze test will be applied on a product by product basis with a separate control for each wholesale and retail product pair
- the margin will be reviewed annually, unless circumstances in the market change significantly
- where eircom makes changes to the prices of existing products or to promotions around existing products eircom will not be required to submit the proposed changes to ComReg prior to notifying the wholesale market.
- where eircom introduces new products or new discount schemes it will be required to submit any proposals to ComReg in advance, and will be required to provide the information necessary to assess whether or not there is a potential margin squeeze.
- Where eircom is supporting multiple retail products from a single wholesale
  product, ComReg believes it is appropriate when assessing compliance with
  the wholesale price control to use the weighted average (by number of
  subscribers) of the retail products as the reference point for the retail minus
  control.
- the wholesale market should be notified 15 working days in advance of any proposed change to wholesale prices.

Appendix A includes a list of the draft directions. A draft of the control mechanism is included as Appendix B. ComReg welcomes comments on its draft directions and the draft control mechanism. The consultation period will run until 21 December 2005.

<sup>&</sup>lt;sup>2</sup> Defined in the "consultation issue leading to Q2" in Section 4.2 as an operator "which shared the same cost function as eircom's own downstream businesses but which did not yet necessarily enjoy the same economies of scale and scope as eircom's overall business."

#### 2 Introduction

#### 3.1 Background

In its Recommendation on relevant product and service markets, the European Commission identified the market for wholesale broadband access as being a market which could be susceptible to ex ante regulation. ComReg initiated a national consultation on this market on 5 March 2004 (ComReg Document 04/25). ComReg received and analysed responses from industry, and responded to the consultation with a proposed draft measure on 29 July 2004 (ComReg Document 04/83). In Decision Notice 03/05 (Document Number 05/11r) ComReg designated eircom with Significant Market Power (SMP) in the market for wholesale broadband access and imposed a price control obligation.

ComReg's conclusion from its analysis of the Market was that a retail minus price control was necessary in the Market for two main reasons, namely:

- firstly the application of a retail minus scheme would ensure that an operator identified as having SMP in the wholesale bitstream market could not leverage its significant market power in the Market into the prospectively competitive downstream retail market for providers of retail broadband services by applying margin squeezes.
- secondly, a suitably designed retail minus measure affords an opportunity for ComReg together with industry to establish an environment in which there would be greater transparency and more certainty for future of the market.

In Document 05/11r³ (following on from its definition of the Market and its designation of eircom as having SMP in the Market), ComReg imposed an interim ex ante retail minus price control on eircom. ComReg indicated that this would be an interim measure that would apply until a subsequent consultation on a replacement price control was completed and ComReg issued a new direction on the application of a retail minus control in the Market.

ComReg published its Consultation<sup>4</sup> on a retail minus price control for the WBA market on 19 August 2005. The consultation period ran until 16 September 2005. ComReg received responses from the following organisations:

- Alto
- BT Ireland
- Digiweb
- Eircom

<sup>&</sup>lt;sup>3</sup> Document No 05/11r; published on 17/02/05 - Market Analysis - Wholesale Broadband Access (Decision Notice - Designation of SMP and Related Remedies)

<sup>&</sup>lt;sup>4</sup> Document No 05/67; published 19/08/05 – **Consultation on retail minus price control for WBA Market** 

- Ireland Offline
- Netsource

ComReg thanks all respondents for their submissions. Having considered the views of all respondents, ComReg sets out in this document its conclusions regarding the implementation of a retail minus price control in the form of a draft directions and a draft control mechanism. ComReg is consulting on the measures contained in the draft directions and the draft control mechanism. The consultation period will run from 23 November 2005 until 21 December 2005.

### 2.1 Regulatory objectives under the Communications Regulation Act, 2002

Section 12 of the Communications Regulation Act, 2002 outlines the objectives of ComReg in exercising its functions. In relation to the provision of electronic communications networks, electronic communications services and associated facilities these objectives are to:

- promote competition
- contribute to the development of the internal market, and
- promote the interests of users within the European Union.

It is ComReg's view that these proposals are in line with the objectives set out in the Communications Regulation Act, 2002. The purpose of the proposals is to seek to promote competition amongst operators to ensure that end-users derive the maximum benefit in terms of price, choice and quality.

#### 2.2 Regulatory Framework

Four sets of Regulations,<sup>5</sup> which transpose into Irish law four European Community directives on electronic communications networks and services,<sup>6</sup> entered into force in

ComReg 05/88

<sup>&</sup>lt;sup>5</sup> Namely, the European Communities (Electronic Communications Networks and Services) (Framework) Regulations 2003 (S.I. No. 307 of 2003), ("the Framework Regulations"); the European Communities (Electronic Communications Networks and Services) (Authorisation) Regulations, 2003 (S.I. No. 306 of 2003), ("the Authorisation Regulations"); the European Communities (Electronic Communications Networks and Services) (Access) Regulations 2003 (S.I. No. 305 of 2003), ("the Access Regulations"); the European Communities (European Communications) (Universal Service and Users' Rights) Regulations 2003 (S.I. No. 308 of 2003), ("the Universal Service Regulations").

<sup>&</sup>lt;sup>6</sup> The new regulatory framework for electronic communications networks and services, comprising of Directive 2002/21/EC of the European Parliament and of the Council on a common regulatory framework for electronic communications networks and services, ("the Framework Directive"), OJ 2002 L 108/33, and four other Directives (collectively referred to as "the Specific Directives"), namely: Directive 2002/20/EC of the European Parliament and of the Council on the authorisation of electronic communications networks and services, ("the Authorisation Directive"), OJ 2002 L 108/21; Directive 2002/19/EC of

Ireland on 25 July 2003. The final element of the European electronic communications regulatory package, the *Privacy and Electronic Communications Directive*, was transposed into Irish law on 6 November 2003.

The new communications regulatory framework requires that ComReg define relevant markets appropriate to national circumstances, in particular relevant geographic markets within its territory, in accordance with the market definition procedure outlined in the *Framework Regulations*<sup>7</sup>.

#### 2.3 Structure of Document

The remainder of this document is structured as follows:

Section 3 summarises the views of respondents and outlines ComReg's conclusion on each of the questions posed in the consultation.

Section 4 outlines the Regulatory Impact Assessment.

Appendix A gives a list of Directions and Decisions.

Appendix B is a draft direction which includes the proposed wholesale retail minus price control mechanism.

Finally, Appendix C gives a description of the discounted cash flow model.

the European Parliament and of the Council on access to, and interconnection of, electronic communications networks and services, ("the Access Directive"), OJ 2002 L 108/7; Directive 2002/22/EC of the European Parliament and of the Council on universal service and users' rights relating to electronic communications networks and services, ("the Universal Service Directive"), OJ 2002 L 108/51; and the Directive 2002/58/EC of the European Parliament and of the Council concerning the processing of personal data and the protection of privacy in the electronic communications sector, ("the Privacy and Electronic Communications Directive"), OJ 2002 L 201/37.

<sup>&</sup>lt;sup>7</sup> Framework Regulation 26.

#### 3 Decision Notice Issues

#### 3.1 Context

In the last consultation, ComReg outlined the context within which it was considering the application of a retail minus price control. In summary, ComReg's view was that the objective of a price control was to prevent a vertically integrated operator with SMP at a wholesale level engaging in a margin squeeze to gain market power in the downstream retail market.

One respondent expressed a view that the potential for a margin squeeze was low, because there was little advantage for the SMP operator. In the consultation paper, ComReg set out its view on the incentives for a vertically integrated SMP operator to engage in margin squeeze. This view was supported by all other respondents, some of whom provided examples of perceived margin squeezes.

ComReg therefore maintains its view that a price control is necessary in this market because of the market conditions identified in the market review<sup>8</sup>, and notes industry support for this perspective. Furthermore, ComReg maintains its view that of the available forms of price control, a retail minus price control is the most appropriate.

#### 3.2 Response to consultation questions

Responses to each of the questions raised in the consultation are summarised below. In relation to each question, ComReg has summarised the consultation issue, restated the question, followed by a summarised version of the key points raised by respondents and ComReg's position on the issue. ComReg has carefully considered all responses in order to form its conclusions. Subsequent to the publication of this paper, ComReg will publish non-confidential versions of all submissions received in line with its standard practice.

### 3.2.1 Principles of ex ante and ex post application of retail minus Consultation issue leading to Q1

In the consultation, ComReg outlined the advantages and disadvantages associated with ex ante and ex post approaches. It was ComReg's proposal that ex ante measures were required to prevent vertical leveraging by means of a margin squeeze in the WBA market, and to ensure that competition became established.

ComReg's analysis in the WBA market review concluded that the wholesale market was not effectively competitive, and that regulatory intervention was required to address this. It was seen as unlikely that ex post regulation would be sufficient to ensure that eircom did not leverage its SMP in the wholesale market into the potentially competitive downstream market. This was because ex post measures

ComReg 05/88

<sup>&</sup>lt;sup>8</sup> Document Number 04/83 - Market Analysis: Wholesale Broadband Access (Response to Consultation Document 04/25 and Draft Decision)

generally address the abuse of a dominant position, rather than the holding of a dominant position. Specifically in this market, this means that by the time problems associated with margin squeeze were recognised and reported, and an ex post test was carried out, market share could already be secured by a vertically integrated firm, and alternative market entry prevented. In determining that a market should be subject to ex ante regulation, the European Commission in its Recommendation on Relevant Markets<sup>9</sup> has already concluded that competition law is insufficient to address the competition failures identified.

In considering the relative merits of ex ante and ex post application, ComReg returned to its overall regulatory objectives. Given the current situation of the broadband market in Ireland, ComReg believed that it would not yet be sufficient to rely on an ex post margin squeeze test to control market power. Were the market more mature and closer to effective competition, then an ex post test would be the appropriate way in which to test for a margin squeeze.

An ex ante approach would offer greater transparency in the market, because any proposed changes to price would be assessed for potential margin squeeze before they came into effect, using a method and a process which had already been agreed with industry. This would effectively preclude the possibility of margin squeeze. The benefit for the regulated firm was that it would know what margin squeeze test would be applied and how, and would therefore be able to ensure compliance. The benefit for OAOs would be that potential margin squeeze would be excluded, and would be seen to be excluded

A second benefit of an ex ante approach was that it would offer greater predictability in the market. By agreeing a retail minus price control which would run for a specified period, all operators would be able to plan their product offerings and business strategies with more secure financial information

ComReg noted that the principles of transparency and predictability are important, and that these principles would best be achieved by implementing an ex ante approach.

### Q. 1. Do you agree with the application by ComReg of an ex ante retail minus test?

Views of respondents

Five respondents agreed that an ex ante retail minus test was appropriate and proportionate. Respondents noted that the market was not effectively competitive,

8

<sup>&</sup>lt;sup>9</sup> Official Journal of the European Union, Commission Recommendation of 11 February 2003(2003/311/EC).

and believed that eircom was dominant in the wholesale and retail markets. It was indicated that an ex post test would not be enough to prevent margin squeeze, as market entry could be deterred before detection.

One respondent, who agreed that an ex ante test should be carried out, expressed concern about the limitations and boundaries of the test proposed by ComReg. It was suggested that the test should address all elements of the cost structure, as otherwise, the margin squeeze could be applied at a different charging point. This issue was picked up by another respondent who asked for clarity and transparency in the definition of the retail product.

One respondent did not agree that an ex ante retail minus test was appropriate, primarily because of the risk of false positives<sup>10</sup>. This respondent proposed that a false positive would lead to inefficient entry in the down stream market. The respondent believed that current bitstream prices were more than sufficient to maintain a margin which would sustain competition.

#### ComReg's position

As stated in the consultation paper (ComReg 05/67), ComReg believes that eircom, as a vertically integrated operator, has a clear incentive to squeeze margins<sup>11</sup>. ComReg has considered carefully the implications of carrying out an ex ante and an ex post approach. An ex ante approach increases transparency and predictability for all operators, and precludes the possibility of margin squeeze. However, it is more burdensome on the regulated firm and on the regulator, and may delay product innovation. An ex post approach would be more flexible, and less burdensome on both the regulator and the regulated operator. Its key disadvantage is that by its nature it is applied after the event, by which time market share may already be secured, or market entry prevented.

ComReg is very clear that the eventual aim is to have an effectively competitive broadband market. At some point in the future, as the market moves towards effective competition, it is possible that an ex post test would be the appropriate way to test for any perceived margin squeeze. However, the analysis of the WBA market in Ireland indicated conclusively that the broadband market was not yet effectively competitive. Because of the current status of the broadband market, and given the very real possibility of foreclosure at the retail level, ComReg maintains its view that its regulatory objectives would be best served by adopting an ex ante approach.

9

ComReg 05/88

<sup>&</sup>lt;sup>10</sup> This is sometimes known as a "Type 1 error" – where a margin squeeze is not taking place, yet because of using incorrect information (perhaps that the costs attributed are incorrect) the regulator concludes a margin squeeze exists. ComReg is aware of this problem, and is taking steps to avoid it. However, it should be pointed out that there are also "type 2 errors", where a margin squeeze is taking place yet the regulator concludes that it is not. This is also harmful as it would result in efficient entry being deterred and consequent harm to competition and consumers.

<sup>&</sup>lt;sup>11</sup> Refer to Section 3.3 of Document No 05/67; published 19/08/05 – **Consultation on retail** minus price control for WBA Market

ComReg returns to the points raised by respondent in relation to the cost structure and the definition of what constitutes a new product in its response to Q5 and Q15 respectively.

#### Decision No. 1. An ex ante retail minus test will be applied.

#### 3.2.2 Similarly efficient operator

Consultation issue leading to Q2

ComReg proposed to use as the benchmark the concept of a 'similarly efficient operator' i.e. one which shared the same cost function as eircom's own downstream businesses but which did not yet necessarily enjoy the same economies of scale and scope as eircom's overall business.

ComReg considered that the appropriate conceptual approach for the margin squeeze analysis would be to establish a margin which would allow a similarly efficient operator to enter the market today, to incur the relevant start-up costs and initial losses and still expect to be able to recover costs over a reasonable period of time, and to compete effectively with eircom in the broadband services market.

ComReg recognised that in making adjustments to costs and applying the margin squeeze framework to a similarly efficient operator, it needed to avoid promoting inefficient entry, and to avoid the distortion of incentives to invest in infrastructure.

## Q. 2. Do you agree that a "similarly efficient operator" constitutes an appropriate benchmark for a market squeeze analysis?

Views of respondents

All respondents agreed that the concept of a "similarly efficient operator" should be the benchmark for a market squeeze analysis.

One respondent proposed that, in order to avoid distorting the incentive to invest in LLU or in direct connection, the appropriate benchmark was an operator with the same economies of scope as eircom.

However, another respondent believed strongly that ComReg should be setting a margin which was reasonable for the WBA market in its own right, without reference to the LLU market. The respondent suggested further that an approach which used considerations from the LLU market in setting margins for WBA could promote inefficient entry to LLU.

ComReg's position

ComReg notes general agreement that the concept of a "similarly efficient operator" should be used as a benchmark for a margin squeeze analysis.

Several respondents raised concerns about the linking of regulatory policy on LLU with regulatory policy on broadband. ComReg's discussion of LLU was in the context of perceived longer term development of the market for the provision of broadband services, and it remains essential that a price control does not distort incentives to invest in infrastructure by operators.

### Decision No. 2. A "similarly efficient operator" constitutes an appropriate benchmark for a margin squeeze analysis.

### 3.2.3 Dynamic issues – historic costs or forecasted future costs? Consultation issue leading to Q3

ComReg proposed that it was appropriate in this market to apply retail minus based on current data and judgements about future variables.

In applying a margin squeeze test in this market ComReg could rely on historical or current data when assessing the unit cost function of a similarly efficient operator. The benefit of doing this is certainty: the data are known. The downside of doing this is the possibility that recent data are a poor indicator for the values of variables in the future. Where such data are employed they might suggest that the ex ante retail minus test would require that the vertically integrated firm's downstream unit cost is less than or equal to the retail price minus the wholesale price. This margin could be large and may overstate current and future downstream costs and as a result would impose an unfair burden on the regulated firm. Alternatively, by using the costs of the regulated firm the value of the vertically integrated firm's downstream unit cost may be low (as this firm may already enjoy substantial economies of scale due to first mover advantages). In this case the margin could be too small and would impose an unfair burden on entrants.

Another option would be for ComReg to rely on current data and judgements about the values of variables in the future. The benefit of doing this is that the effects of scale and learning economies may be factored in, along with any effects associated with introductory offers. The costs of adopting this approach would derive from errors in the judgements formed.

As the broadband products sold in this market have a customer life-cycle of many months (typically more than twelve and possibly up to forty eight), it will be necessary to form judgements about the value of future costs and prices (or revenues). In addition the prevalence of scale and learning economies also favours the application of current data and judgements about the values of the relevant variables.

Q. 3. Do you agree that in this market it is more appropriate to adopt an approach that can consider future costs and revenues rather than relying on historical costs and revenues?

#### Views of respondents

All respondents agreed that an appropriate approach would consider future costs and revenues. It was noted generally that a balance was needed between incorporating current data and understanding that a forward looking approach would need to include judgements about a range of potential developments in the market. One respondent suggested that experience with the development of the broadband market in other countries should be considered. Another suggested sunk costs should be included as part of the true cost of supply, and also that the accuracy of eircom's previous forecasts should be considered when assessing new forecasts.

#### ComReg's position

ComReg welcomes support for this proposal, and will take into account practical suggestions made by respondents.

ComReg believes that in determining the appropriate margin, ComReg should model the costs of a hypothetical new entrant growing the business to maturity. This will entail the use of forward looking assumptions on the evolution of costs and revenues. However, in making these forecasts, ComReg will take account of recent and current evidence of costs and revenues actually incurred by operators.

ComReg shares the concerns of one respondent in relation to sunk costs. However ComReg believes that the basis for setting margins should be the costs incurred by a theoretical new entrant rather than those incurred by eircom's retail operation and therefore the issue of sunk costs will not arise.

#### Decision No. 3. Future costs and revenues will be used in ComReg's analysis.

#### 3.2.4 Discounted cash flow analysis

Consultation issue leading to Q4

ComReg proposed to use a DCF method to assess margins. The proposed approach would use eircom's costs – both historic and forecast – as a data source, but would adjust these costs to reflect the position of a market entrant today.

ComReg noted that a DCF approach is often used to assess business investment decisions. It uses the future value of cash flow projections and discounts them to arrive at a present value. An investment is worthwhile if the sum of the DCFs is non negative.

To apply retail minus ex ante using DCF requires an assessment of margins over time, which may be set as time varying or time invariant as retail prices change and volumes grow. Having established an appropriate margin between retail and wholesale prices, that produces non negative returns from the DCF analysis, an ex ante retail minus control is formulated that will preclude the application of a margin squeeze.

The application of the DCF method would be undertaken by ComReg by assessing the profitability of a similarly efficient operator to eircom, using data from eircom but making appropriate adjustments to account for the evolution of future prices and costs specific to a new entrant. The DCF approach would be employed in such a way that the similarly efficient operator obtained a non-negative return over some defined period. Having ensured this outcome, the model would inform ComReg of the appropriate margins.

Applying a DCF method using forward looking assessments for the relevant variables requires judgements to be made about costs and revenues, as well as about the appropriate cost of capital.

A DCF approach seemed to be the most appropriate tool in the circumstances and had the virtue that it is widely understood and used in business planning for analysing capital investment projects where the returns are realised over a number of time periods.

## Q. 4. Do you agree that a discounted cash flow (DCF) method should be used to assess margins?

#### Views of respondents

All respondents agreed that a DCF method should be used. One respondent proposed that other techniques such as internal rate of return (IRR) should be used in addition to DCF, and that the analysis should apply to all elements of the cost structure such as backhaul and service fees. Another respondent expressed a view that ComReg should consider actual costs of an efficient operator, and that its view was that eircom was not an efficient wholesaler.

#### ComReg's position

ComReg welcomes support for the use of a DCF method in the analysis. ComReg's proposed approach involves a modelling exercise which will use eircom's costs as a basis from which to build the hypothetical costs of a similarly efficient operator. In performing a DCF analysis, ComReg has taken account of all the costs incurred by an operator availing of the bitstream service (including backhaul). For clarification, a DCF analysis should return the same result as IRR (as IRR is the discount rate

where the NPV of the cash flows equals zero) but is simpler and does not return multiple solutions as IRR can sometimes do.

#### Decision No. 4. A discounted cash flow model will be used to assess margins.

#### 3.2.5 What time horizon?

Consultation issue leading to Q5

ComReg proposed that it was reasonable to carry out the DCF analysis over a five year period and truncate the analysis at that point with the inclusion of terminal values. The consultation outlined the need to balance the requirement for a sufficient length of time for an entrant to recover legitimate losses against the issue of reliability of forecast data.

In assessing the profitability of a similarly efficient operator, there are a number of approaches for dealing with assessing profitability over time. For example, the profitability of the key current investments would imply using the economic life of those assets. Alternatively a very long-run approach which would consider profitability over the whole lifetime of the business or at least over multiple investment cycles could be employed.

Inevitably there is a trade-off between reliability and the horizon chosen. The longer the horizon, the more demanding the forecasting exercise and the more unreliable the forecasts – particularly as it would require the formation of judgements about future technologies.

Given the considerable difficulties associated with taking a very-long run approach with services which are still developing, as is the case here, ComReg proposed to adopt an approach which focuses on the nature of the underlying investments and an appropriate period for the recovery of those investments.

ComReg considered that an appropriate timescale would be one under which the current investments were profitable and where the firm did not rely on earning supernormal profits on future investments. On this basis, ComReg considered that it would be more reasonable to specify a time period that was related to the economic life of the underlying assets rather than the very long-run.

It was ComReg's view that a shorter time period could be chosen, but this would run the risk that it would not allow sufficient time for an entrant to recover early year losses, resulting in an inflated margin in order to pass the margin squeeze test, and thus encouraging inefficient entry. Conversely, a longer time period places reliance on forecasts for an extended period into the future, increasing the risk of forecast error and the likelihood of generating unreliable results.

## Q. 5. ComReg proposes to carry out the DCF analysis over five years. Do respondents agree?

#### Views of respondents

Respondents had mixed views. Two respondents believed that five years was appropriate. The remaining four respondents believed that a five year assessment would be prone to error, and noted the rapidity of change in the market. It was proposed that three to four years was a better option.

#### ComReg's position

ComReg recognises that the selection of time period for the DCF analysis has to balance the risks between taking a view that is too short against the risks of a view which is too long. ComReg has considered the views of respondents and concluded that five years is the minimum credible period to model a hypothetical operator from entry in the market to maturity in the market. This will minimise the risk of forecasting error, while still allowing time to consider the growth pattern of new entrants, and is appropriate for a rapidly changing market.

ComReg is also advocating an annual review to verify if the modelling assumptions and parameters have changed. This is elaborated in ComReg's response to Q7.

### Decision No. 5. The discounted cash flow analysis will be carried out over five years.

#### 3.2.6 Terminal value

Consultation issue leading to Q6

ComReg proposed to apply the DCF analysis for five years and truncate at that point instead of trying to project the cash flows to infinity. The standard approach for dealing with such truncation is to consider the terminal value associated with the activity at the time of truncation to reflect the fact that the business will continue beyond this time and that assets have an on-going economic value (or earning power).

The approach which ComReg proposed would allocate the costs of assets to be recovered (including cost of capital) between two periods of time: the explicitly-modelled period of five years from launch up to the point of truncation and the years beyond the point of truncation. The smaller the terminal value is, the greater the amount of cost to be recovered in the explicitly modelled period.

Q. 6. Do you agree that it is appropriate to apply a terminal value? If so, what do you believe would be an appropriate method for calculating this value?

#### Views of respondents

Four respondents commented on this issue. One did not think that it was appropriate to apply a terminal value because costs associated with selling bitstream are more concerned with operating costs than capital costs. Another respondent proposed that if a DCF was carried out over a five year period, the terminal value would tend towards zero, because investment in retail markets which did not involve infrastructure would be made with an expectation of profit within a five year period.

One respondent proposed the use of the Modern Equivalent Asset method.

Another respondent proposed that the most appropriate method was to add the cash flow of the five years covered by the DCF analysis to three subsequent stable years, and then close the business. It was the respondent's view that in a dynamic broadband market, prices would drop towards the competitive over five years, so that the free cash flow for three further years could be confidently forecast.

#### ComReg's position

ComReg has considered the views of respondents and concluded that a terminal value is required. ComReg acknowledges that there is artificiality in truncating a modelling period; in reality, customers are clearly acquired throughout the modelled period. These customers, especially those acquired towards the end of the modelling period would continue to contribute positive cash flows in future periods. Therefore, to exclude the revenue generated by these customers would artificially deflate the expected cash flows.

ComReg has considered a number of different methods of computing terminal values including those suggested by respondents. In addition ComReg has performed a benchmarking exercise looking at recent valuations of broadband ISPs. In light of the analysis performed, ComReg has concluded that using the net cash flow of the final year, carried forward for three subsequent years is an appropriate method of computing the terminal value.

Decision No. 6. ComReg will apply a terminal value. The terminal value will be calculated by using the net cash flow of the fifth year of the discounted cash flow analysis, carried forward for three subsequent years

#### 3.2.7 How often should the margin be reassessed?

Consultation issue leading to Q7

ComReg proposed that the retail minus control should be subject to annual review, but that significant changes in the market would initiate an immediate review.

The consultation considered whether the retail minus control should be invariant over a fixed period, and if so, how long. A range of options were examined. One option would be to review the appropriate margin every time there was a retail price change. While this would ensure that the control was highly responsive to the market, and that any changes in product or service offering could be accommodated, ComReg considered that this approach could not provide the level of predictability which is required by all operators. As the provision of greater certainty as regards the margin is a key purpose of proposing this measure, triggering a review each time there is a retail price change would not achieve the regulatory objectives.

Another option would be to fix the margin for a multi-year period, perhaps by linking the margin review with the next market analysis of the WBA market. The advantage of this approach is that at this time ComReg will revisit the issues of market definition and its assessment of market power. ComReg would also have the opportunity to assess whether the remedies imposed as a result of the first market review had been effective and proportionate in addressing SMP in the WBA market, and would examine the level and nature of market entry.

However, while this approach would have appeal in a market which was mature and fairly predictable, a price control applied over a multi-year period may not be the best option in a dynamic and new market. ComReg noted that other regulatory interventions, such as Local Loop Unbundling (LLU) are likely to impact on the WBA market over the next few years. While a longer term fixed margin would deliver greater certainty for operators, the lack of responsiveness to change would be a major disadvantage to eircom, the other operators and the end customer.

In order to balance the provision of greater certainty with the need for flexibility, ComReg therefore proposed that the retail minus control should be subject to annual review to ensure that it was providing the appropriate margin. ComReg noted that an overall analysis of the principles underpinning the setting of a retail margin would form part of the next market review of the WBA market, and that this process may take precedence in the review timetable. Further, ComReg would wish to retain the flexibility to initiate a review of the margin should circumstances in the market change significantly.

Q. 7. Do you agree that the margin should be subject to annual review, unless triggered by significant change in the market?

Respondents agreed with the principle of review. One respondent agreed with ComReg's proposal and proposed that a reasonable complaint should also trigger a review. A second respondent proposed that not only the margin but the whole structure of the price control should be reviewed frequently. A more frequent review was also proposed by another respondent. One respondent suggested that the timing of the review was less important than its transparency.

#### ComReg's position

The objective in establishing a structure for the review of the price control is to increase predictability in the market. ComReg therefore has concerns about a review period which was less than one year. Equally, given the uncertainties in the forecasting process, ComReg believes that leaving the review for longer than a year would be inappropriate.

ComReg does not feel it would be appropriate to trigger a review of the margins on receipt of a complaint. However, should an operator feel that eircom is acting in a manner that is inconsistent with the objectives of the price control, or is not in compliance with its obligations, that operator is entitled to raise a formal dispute on the subject with ComReg.

In respect of one respondent's point that not only the margin but the whole structure of the price control should be reviewed frequently, ComReg believes this has some validity. Significant unforeseen developments in the market could require an amendment to this control. Therefore, ComReg will continuously monitor the functioning of the control and may from time to time amend it if necessary. Additionally, on the first annual review of the margins, ComReg will consider the functioning of the control in light of practical experience and may issue amendments if appropriate.

Decision No. 7. The margin will be subject to annual review. In the event of significant unforeseen changes in the marketplace, ComReg will review the price control and may issue amendments.

#### 3.2.8 What discount rate should be applied?

Consultation issue leading to Q8

ComReg proposed that eircom's cost of capital would be used as a basis for calculating the relevant discount factor in the DCF analysis.

In the DCF exercise ComReg will need to apply a discount rate. It was proposed that it would be appropriate to use a measure of the weighted average cost of capital (WACC). This may be applied on a pre-tax or post-tax basis. ComReg noted that eircom currently applied a pre tax WACC of 11.5% in its separated accounts.

## Q. 8. Do you agree that eircom's cost of capital should be used as the basis for calculating the discount factor in the DCF analysis?

#### Views of respondents

Respondents had mixed views on this. Three respondents were generally in favour, but qualified this by proposing that the rate should be for new capital employed, or that adjustments should be made to reflect differences in market position.

Three respondents did not agree. One respondent suggested that eircom's weighted average cost of capital (WACC) was a return reflecting the risk of investing in infrastructure, which was a risk not incurred by a bitstream access seeker. For the latter, most business costs were seen as variable and the debt structure was likely to differ.

Two respondents proposed that a new OAO would have higher capital costs than eircom, and so eircom's costs would not be appropriate. It was noted that eircom was a long established utility and was active in all markets. Eircom would be considered low risk, and could access capital on more favourable terms than a new entrant.

#### ComReg's view

ComReg has reviewed the responses received and acknowledges the arguments for higher and lower discount rates than the one proposed in the consultation. During the sensitivity analysis performed as part of the DCF modelling exercise, it was noted that the outcome was not particularly sensitive to changes in the discount rate. ComReg has concluded that eircom's cost of capital is a useful benchmark that has been formally reviewed and approved under the price cap. Therefore, ComReg proposes to use it as the basis for the DCF analysis.

### Decision No. 8. eircom's cost of capital will be used as the basis for calculating the discount factor in the discounted cash flow analysis.

#### 3.2.9 Individual services or a portfolio?

Consultation issue leading to Q9

ComReg proposed to apply the ex ante retail minus framework on a product by product basis rather than on a portfolio basis.

eircom currently provides a number of retail ADSL products. Each of these retail products has a corresponding wholesale bitstream product. Entrants are also likely to offer a portfolio of services and could choose to compete with eircom across a similar product portfolio. However, it seemed premature for ComReg to make judgments about the nature of entry. ComReg's margin squeeze testing to date has

been based on the price differential between each of the retail products and its equivalent wholesale product.

The consultation recognised that the main benefit of conducting a margin squeeze test across a portfolio of products was that it offered the operator greater flexibility in designing its offerings, and consequently could lead to greater innovation in the market. However, a portfolio approach is necessarily less transparent, and risks an operator squeezing the margin on one product while remaining within the overall balance of the portfolio.

Ensuring that there is no margin squeeze on any individual product should avoid an entrant having to replicate eircom's product mix in order to be viable. Conducting the margin squeeze at the level of the individual product would allow OAOs to enter the market and target specific retail segments. If the analysis was performed on a portfolio basis eircom could potentially cross subsidise between retail products such that addressing certain segments in isolation would be unattractive.

## Q. 9. Do you agree that the margin squeeze test should be conducted on the basis of individual products?

#### Views of respondents

Five respondents agreed with this proposal. All noted that if the test were carried out on a portfolio basis, it would be possible to pass the test even if there were a margin squeeze on some individual components of the portfolio. One respondent stated that this approach was not just desirable but essential, and extended this to suggest that the entire cost of the product should be taken into account when applying the test.

One respondent did not agree that the margin squeeze test should be carried out on the basis of individual products. This respondent's view was that bitstream access seekers offered some services below cost in order to build market share, and that reductions in wholesale prices were anticipated in the pricing of retail offers. The respondent suggested that access seekers did not replicate eircom's retail offering, but built innovative solutions.

#### ComReg's position

ComReg has considered carefully the advantages and disadvantages of calculating the margin on the basis of a portfolio of wholesale products and on the basis of individual wholesale products. A portfolio approach would offer the regulated operator greater flexibility, and could promote greater innovation. However, a portfolio approach is necessarily less transparent, and risks an operator squeezing the margin on one product while remaining within the overall balance of the portfolio. An individual product approach offers less flexibility, but allows OAOs to enter niche markets.

ComReg's view is that a margin squeeze test at the level of individual wholesale products would preclude the possibility of a deliberate or inadvertent margin squeeze of elements of a portfolio, even when the overall portfolio passed a margin squeeze test. This approach would also fulfil the requirement for transparency, and is the most appropriate way to meet overall regulatory objectives.

### Decision No. 9. The margin squeeze test will be conducted on the basis of individual products.

#### 3.2.10 Wholesale products without retail equivalents

Consultation issue leading to Q10

ComReg proposed that wholesale products, without an equivalent retail product offered by eircom, should be excluded from the margin squeeze analysis.

In addition to the wholesale products that eircom uses to support its own retail products and services, it also offers two wholesale products which its own retail arm does not avail of. ComReg document 05/11r imposes upon eircom an obligation to meet reasonable requests for wholesale bitstream products and therefore it is possible that additional products will be developed at the wholesale level which eircom will not use at the retail level. As eircom does not offer retail equivalents of these wholesale products, the application of a retail minus price control would clearly be impossible.

In order to avoid distorting the market, ComReg proposed that these products should be priced in a manner consistent with the nearest equivalent wholesale product (which eircom does avail of at the retail level) suitably adjusted to reflect differences in functional characteristics.

## Q. 10. Do you believe that this is an appropriate way to treat wholesale products which eircom does not avail of at the retail level?

Views of respondents

There were mixed views on this issue. Three respondents agreed with ComReg's proposal, and noted that ultimately OAOs sought to offer broadband services which fell into the same general category of products offered by eircom.

The respondents who disagreed did so for different reasons and proposed different solutions. One respondent believed that where a bitstream access seeker requested a wholesale product which did not have an eircom retail equivalent, then the price should be set by commercial negotiation. If there was no agreement, the functional characteristics of the product should be used to position the service in eircom's broadband portfolio.

The other respondents proposed that all products should be included in the margin squeeze analysis, by considering the retail product based on the nearest equivalent wholesale product suitably adjusted to reflect functional characteristics. It was suggested that this should be done on a compulsory basis rather than on "best endeavours". One of these respondents suggested that letting these products fall outside the scope of some form of control was more distorting to the market.

#### ComReg's position

ComReg has considered the views of respondents and supports the proposal by one respondent that in the first instance commercial negotiation should be used to determine the price of wholesale products not availed of by eircom retail. ComReg believes that the prices reached must be reasonable and if a reasonable price cannot be reached by commercial negotiation, ComReg will intervene to determine the price by reference to similar wholesale products suitably adjusted to take account of functional differences.

Decision No. 10. Commercial negotiation will be used to calculate the price for wholesale products that do not have an eircom retail equivalent. If such negotiation fails to determine a reasonable wholesale price, ComReg will intervene.

#### 3.2.11 Retail prices

Consultation issue leading to Q11

ComReg proposed to conduct sensitivity analysis when evaluating margins to assess the impact of competition on prices

A key element of the DCF analysis in an ex ante margin squeeze test is the prices of the downstream services, which generate the revenues. As the DCF exercise is forward-looking, ComReg would need to assess the likely path of costs and revenues for a similarly efficient operator to eircom. It is important when doing this that ComReg does not include the effect on revenues of the application of market power.

If a similarly efficient operator were to generate positive returns in the future due in part to its ability to exercise market power, it would be incorrect to include this in the margin calculations. Therefore the assessment of revenues will be based on the presumption that the market will become effectively competitive in future years.

The assumption is that in a competitive market, a large fall in costs would generally be expected to lead to lower prices, though this may not occur immediately. However, it seems unreasonable to suppose that firms could maintain prices above costs indefinitely in the face of competition.

ComReg suggested that real unit costs in this market are likely to fall, rather than increase, over time, and that competition among providers would reduce prices.

Q. 11. Do you agree that ComReg should perform sensitivity analysis around rates of decrease of retail prices in doing its DCF evaluation of margins?

Views of respondents

Five respondents commented, and all agreed that sensitivity analysis should be performed. One of these respondents proposed that sensitivity of prices should be tested over at least 12 months.

ComReg's position

ComReg notes agreement that rates of decrease of retail prices should be subject to sensitivity analysis, and has taken this into account in arriving at its price controls.

Decision No. 11. ComReg will perform sensitivity analysis around rates of decrease of retail prices in doing its discounted cash flow evaluation of margins.

#### 3.2.12 How should we approach costs?

Consultation issue leading to Q12

ComReg proposed to adopt a forward-looking approach which analysed economic costs and revenues over time.

The consultation considered whether it was more appropriate to use historical costs or projected costs when assessing the cost of offering service. Generally, historical costs would look at actual costs incurred over a specified period. Standard accounting techniques would be used to analyse costs and assess profits. This approach would normally be used where there is reliable time series data on which to base projections, and where the market is reasonably stable, so that past performance could provide a reasonable guide to future performance.

The WBA market in Ireland does not exhibit the characteristics which would tend towards adopting a historical costs approach. The market is relatively new, subject to potentially rapid change, and is not likely to develop maturity in the short to medium term. An alternative approach to the accounting-based focus on historic costs is to adopt a forward looking approach which analyses economic costs and revenues over time. This approach would be standard practice in business planning, and ComReg considered that, given the characteristics of the WBA market in Ireland, the most

appropriate approach was to utilise a forward looking assessment of economic costs and revenues.

## Q. 12. Do you agree that it is more appropriate to adopt a forward-looking approach to the assessment of costs?

#### Views of respondents

All respondents agreed. One respondent proposed that sunk costs should be included. Another respondent suggested that experience in other NRAs could be used to inform judgements about likely future developments.

#### ComReg's position

ComReg welcomes agreement on the overall approach. In relation to the respondent that raised the issue of sunk costs, ComReg has dealt with this issue in the section relating to Q3.

### Decision No. 12. ComReg will use a forward looking approach in assessing costs.

#### 3.2.13 Whose costs?

Consultation issue leading to Q13

ComReg proposed that eircom's costs should be used as a starting point for the assessment, and that they should be modified to take account of costs borne by similarly efficient new entrants.

The key aim of an ex ante retail minus rule to apply a margin squeeze test is to ensure that similarly efficient operators to eircom are able to compete and build a profitable business.

ComReg noted that a competition analysis approach would use eircom's costs in calculating the margin. The analysis would assess the level of downstream profitability which eircom could expect assuming it paid the same wholesale price as its competitors. This means that the calculation would be of the additional costs which eircom incurs in order to offer its retail products. These are incremental costs. While this approach has some appeal – notably that eircom's costs are known and reported – the disadvantage is that it cannot take into account costs which a new entrant would incur, but which eircom would not.

An alternative approach would be to start from the costs which would be borne by a new entrant to the market, and to build a bottom-up model of a new entrant's costs. While this may appear to be a more realistic measure of the actual experience of a new market entrant, ComReg proposed that the cost assessment was more reliable if

built on actual costs rather than hypothetical costs, and that eircom's cost profile provided the best starting point for the analysis.

ComReg's proposed approach was to set a margin which would allow an entrant of *similar efficiency* to eircom to enter the market and compete on a forward looking basis. In ComReg's view, this meant that costs necessarily incurred by a new entrant but not by eircom should be incorporated into the analysis. However, it was emphasised that new entrant costs must be considered carefully to ensure that there was no reward for inefficient entry.

## Q. 13. Do you agree that eircom's costs should be used as the basis for establishing the costs of a similarly efficient operator?

#### Views of respondents

Five respondents agreed that eircom's costs were a suitable basis for establishing the costs of a similarly efficient operator, and respondents suggested adjustments which should be made to eircom's costs to reflect costs incurred by a new entrant but not by eircom. Respondents also suggested that experience in other countries could provide useful guidance on defining efficiency.

One respondent believed that the profile of the "similarly efficient operator" is crucial in determining what costs are included, and this respondent indicated that the costs should be those of a WBA operator.

#### ComReg's position

ComReg believes that its proposed approach to use eircom's costs as a starting point, and then to make adjustments which reflect the unavoidable costs of a similarly efficient new entrant, answers all concerns raised by respondents. In modelling the costs of a hypothetical new entrant, ComReg is simply using information from eircom as a source of benchmark data to ensure that the standard of similar efficiency is maintained.

### Decision No. 13. eircom's costs should be used as the basis for establishing the costs of a similarly efficient operator.

#### 3.2.14 Assessment of compliance - existing products

Consultation issue leading to Q14

ComReg proposed that in advance of eircom making a retail price change, it should be obliged to issue ComReg with a statement of compliance setting out the precise nature of the retail price change. The statement should demonstrate the corresponding adjustments which would be made to the wholesale prices such that

compliance with the obligation was achieved. Once eircom had submitted such a statement to ComReg, it would be free to immediately notify wholesale operators of the wholesale price change without awaiting formal approval from ComReg.

In order to assess compliance with a retail minus price control, ComReg would clearly require visibility of both retail and wholesale prices, and as ComReg is proposing an ex-ante control, ComReg would need to have visibility of changes in retail or wholesale prices prior to their implementation. ComReg has not defined a retail broadband market that is suitable for ex-ante regulation and is in no way seeking the power to regulate retail prices. Any notification to ComReg by eircom of retail price changes in advance of their implementation is solely for the purpose of assessing compliance with the wholesale price control. Should ComReg find eircom to be non compliant with the wholesale price control, any regulatory interventions by ComReg would relate only to wholesale prices.

ComReg proposed that it would be undesirable if a lengthy approvals process were to delay innovation in the market place. It is in the best interests of all operators and consumers that product and service innovation is encouraged and that any mechanisms put in place cause the minimum delay. ComReg recognised that eircom should be encouraged to innovate and must be allowed to respond to price changes by other operators in the retail market. There is a need to strike a balance between protecting competitors in the downstream market from the possible application of a margin squeeze and enabling the vertically integrated SMP operator to compete fairly in the downstream market.

ComReg considered whether eircom should be required to submit proposed retail and wholesale price changes to ComReg in advance of their implementation, so that ComReg could assess whether the proposed wholesale prices were compliant with the wholesale price control. Where eircom wished to implement a change in the retail price of one of its existing products (including amendments to promotions), ComReg noted that the application of a retail minus formula was a sufficiently transparent mechanism that compliance with the obligation should be readily determined.

Following receipt of a satisfactory statement of compliance, ComReg would only intervene if it found that the proposed changes in retail and wholesale prices were not in compliance with the price control obligation. If the proposed changes were not in compliance, ComReg would either initiate enforcement proceedings or direct amendments to the wholesale prices.

Q. 14. Do you agree that in the case of amendments to prices of existing products (including amendments to promotions) it is sufficient for eircom to issue ComReg with a statement of compliance and simultaneously notify wholesale prices to other operators?

#### Views of respondents

Three respondents agreed with this proposal. Respondents noted that the proposal would encourage innovation in the market, and one respondent suggested that, in the interests of transparency, ComReg should publish the statement of compliance on its website. Another respondent, while welcoming the proposal, believed that operators did not need any advance notification of changes to products in the retail market because the "product" being considered is the wholesale product.

The respondents who did not agree with the proposal believed that operators should be notified of changes in the retail product before they came into effect, and the suggested period ranged from 15 to 20 working days.

#### ComReg's position

ComReg recognises the concerns expressed by some respondents, but emphasises that the proposal to withdraw advance notification applies to price changes, not to product changes, and was proposed with reference to changes to existing products.

ComReg has not defined a retail broadband market that is suitable for ex-ante regulation. Any notification to ComReg by eircom of retail price changes in advance of their implementation is solely for the purpose of assessing compliance with the wholesale price control. Should ComReg find eircom to be non compliant with the wholesale price control, any regulatory interventions by ComReg would relate only to wholesale prices.

It is ComReg's view that a transparent price control mechanism should make compliance straightforward, and that the proposed approach should encourage innovation and minimise delays.

#### Decision No. 14. See control in Appendix B.

#### 3.2.15 New products

Consultation issue leading to Q15

It was proposed that prior to the introduction of new products, eircom should submit to ComReg a proposal setting out precise details of the new retail product and of the corresponding wholesale product<sup>12</sup>. Within a period of five working days ComReg would revert to eircom, either confirming that the proposal was in compliance with the obligation in which case eircom could proceed immediately to notify wholesale operators, or informing eircom that further analysis to determine compliance of the proposal was required. In this case, indicative timelines for that further analysis would be set out. During the five days subsequent to eircom making a proposal to ComReg or during a period of further analysis, eircom would be prevented from introducing new wholesale products.

In the case where eircom introduced new retail products, whether by changing the technical specifications of the product or by changing the basis of pricing<sup>13</sup> of the products, the assessment of compliance with the obligations would necessarily be less clear cut than is the case of amendments to existing products. There is a tension between the desire for transparency in the wholesale price control and the need to retain flexibility so as not stifle innovation. ComReg proposed that it would be impractical to try and create a wholesale price control that accommodated all potential innovations at the retail level in advance of their introduction. Therefore, in order to prevent a potential margin squeeze, it would not be sufficient to adopt the same assessment process as proposed for amendments to existing products in the case of the introduction of new products.

In selecting the assessment procedure associated with the wholesale price control, ComReg should seek to ensure that it did not stifle innovation and cause unnecessary delays to the introduction of new products. The consultation considered whether it would be appropriate to set a fixed time limit in which to consider the compliance of new products. ComReg noted two disadvantages to this approach. On the one hand, as it is impossible to anticipate all future developments, it may not be possible to complete all assessments within a defined period. On the other hand, a fixed time period may in fact slow the introduction of new products where the assessment turned out to be relatively straightforward.

ComReg also considered whether it would be appropriate to make no commitment about the time taken to complete the assessment of new products. It concluded that this would create unnecessary and undesirable uncertainty in the market.

The process proposed by ComReg entailed eircom submitting, in advance of its introduction, details of any new retail product and its corresponding wholesale product. ComReg would revert within five working days, either by confirming compliance, or by indicating a timetable for further analysis

28

<sup>&</sup>lt;sup>12</sup> Where eircom introduces a new retail product that is not adequately supported by the existing wholesale product set, eircom will be required to introduce a new wholesale product providing equivalent functionality and/or tariff structure.

<sup>&</sup>lt;sup>13</sup> For instance, ComReg regarded the introduction of eircom's time based retail product (with its associated wholesale product) as a new product because although it had the same technical specification as the pre existing flat rate product, it was a significantly different tariff structure.

ComReg would approve the new wholesale prices once it was satisfied that the margin between retail and wholesale prices was sufficient to avoid a margin squeeze. Having determined the appropriate relationship between retail and wholesale prices for the new products, ComReg would update the retail price mechanism such that any subsequent amendments to the prices of these products would be assessed using the mechanism set out for existing products.

Q. 15. Do you believe that this proposal [eircom should submit details to ComReg of any new retail product and corresponding wholesale product; ComReg would commit to timescale for response] for price changes to new products is an appropriate mechanism for assessing the compliance of new products?

#### Views of respondents

Five respondents agreed in principle with ComReg's proposal. One of these stated that it was essential to have clarity in the definition of what constituted a retail product, and also what constituted a new product.

One respondent did not agree, and put forward the view that the only circumstance in which notification could be required would concern changes to the wholesale product. This respondent questioned what would happen should ComReg fail to meet its target response time.

#### ComReg's position

In the consultation, ComReg differentiated between changes to existing products and services, and the introduction of new products and services. While ComReg believes that it is appropriate to withdraw advance notification of price changes to existing products, it does not believe that the market is sufficiently mature to use this approach in addressing new products.

ComReg notes that there were some differences in interpretation amongst respondents. It can be clarified that this proposal refers to the introduction of a new retail product, and that the intention would be that eircom would submit price details of a new retail product and its corresponding wholesale product. ComReg would then have a 5 working day period where it would either approve the new product, or provide a timetable for its approval. Eircom would not be able to launch a new product without ComReg approval, whether this is granted within the 5 day period, or in line with the timescale produced. Therefore, eircom would not be able to launch a new wholesale product (including supplying to its own retail arm) prior to ComReg granting approval.

#### Decision No. 15. See control in Appendix B.

#### 3.2.16 Price control on connection fee

Consultation issue leading to Q16

The wholesale bitstream service is composed of a number of elements, such as connection fee, monthly rental, usage fees, cessation, backhaul and service establishment. ComReg proposed that the retail minus price control should only apply to the connection and monthly rental elements of the service.

ComReg acknowledged that there are fixed and variable elements to the cost stacks associated with the retail bitstream products. As discussed earlier, a DCF analysis was advocated for a defined period in order to take relevant account of the intricacies of the cost stacks and the product life cycle.

ComReg suggested in the consultation that it would be premature to consult on the appropriate size of the margin required for the wholesale price control. However, in the interests of transparency it was thought appropriate to consult on the form of the price control and to seek interested parties' views on whether ComReg proposals satisfied the requirements for transparency, predictability and flexibility while avoiding the possibility of margin squeeze during the life of this control.

ComReg put forward a view that the wholesale connection fee should always be cheaper than the standard (i.e. excluding temporary promotions or discount schemes) retail connection fee. Therefore, ComReg proposed that the wholesale price control for the connection fee should take the following form:

Wholesale connection fee = Standard retail connection fee \* fixed percentage(<100%)

Q. 16. Do you believe that this [wholesale connection charge as a fixed percentage of retail connection fee] is an appropriate form of price control for the connection fee?

#### Views of respondents

Five respondents were in favour of the proposal. One of these indicated that, in its view, the price control should apply to all service costs including backhaul. Another expressed concern that cost-based pricing for connection charges could act as a barrier to market growth, and suggested there may be scope to include part of the cost based charge in the rentals.

One respondent did not believe that movement in the retail connection fee should affect the recovery of the connection costs for bitstream. This respondent suggested

30

that a price control for bitstream connection could consider the discounted costs represented by the bitstream connection and the rental charges over the customer life.

## ComReg's position

ComReg notes the view of one respondent that the wholesale connection fee is potentially a barrier to competition. However, at its current levels, ComReg does not believe this to be the case. Moreover, ComReg believes that it is appropriate that there should be a discrete charge for connection, to reflect the discrete activities and costs associated with connecting a line, and does not believe that the costs associated with connections should be recovered through the recurring fees.

Another respondent suggested that the recovery of costs associated with wholesale connections should be unaffected by changes in retail prices. This respondent proposed that rather than have a separate price control for the connection, the costs to OAOs of the wholesale connection charge should be factored into the margin analysis. Having carefully considered the submissions to the consultation, ComReg believes that this proposal is superior to that set out in its consultation paper. ComReg believes that removing the linkage between retail and wholesale connection charges is both simpler and better reflects the principle of cost causation than its previous proposal. Clearly there is the potential for the costs of connection to the OAOs being greater than the revenue stream associated with retail connections. In order to avoid a margin squeeze, the costs associated with connection will have to be factored into the DCF analysis of margins and the margin between retail and wholesale rental charges will have to be sufficient to allow the recovery by the OAO of the costs of connection not recovered through retail charges.

ComReg will not impose a retail minus price control on the wholesale connection. In the absence of a retail minus control, ComReg believes that the cost for connection (or port transfer) should be cost oriented. To that end, eircom is directed to reduce the wholesale connection and port transfer prices from their current price of €60 to €30 effective from 1 January 2006. eircom shall not increase these charges without the prior approval of ComReg. ComReg would give approval for a change in this charge where there is an appropriate cost based justification.

Decision No. 16. The prices for wholesale connection and port transfer shall be cost oriented. From 1 January 2006, the price for wholesale connection and port transfer shall be €30. eircom shall not increase this charge without the prior approval of ComReg.

### 3.2.17 Price control on wholesale monthly rental

31

Consultation issue leading to Q17

ComReg proposed that the control should be a combination of a fixed monetary value and a fixed percentage.

ComReg considered three different mechanisms for the wholesale monthly rental price control.

The first option that ComReg considered was where the margin was expressed as an absolute fixed monetary value. This means that the gap between retail prices and wholesale prices is a fixed monetary value. This method was employed by ComReg for the interim price control.

While this was seen as the appropriate control for an interim period, prior to the completion of this consultation and its associated analysis, it was not thought to be an appropriate basis for a permanent price control. While a price control of this form had the advantage of preventing any decrease in the margins available to OAOs (which was the objective in the interim period), it did not seem appropriate going forward in the context of reducing retail prices and increasing volumes. By expressing the retail minus as an absolute value, there is a significant risk that an artificial price floor could be created at the wholesale level if the price were to remain above cost. This could potentially stifle expansion in this market at the retail level. Alternatively, the wholesale price could be forced below cost which would place an unfair burden on eircom and would distort the investment incentives for this product as well as other wholesale products such as unbundled local loops.

The second option that ComReg considered was to express the margin as a fixed percentage of the retail price. This scenario had the advantage of increased flexibility in terms of product pricing. However, there was a risk that if the retail prices were dramatically reduced, the corresponding margin, while maintained in percentage terms, would be reduced in absolute monetary terms to such a degree that there was a margin squeeze. ComReg's preliminary analysis of the costs associated with providing retail service in this market suggested that a significant number of unit costs were unlikely to reduce significantly as prices declined and volumes grew. In order to ensure that sufficient margin was maintained, if prices were to fall significantly, there was a risk that margins for the current prices would have to be set too high.

Finally, the third option considered by ComReg was a combination of a fixed monetary value and a fixed percentage. ComReg advocated this third option which lies between the two extremes examined above. This approach while avoiding a margin squeeze also takes into account the evolving nature of this market. ComReg acknowledged that prices were set to decrease over time as the corresponding volumes increased. Using a hybrid formula the absolute monetary value of the margin would decline but the percentage margin would increase. Such a hybrid approach also balanced flexibility with transparency and predictability for all market players. ComReg's proposed control would therefore be:

Wholesale rental price = (Retail rental price \* fixed percentage <100%) – fixed monetary value

ComReg proposed to set retail minus controls, in accordance with the formula above, for all current products. These controls would be applied by reference to the standard retail rental i.e. excluding promotions and discounts.

If eircom were to change any of the characteristics of either its retail or its wholesale products during the period of the control, it would be required to follow the approval and notification processes as described.

Where usage charges make up a significant portion of retail or wholesale prices (such as eircom's 'time' and 'kronos') ComReg proposed to convert usage charges into average monthly revenues (at retail and wholesale levels) and then apply retail test to the average recurring revenues.

Q. 17. Do you think that this is an appropriate way to apply the retail minus formula? If you feel that there is a superior formula please provide your reasoning.

# Views of respondents

Four respondents believed that ComReg's proposal was appropriate. It was noted that this was agreement in principle, as it was difficult to comment definitively in the absence of values.

One respondent questioned the treatment of commissions, and whether commissions could be used to circumvent the control. This respondent suggested that ComReg should treat commissions as a reduction in revenue.

One respondent, while agreeing that the general structure of the control was appropriate, proposed that a control should allow for movement in both connection and rental charges at the retail level in determining the wholesale rental. A separate control which isolated and linked retail and wholesale connection fees was therefore seen as inappropriate.

This respondent asked for clarification as to whether a single fixed percentage and money amount would apply to all products, or if separate parameters would apply to each wholesale/retail pair. The respondent also expressed a view that eircom was exposed to risks in recovering start-up costs just as were OAOs.

### ComReg's position

ComReg welcomes the overall agreement with the format of the control proposed by ComReg during its consultation. As stated in response to Q16 the costs to OAOs of the wholesale connection fee will now be incorporated as part of this retail minus control.

33

In Q9, ComReg stated that a separate control will apply to each wholesale product A retail minus formula will be set for each of the current products on offer (excluding promotions and discounts which are dealt with in Q18 and Q19 respectively) with separately computed percentages and fixed monetary amounts. It should be noted that any given wholesale product could support two or more eircom retail products. Where eircom is supporting multiple retail products from a single wholesale product, ComReg believes it is appropriate when assessing compliance with the wholesale price control to use the weighted average (by number of subscribers) of the retail products as the reference point for the retail minus control.

As proposed in the consultation, where usage charges make up a significant portion of retail or wholesale prices, these charges will be converted into average monthly revenues and the control will be applied accordingly.

ComReg notes the concerns raised by one respondent in relation to commission payments. Commission payments are part of the costs of customer acquisition and ComReg believes that these costs should be factored into the DCF analysis of appropriate margins underpinning this control.

## Decision No. 17. See control in Appendix B.

### 3.2.18 Retail discount schemes

Consultation issue leading to Q18

The consultation proposed that if eircom made any changes to its discount scheme during the relevant control period, then ComReg may require corresponding changes to wholesale bitstream prices. ComReg proposals would be carried out in accordance with the assessment process proposed for amendments to existing products.

eircom currently has one retail discount scheme – the Bulk Connection Discount Scheme. This provides discounts on the self install connection charge for bulk orders for DSL service. It is available for eircom's retail DSL products other than eircom Broadband Home Starter. There are a number of discount levels depending on the size of the order.

Where eircom wished to amend or introduce a new discount scheme, ComReg proposed that it would be subject to the same assessment process as outlined for new products.

Q. 18. Do you think that this [new or amended discounts treated in the same way as new products] is an appropriate way to treat discounts? If not, please state clearly the reasons for your disagreement

Five respondents generally agreed with this proposal. One emphasised the need for clarity and transparency in the definition of products and new products. Another suggested that OAOs should be given 15 working days' advance notice, during which time they could request that ComReg carried out a margin squeeze test, and a further respondent proposed a 20 working day advance notice period.

The respondent which did not agree proposed that revenues should be treated net of discount at both wholesale and retail levels when evaluating the margin on an ex post basis.

### ComReg's position

ComReg welcomes the general agreement on this proposal and remains of the view that if eircom wish to amend or introduce a new discount scheme, it would require a submission prior to its launch in line with the procedure set out in response to Q15 for the introduction of a new product. ComReg notes the disagreement of one respondent and its suggestion that a margin squeeze test should be performed ex post, but ComReg has addressed this issue in its response to Q1.

To the extent that eircom can objectively demonstrate that a retail discount scheme is based on savings achieved at the retail level, as is the case with the current bulk order discount scheme, then retail discount schemes will not necessarily impact wholesale prices. However, where a discount scheme is not based on savings at the retail level, it would constitute a reduction in the retail price. In these circumstances, compliance with the retail minus control would be assessed on the weighted average of discounted and undiscounted lines.

In the event that eircom offer retail broadband as part of a bundle of services, where the price of the bundle is less than the price of the sum of the constituent elements, ComReg would regard this as a discount scheme being applied to broadband. It would therefore require eircom to make a submission prior to its launch. This submission would follow the same process as that set out for the introduction of a new product.

### Decision No. 18. See control in Appendix B.

### 3.2.19 Retail promotions

Consultation issue leading to Q19

ComReg proposed to treat any retail promotions and their corresponding wholesale promotions relating to existing products (including extension of deadline or early withdrawal) in the same way as it would for the assessment of price changes of existing products.

eircom may from time to time introduce temporary promotions. In such circumstances, ComReg proposed that controls on promotions were necessary in

order to avoid the damaging effects of a margin squeeze on competition in the provision of retail DSL services. For many new customers the margin on the standard product will be irrelevant because they have taken up DSL under a promotion. This is particularly important for this market because of the extensive use of promotions to sell DSL products where eircom adopts a policy of near continuous, but varying, promotions.

Given the frequency with which new promotions are introduced at the retail level, ComReg's view was that it would be undesirable to amend the wholesale prices every time a new retail promotion was introduced or withdrawn. Therefore, ComReg proposed a continuation of the current arrangement whereby each retail promotion was matched by a corresponding wholesale promotion, the value of which was computed by assessing the revenue foregone at the retail level between the promoted prices and the standard prices. This wholesale promotion would take the form of a one off rebate payment.

# Q. 19. Do you believe that this is an appropriate way to treat promotions? If not, please state clearly the reasons for your disagreement

### Views of respondents

Several respondents, while generally supportive of the proposal, raised other issues and asked for clarification. One respondent requested more information about the rebate for OAOs and noted the free offerings that frequently accompany eircom promotions (such as an MP3 player). Another respondent asked how the mechanism would treat what it perceived as revenue foregone as a result of commissions offered to eircom's external sales channel, or as a result of bundling of broadband with other services. Another respondent expressed the view that retail promotions were likely to be frequent and would continue in the market. Its view was that where a retail promotion was matched by a corresponding wholesale promotion, then the wholesale rebate should be the revenue foregone at the retail level between the promoted prices and the standard prices.

One respondent did not accept that a wholesale promotion was required to match every retail price promotion. This respondent believed that OAOs were pricing retail services below the wholesale price in order to build market share, and that eircom's retail arm should have the same possibility. The respondent suggested that there was a difference between a very short term limited promotion, and one which was available over a longer time to a broader range of customers. On the question of a rebate, the respondent did not agree that there should be a single rebate payment, and proposed that it was more appropriate to require that the net present value of revenues foregone at the retail level was the same as the wholesale promotion.

ComReg's position

Having considered the responses to consultation, ComReg remains of the view that if eircom introduces retail promotions, such promotions must be matched at the wholesale level. Having determined the appropriate margin between retail and wholesale prices on the basis of the standard undiscounted tariffs, there would clearly be a margin squeeze if retail promotions were not matched by wholesale equivalents. ComReg believes the value of the wholesale rebate should be computed by assessing the revenue foregone at the retail level between the promoted price and the standard price.

One respondent suggested that OAOs are setting retail prices below wholesale costs in order to build market share, and that eircom retail ought to be allowed to do the same. ComReg does not agree with this suggestion. The purpose of this price control is to prevent eircom, as a vertically integrated operator with SMP at the wholesale level, from leveraging its market power into the retail market by means of a margin squeeze. Allowing eircom to supply OAOs at a wholesale price greater than its retail arm was supplying end customers would completely undermine the purpose of the control.

This respondent also suggested that that the wholesale rebate need not be a single payment as proposed by ComReg in the consultation. Having considered this issue, ComReg agrees that so long as the rebate is paid within reasonable period of time, and the net present value of the wholesale rebate is equal to the revenue foregone at retail level, then rebates could be paid in two or more instalments.

Two respondents raised concerns around the issue of retail incentives used by eircom retail such as the inclusion of free gifts and the use of commission payments. ComReg believes that these retail incentives should be outside the scope of the price control. The average cost of customer acquisition used in the margin squeeze analysis underpinning this price control has already factored in the costs of these activities. While ComReg believes that these retail activities are outside the scope of this control, it is clear that eircom could potentially exploit this freedom to create a margin squeeze. Therefore, ComReg will closely monitor eircom's retail activities and will investigate if it believes that a margin squeeze may be occurring. If, upon investigating ComReg concluded that eircom's retail activities were inconsistent with the cost assumptions made in the margin squeeze analysis, then ComReg would direct eircom to pay a wholesale rebate to OAOs.

ComReg remains of the view that where eircom amends or introduces promotions around existing products, then eircom should follow the compliance procedure set out in response to Q14 for the amendment of existing products. That is eircom shall issue ComReg with a statement of compliance and simultaneously notify OAOs of the imminent amendment of wholesale terms and conditions. Where eircom offer two or more promotions at the retail level supported by a single wholesale product, compliance would assessed on the basis of the expected weighted average impact of the relevant promotions.

### Decision No. 19. See control in Appendix B.

3.2.20 Other price controls – service establishment, cessation charge and bitstream connection service

Consultation issue leading to Q20

ComReg proposed that a range of other bitstream related wholesale charges and services were not so amenable to a retail minus control, mainly because they did not have direct retail equivalents. The proposed approach to controlling prices for each of these additional services is as follows:

- eircom currently sets a service establishment charge of €8,035 per Access Seeker and aims to recover eircom's costs in providing authorisation and training in bitstream IT systems for Access Seeker staff. ComReg proposed that eircom make no increase in this charge, or changes to other terms or conditions, without prior approval from ComReg. ComReg would expect to give approval where there is an appropriate cost-based justification. This charge would be reviewed at the beginning of each new price control period.
- eircom currently sets a wholesale **cessation charge** of €60 per port. This charge does not apply to Access Seekers who agree to amendment of their contracts to reflect a minimum 6 month term for each bitstream subscriber port. ComReg proposed that eircom should make no increase in this charge, or changes to other terms or conditions, without prior approval from ComReg. ComReg would expect to give approval where there is an appropriate cost-based justification.
- ComReg stated in the previous consultation that it believed that the prices for the bitstream connection services should be based on FL-LRIC costs. However, ComReg has reconsidered this position in light of the constructive engagement between eircom and access seekers in developing new backhaul products. Therefore, ComReg proposed that it would be appropriate to forbear from imposing specific obligations on these services other than that access to these services should be at reasonable prices.
- Q. 20. Do you believe that this [cost justification for service establishment and cessation charges and reasonable prices for bitstream connection service] is an appropriate way of treating the service establishment charge, cessation charge and bitstream connection service?

### Views of respondents

Several respondents answered this question in some detail, and a range of views was expressed. One respondent agreed with the proposed approach to the service establishment and cessation charges, but proposed that the cessation approach should be built into the margin test as an input cost. This respondent expressed a view that

cost based pricing for connection services could act as a barrier to market growth, and proposed that part of the cost based charge should be included in the rentals.

A second respondent broadly agreed with ComReg's proposal, and noted that treatment of OAOs should be non-discriminatory, in that for these charges which applied to OAOs and eircom's retail arm, there should be equivalence of treatment.

Another respondent agreed with the proposal for the service establishment charge. On the cessation charge, this respondent proposed that ComReg should require eircom to bring this charge into line with its retail practices, and to use the same formula as defined for the connection charge where the cessation charge applies to a port over 6 months in service. The respondent suggested that for ports under six months in service, ComReg should impose a maximum charge of €60 on the penalty rental. In the respondent's view, this would discourage superfluous signups, and would reflect the fact that this wholesale charge is given away free at the retail level. As for the connection charge, this respondent felt strongly that it was not adequate to propose an approach based on reasonable prices.

### ComReg's position

Having considered the responses, ComReg remains of the view that these wholesale charges do not have direct retail equivalents and are therefore unsuitable inputs to a retail minus price control.

One respondent proposed that the cessation fee be included in the computation of the margin. ComReg does not believe that this is appropriate as the cessation fee is only payable by operators who refuse to accept a six month minimum contract for each line. ComReg has taken account of the costs associated with the minimum contract terms. ComReg has addressed the issue of the connection fee in its response to Q16.

Another respondent proposed that the maximum rental charge for lines ceased within six months should be capped at the cessation fee of €60. ComReg believes that the current arrangements, where eircom offer two contractual alternatives - one requiring a commitment to a six month minimum term but no cessation fee, the other requiring no commitment to minimum term but requiring a cessation fee - are reasonable and amending this would be inappropriate. ComReg notes this respondent's concerns about the price of the bitstream connection service, and acknowledges that this is a significant cost for very small operators. However, as an operator's volumes grow the proportionate impact of this charge falls significantly. Furthermore, over time new products, more tailored to individual operator's requirements, will be developed and ComReg believes that reasonable prices should be achievable through commercial negotiation. Should this not happen ComReg would intervene to ensure that prices are reasonable.

The service establishment charge will be reviewed at the commencement of each new price control period. In addition, a change in price or terms and conditions for the service establishment charge or the cessation charge requires prior approval from

ComReg. In order to grant such approval, ComReg requires that eircom submit a cost based justification.

As stated in the consultation, the prices for bitstream connection services will not have specific obligations other than that access should be at reasonable prices.

- Decision No. 20. The current arrangements, where eircom offer two contractual alternatives one requiring a commitment to a six month minimum term but no cessation fee, the other requiring no commitment to minimum term but requiring a cessation fee are reasonable and amending this would be inappropriate. Over time new products, more tailored to individual operators' requirements, will be developed and reasonable prices should be achievable through commercial negotiation. Should this not happen ComReg would intervene to ensure that prices are reasonable.
- Decision No. 21. The service establishment charge will be reviewed at the commencement of each new price control period. In addition, a change in price or terms and conditions for the service establishment charge or the cessation charge requires prior approval from ComReg. In order to grant such approval, ComReg requires that eircom submit a cost based justification.
- Decision No. 22. The prices for bitstream connection services will not have specific obligations other than that access should be at reasonable prices.

### 3.2.21 Publication / notification to the wholesale market

Consultation issue leading to Q21

ComReg proposed that OAOs should be notified by eircom of proposed changes to wholesale prices 15 working days before they came into effect. There should no longer be a requirement for general publication of proposed changes.

The consultation indicated that there was the potential for a retail minus mechanism to give eircom's retail operation an unfair advantage in the market, in that they know that any change they make to retail prices will automatically trigger a change in the wholesale price. However, if a competing operator were to make a price reduction it would continue to pay the same wholesale price, and hence would have lower margins. To prevent this effect from distorting competition, ComReg proposed that it was essential that prior to eircom making any retail price change that would cause a change in the wholesale price, eircom must notify customers of its wholesale bitstream services in advance of the imminent wholesale price change.

The purpose of this requirement is to give OAOs time to consider their retail offer, given changes in the wholesale prices. ComReg considered how long the period between notification and implementation of the wholesale prices should be. In reaching an assessment of an appropriate period, ComReg balanced the need to give OAOs sufficient time to respond to changes of wholesale prices with changes to

retail prices with the desire to avoid creating unnecessary impediments on eircom's flexibility. While it is reasonable that in the current state of the market there should be some advance notification of wholesale changes, ComReg did not propose changing the notification period from the current obligation of 15 working days and did not believe that a longer period was required for OAOs to develop products and services.

The consultation noted that the prime objective of advance publication of proposed wholesale price changes was to enable the purchasers of wholesale bitstream to adjust their retail offer, should they so wish. It was therefore a matter to be addressed in the wholesale market, and the wholesale market comprises eircom and OAOs who purchase bitstream services. Given this, ComReg considered that it may not be necessary to widely publish proposed changes to wholesale prices, and that the objective may better be served by direct communication between eircom and OAOs.

Q. 21. Do you agree with ComReg's proposal that changes to wholesale prices should be notified to the wholesale market 15 days before coming into effect?

### Views of respondents

Three respondents agreed with the proposal insofar as changes were restricted to prices, but believed that 15 days' notice was not sufficient if changes included product changes. One of these respondents extended this concern to include changes to pricing structure. Another suggested that changes needed to be notified to a wider population than current service operators, and should include, for example, prospective market entrants.

Two respondents indicated that the notice period was too short, and proposed notice periods ranging from 20 days to 45 days.

One respondent expressed a view that the effect of introducing any period between the notification and implementation of price changes was to stall the market.

### ComReg's position

ComReg is concerned to balance the need to give OAOs sufficient time to respond to changes of wholesale prices with the need to avoid creating unnecessary impediments on eircom's flexibility. ComReg does not agree that, given the current state of the market, it is appropriate to remove advance notification, and notes that it is supported in this view by all but one of the respondents.

ComReg recognises the concerns expressed by respondents about the desirable length of advance notification, and emphasises that this consultation is restricted to considering changes to price. On that basis, ComReg maintains that 15 working days balances the requirements of all operators. This requirement relates solely to the price control. In decision notice 05/11r, ComReg imposed obligations relating to access and non discrimination. Where eircom introduces new wholesale products, eircom may have to notify OAOs of other non-price related aspects of these new products significantly further in advance in order to be in compliance with all its obligations.

ComReg notes the point that wholesale price changes may be of interest to a wider community than eircom and current OAOs. It is ComReg's intention that wholesale prices will continue to be publicly available on eircom's website. However, the purpose of providing advance notification of these changes is to allow OAOs to respond in their retail offerings, and this can only impact on current OAOs. Therefore, while wholesale prices will be publicly available, advance notice of changes to these prices will not be.

Decision No. 23. See control in Appendix B.

# 4 Regulatory Impact Assessment ('RIA')

### Consultation issue

In the consultation, ComReg examined the impact of elements of the retail minus price control on affected parties.

It was noted that the regulatory impact of imposing a retail minus price control had been examined and consulted on as part of the process of reviewing the WBA market, and was therefore outside the scope of this current consultation. The focus at this stage was on the relative impacts of the different choices relating to the implementation of the retail minus control.

ComReg identified the core regulatory issues as follows:

- 1. The control is in the form of an ex ante retail minus formula.
- 2. The margin squeeze test will be reassessed annually to ensure that the assumptions made in the DCF analysis are accurate, and that the coefficients of the formula are amended if necessary.
- 3. Each wholesale product will have an individual price control formula
- 4. Where eircom wish to introduce a new retail product it will be required to submit a proposal to ComReg for an assessment of a margin test prior to notifying OAOs of the impending wholesale change.
- 5. Before implementing any wholesale price changes, including allowing eircom retail to avail of the product, eircom must notify OAOs 15 working days in advance. This notification should be restricted to those OAOs with bitstream contracts and should not be publicly available.

For each core regulatory issue, ComReg outlined alternative proposals, and examined their likely impact on eircom, OAOs and consumers.

# Views of respondents

ComReg received one direct response in relation to its RIA. This respondent was generally positive about the need for regulatory intervention in the WBA market. However, it raised concerns around what it interpreted as ComReg's tampering with variables in the WBA market to improve the impact of regulatory intervention in the LLU market.

### ComReg's position

ComReg believes that enhancing competition in the WBA market will be of benefit to all competitors. It was not ComReg's intention to infer that intervention in one market would be used to address problems identified in another. However, ComReg does believe that longer term benefits to do with lowering costs and enhancing

innovation will best be achieved with a price control which does not limit incentives to invest in infrastructure.

In relation to the 5 core regulatory implications identified, ComReg has concluded the following:

- 1. The control is in the form of an ex ante retail minus formula. ComReg believes that the benefits to OAOs and consumers of using an ex ante approach justify the greater regulatory impact upon eircom compared to using an ex post approach for the reasons set out in consultation 05/67 and in Section 3.2.1 in this document.
- 2. The margin squeeze test will be reassessed annually to ensure that the assumptions made in the DCF analysis are accurate, and that the coefficients of the formula are amended if necessary. ComReg believes that the benefits to OAOs and potentially eircom of reassessing the margin squeeze test annually to ensure that the assumptions made in the DCF analysis are accurate justifies the greater regulatory impact upon eircom compared to less frequent reassessments.
- 3. Each wholesale product will have an individual price control formula. ComReg believes that the benefits to OAOs and consumers of each wholesale product having an individual price control formula justifies the greater regulatory impact upon eircom compared to the portfolio approach. This will promote competition for all products to the benefit of consumers.
- 4. Where eircom wish to introduce a new retail product it will be required to submit a proposal to ComReg for an assessment of a margin test prior to notifying OAOs of the impending wholesale change. ComReg believes that the benefits to OAOs and consumers of assessing the compliance of new products in advance of wholesale notification justifies the greater regulatory impact upon eircom compared to a statement of compliance for new products with simultaneous wholesale notification. This will reduce the risk of foreclosure and promote competition to the benefit of consumers.
- 5. Before implementing any wholesale price changes, including allowing eircom retail to avail of the product, eircom must notify OAOs 15 working days in advance. This notification should be restricted to those OAOs with bitstream contracts and should not be publicly available. ComReg believes that the benefits to OAOs and consumers of an advance notification period of 15 working days prior to the implementation of any wholesale price changes justifies the greater regulatory impact upon eircom compared to no notification period. ComReg believes that a period greater than 15 days would be disproportionate and burdensome on eircom and would negatively affect consumers by unnecessarily delaying innovation.

# Appendix A – List of Draft Directions and Decisions

# **List of Draft Directions and Decisions**

Decision No. 1.	An ex ante retai	il minus test will be applied.
Decision No. 2.	A "similarly efficient operator' margin squeeze analysis.	" constitutes an appropriate
Decision No. 3.	Future costs and revenues will be	used in ComReg's analysis.
Decision No. 4.	A discounted cash flow model will	be used to assess margins.
Decision No. 5. years. 15	The discounted cash flow analysis	will be carried out over five
Decision No. 6. calculated by usin analysis, carried for Decision No. 7. significant unforest control and may is Decision No. 8. the discount factor Decision No. 9. individual product Decision No. 10. wholesale product negotiation fails to intervene. 22 Decision No. 11.	Commercial negotiation will be us s that do not have an eircom retail of determine a reasonable wholesale	of the discounted cash flow 16 nual review. In the event of pmReg will review the price 18 If as the basis for calculating 21 21 22 24 25 26 26 26 27 28 28 29 29 20 20 20 20 20 20 20 20 20 20 20 20 20
Decision No. 12. 0	ComReg will use a forward looking a	_
Decision No. 14.	eircom's costs should be used as the efficient operator.	he basis for establishing the 25 See control in Appendix B.
Decision No. 15.		See control in Appendix B.
Decision No. 16. cost oriented. Fro	The prices for wholesale connection 1 January 2006, the price for who can be sircom shall not increase this can be seen as the	olesale connection and port
Decision No. 17.		See control in Appendix B.
Decision No. 18.		See control in Appendix B.
Decision No. 19.		See control in Appendix B.
Decision No. 20.	The current arrangements, where	eircom offer two contractual

alternatives - one requiring a commitment to a six month minimum term but no

cessation fee, the other requiring no commitment to minimum term but

45

requiring a cessation fee - are reasonable and amending this would be inappropriate. Over time new products, more tailored to individual operators' requirements, will be developed and reasonable prices should be achievable through commercial negotiation. Should this not happen ComReg would intervene to ensure that prices are reasonable. The service establishment charge will be reviewed at the Decision No. 21. commencement of each new price control period. In addition, a change in price or terms and conditions for the service establishment charge or the cessation charge requires prior approval from ComReg. In order to grant such approval, ComReg requires that eircom submit a cost based justification. 40 Decision No. 22. The prices for bitstream connection services will not have specific obligations other than that access should be at reasonable prices. Decision No. 23. See control in Appendix B.

42

# Appendix B – Draft Control Mechanism

### STATUTORY POWERS GIVING RISE TO THIS DECISION INSTRUMENT

- 1.1 This decision instrument relates to the market for wholesale broadband access ("WBA") for the purpose of delivering broadband content to end users<sup>14</sup> and is made by the Commission for Communications Regulation ("ComReg"):
  - I. Having had regard to sections 10 and 12 of the Communications Regulations Act 2002;
  - II. Having taken account, of its functions under Regulation 6 (1) of the European Communities (Electronic Communications Networks and Services) (Access) Regulations 2003<sup>15</sup>;
  - III. Having (where appropriate) complied with the Policy Directions made by the Minister<sup>16</sup>;
  - IV. Having taken the utmost account of the EU Commission's Recommendation<sup>17</sup> and the Guidelines<sup>18</sup>;
  - V. Having had regard to the market definition, market analysis and reasoning conducted by ComReg in decision 03/05<sup>19</sup>, the analysis and reasoning set out

<sup>&</sup>lt;sup>14</sup> As referred to in the EU Commission's Recommendation.

<sup>&</sup>lt;sup>15</sup> S.I. No. 305 of 2003 the European Communities (Electronic Communications Networks and Services) (Access) Regulations 2003 which transposes Directive 2002/19/EC of the European Parliament and the Council of 7 March 2002 on access to, and interconnection of, electronic communications networks and associated facilities

<sup>&</sup>lt;sup>16</sup> Policy Directions made by Dermot Ahern T.D. (the then) Minister for Communications, Marine and Natural Resources on 21 February 2003 and 26 March 2004.

<sup>&</sup>lt;sup>17</sup> EU Commission Recommendation of 11 February, 2003 on Relevant Product and Service Markets within the electronic communications sector susceptible to ex ante regulation in accordance with Directive 2002/21/EC of the European Parliament and the Council of 7 March 2002 on a common regulatory framework for electronic communications networks and services.

<sup>&</sup>lt;sup>18</sup> Commission Guidelines on market analysis and the assessment of significant market power under the Community regulatory framework for electronic communications networks and services.

<sup>&</sup>lt;sup>19</sup> Designation of SMP and Related Remedies – Market Analysis – Wholesale Broadband Access dated 24 February, 2005.

in document No. 05/67<sup>20</sup> and the reasoning and individual decisions set out previously in this document, each of which form part of and shall be construed with this decision instrument;

- VI. Having taken account of the submissions received in relation to document No. 05/67; and
- VII. Pursuant to Regulations 27 (4) of the European Communities (Electronic Communications Networks and Services) (Framework) Regulations 2003<sup>21</sup>, Regulations 9 and 14 of the European Communities (Electronic Communications Networks and Services) (Access) Regulations 2003.

### 2 **DEFINITIONS**

2.1 In this decision instrument, unless the context otherwise suggests:

"Access Regulations" means the European Communities (Electronic Communications Networks and Services) (Access) Regulations 2003;

"amendment" means, in respect of a tariff for any existing product, a change, adjustment, modification and any other cognate word or expression;

**"bundle"** means a package, consisting of both a product and one or more goods and / or services, which is on offer or sale to end users;

"decision instrument" means this decision instrument;

"discount" means an offer or sale of a product at less than its standard price. Examples shall include: a price reduction (including a volume related price reduction) a rebate, a reimbursement, a refund, set-off and any other cognate words or expressions;

**"end user"** has the same meaning as in Regulation 2 (1) of the European Communities (Electronic Communications Networks and Services) (Framework) Regulations 2003);

"existing product" means any product on offer or on sale to end users on or prior to the effective date of this decision instrument and any new product that ComReg has decided complies with the obligations referred to in this decision instrument;

<sup>&</sup>lt;sup>20</sup> Consultation Paper - Consultation on retail minus wholesale control for the WBA market dated 19 August, 2005.

<sup>&</sup>lt;sup>21</sup> S.I. No. 307 of 2003 the European Communities (Electronic Communications Networks and Services) (Framework) Regulations 2003 which transposes Directive 2002/21/EC of the European Parliament and the Council of 7 March 2002 on a common regulatory framework for electronic communications networks and services.

"new product" means any product on offer or sale subsequent to the effective date of this decision instrument which has different functional and / or technical characteristics and / or a different tariff structure to existing products;

"product" means any *eircom* retail broadband product on offer or sale to end users which uses *eircom*'s copper network equipment to transmit data signals and shall include existing products and new products;

**"OAO"** means other authorised operator, being an authorised undertaking for the purposes of the European Communities (Electronic Communications Networks and Services) (Authorisation) Regulations 2003;

"promotion" means an offer in respect of a product which is available for a finite period of time that offers a tariff reduction but does not include incentives provided for within normal sales costs;

**"SMP"** has the same meaning as set out in Regulation 25 of the European Communities (Electronic Communications Networks and Services) (Framework) Regulations 2003; and

"working day" means a day other than Saturday, Sunday a bank holiday or a public holiday.

### 3 SMP OBLIGATIONS IN FORCE

- 3.1 By virtue of the decision 03/05 *eircom* had imposed upon it the SMP obligations of access, non-discrimination, transparency, price control and cost accounting.
- *eircom* notified an appeal to the Electronic Communications Appeals Panel ("the ECAP") on 16 March, 2005 in respect of the price control SMP obligation. On the 24 June, 2005 the Minister for Communications, Marine and Natural resources appointed the ECAP to hear the notified appeal.
- 3.3 The other SMP obligations contained in decision 03/05 are not the subject of any appeal by any OAO, or suspension order by the ECAP. Accordingly, they remain in force in their entirety and shall where necessary, be construed together with this decision instrument.

# 4 PRICE CONTROL OBLIGATION: GENERAL

- 4.1 *eircom* shall, pursuant to Regulation 14 of the Access Regulations, have obligations relating to wholesale price control in the market for WBA, in the form of a retail minus price control, for the following purposes:
  - I. Establishing a wholesale price for products, promotions, discounts and bundles in the market for WBA;

- II. Preventing the potential application by *eircom* of a price or margin squeeze in the market for WBA. In this regard, *eircom shall* not create a margin squeeze in the market for WBA; and
- III. Preventing *eircom* from sustaining prices at an excessively high level in the market for WBA.

### 5 AMENDMENTS TO EXISTING PRODUCTS

- 5.1 Sections 5.2 5.6 apply to any amendment.
- 5.2 In this section, "retail rental price" means the retail price where a single product is supported by a single wholesale offering or, the weighted average (by number of subscribers) of the retail products' individual prices where more than one product is supported by a single wholesale offering. The retail minus price control applying to existing products shall be established by reference to the following formula:

Wholesale Rental Price = retail rental price \* (1 - fixed percentage < 100%) - fixed monetary value ("the price control")<sup>22</sup>.

- 5.3 *eircom* shall ensure that it applies the price control to the equivalent wholesale offering of any existing product. Amendments shall be in accordance with the price control and as specified in Table 1 below. Table 1 may from time to time be amended by ComReg for any of the following reasons:
  - I. When an equivalent wholesale offering of any new product is introduced;
  - II. When ComReg conducts a review of the price control; or
  - III. If ComReg decides to amend the price control or the obligations referred to in this decision instrument as a result of any material unforeseen changes in market conditions that in ComReg's opinion have a material impact on the market.

50

<sup>&</sup>lt;sup>22</sup> The following is an example, for illustrative purposes, of an application of the price control in practice: if the retail price for the "1024/128 @ 48:1 (IP)" product was to be €50, the wholesale price could be no more than €50 \* (1-29%) - €4.37 = €31.13.

Table 1

Wholesale Product	Control	
Description <sup>23</sup>	Percentage (%)	Fixed Monetary Value (€)
1024/128 @ 48:1 (IP) plus usage charge <sup>24</sup>	30%	€3.70
1024/128 @ 48:1 (IP)	29%	€4.37
2048/256 @ 24:1 (IP)	28%	€4.70
3072/256 @ 24:1 (IP)	26%	€6.75
4096/256 @ 24:1 (IP)	24%	€12.64

No later than fifteen working days prior to the date that an amendment is to become operative, *eircom* shall furnish to ComReg a detailed written statement of compliance demonstrating *eircom's* compliance with the price control and the obligations referred to in this decision instrument. The statement of compliance shall make full and true disclosure of all material facts for the purpose of demonstrating compliance with the price control and the obligations referred to in this decision instrument and shall state precisely and in all respects how *eircom* is in compliance with the price control and the obligations referred to in this decision instrument. The statement of compliance shall also have appended to it, all relevant supporting documentation for the purpose of demonstrating compliance with the price control and the obligations referred to in this decision instrument. The statement of compliance shall demonstrate how any adjustments to the price of the equivalent wholesale offering of an existing product are and will be in compliance with the price control and the obligations referred to in this decision instrument.

51

<sup>&</sup>lt;sup>23</sup> The following is an explanatory note and does not form part of the decision instrument that is being notified to the EU Commission: should *eircom* amend the technical characteristics of the wholesale products between the date that the decision instrument is notified to the EU Commission and the adoption of this decision instrument, Table 1 will be updated accordingly.

<sup>&</sup>lt;sup>24</sup> Where usage charges make up a significant portion of retail or wholesale prices, these charges will be converted into average monthly revenues and the price control will be applied accordingly.

- 5.5 At the same time that *eircom* furnishes the statement of compliance referred to in section 5.4 to ComReg, it shall furnish all OAOs availing of or intending to avail of existing products, with written notification of corresponding adjustments to an equivalent wholesale offering of any existing product. *eircom* shall furnish ComReg with a copy of the notification and written confirmation that all OAOs have been furnished with the notification.
- 5.6 Upon receipt of the statement of compliance and the notification referred to in sections 5.4 and 5.5 respectively, ComReg shall review the statement of compliance. Within the fifteen working day period referred to in section 5.4 ComReg may do one or more of the following things:
  - I. Provide *eircom* with both (a) an appropriate written opinion in relation to the statement of compliance referred to in section 5.4 and (b) written confirmation that the making available or offering for sale of the existing product is conditional only upon *eircom* being in compliance with its obligation in section 5.3. Once ComReg provides *eircom* with the written opinion and confirmation referred to in this sub-section;
  - II. Request any further information from *eircom*. *eircom* shall provide the requested information by the deadline and in such format and to the level of detail as stipulated by ComReg. Upon receipt of the requested information from *eircom* and within the fifteen working day period referred to in section 5.4, ComReg may do one or more of the things referred to in sub-sections I, III or IV of this section;
  - III. Inform *eircom* in writing that the amendment(s) would in ComReg's view, not be in compliance with the price control and the obligations referred to in this decision instrument, giving reasons therefor and also inform *eircom* that the amendment if made operative will or could result in the issuing of a notification of non-compliance under Regulation 18 (1) of the Access Regulations; or
  - IV. For the purpose of further specifying requirements to be complied with by *eircom* relating to the price control and the obligations referred to in this decision instrument, issue a direction or directions to *eircom* under Regulation 17 of the Access Regulations, to refrain from making operative the corresponding adjustment(s) to the equivalent wholesale offering of any existing product.

### 6 NEW PRODUCTS

- 6.1 Sections 6.2 6.5 apply to any new product.
- 6.2 *eircom* shall not make available or offer for sale, the equivalent wholesale offering of any new product, until such time as ComReg does that which is specified in

- section 6.5, sub-section I (either within the five working day period referred to in section 6.5 or, upon completion of the analysis by ComReg referred to in section 6.5, sub-section II).
- 6.3 *eircom* shall apply the price control to the equivalent wholesale offering of any new product.
- Prior to the date that a new product is to be made available or offered for sale, *eircom* shall furnish to ComReg a detailed written submission demonstrating *eircom's* proposed compliance with the price control and the obligations referred to in this decision instrument. The submission shall make full and true disclosure of all material facts for the purpose of demonstrating proposed compliance with the price control and the obligations referred to in this decision instrument and shall state precisely and in all respects how *eircom* would be in compliance with the price control and the obligations referred to in this decision instrument. The submission referred to in this section, shall also have appended to it, all relevant supporting documentation for the purpose of demonstrating proposed compliance with the price control and the obligations referred to in this decision instrument. The submission shall demonstrate how any adjustments to the price of the equivalent wholesale offering of a new product would be in compliance with the price control and the obligations referred to in this decision instrument.
- 6.5 Upon receipt of the submission referred to in section 6.4, ComReg shall review same. Within five working days, ComReg may do one or more of the following things:
  - I. Provide *eircom* with both (a) an appropriate written opinion in relation to the submission referred to in section 6.4 and (b) written confirmation that the making available or offering for sale of the new product is conditional only upon *eircom* being in compliance with its obligation in section 6.3. Once ComReg provides *eircom* with the written opinion and confirmation referred to in this sub-section, *eircom* shall provide OAOs that are availing of any existing products (or intending to avail of any existing products) with fifteen working days' prior written notification of corresponding adjustments to the equivalent wholesale offerings of any new products before making such adjustments. *eircom* shall at the same time furnish ComReg with written confirmation that all OAOs have been furnished with the said notification;
  - II. Inform *eircom* in writing that further analysis of the submission referred to in section 6.4 is required by ComReg, in which case ComReg may request any further information from *eircom*. *eircom* shall provide any information that is requested by the deadline, in such format and to the level of detail as stipulated by ComReg. Upon receipt of the requested information, ComReg shall furnish *eircom* with an indicative timetable in writing for the completion of its further analysis. Upon completion of this analysis,

ComReg may do one or more of the things referred to in sub-sections I, III, IV or V of this section;

- III. Inform *eircom* in writing that the new product(s) would in ComReg's view, not be in compliance with the price control and the obligations referred to in this decision instrument, giving reasons therefor and also inform *eircom* that the new product(s) if made available or offered for sale, will or could result in the issuing of a notification of non-compliance under Regulation 18 (1) of the Access Regulations;
- IV. For the purpose of further specifying requirements to be complied with by *eircom* relating to the price control and the obligations referred to in this decision instrument, issue a direction or directions to *eircom* under Regulation 17 of the Access Regulations, requiring that *eircom* makes specified corresponding adjustments to the price of the equivalent wholesale offering of any new product; or
- V. For the purpose of further specifying requirements to be complied with by *eircom* relating to the price control and the obligations referred to in this decision instrument, issue a direction or directions to *eircom* under Regulation 17 of the Access Regulations, to refrain from making available or offering for sale, the equivalent wholesale offering of any new product.

### 7 PROMOTIONS

7.1 Sections 5.2 - 5.6 (excluding Table 1) in relation to amendments to existing products shall apply in like manner to promotions.

### 8 DISCOUNTS AND BUNDLES

8.1 Sections 6.2 – 6.5 in relation to new products shall apply in like manner to discounts and bundles and shall apply to an adjustment to a discount or bundle available, or on sale, on or prior to the effective date of this decision instrument and to a new discount or bundle that is made available or offered for sale after the effective date of this decision instrument.

## 9 STATUTORY POWERS NOT AFFECTED

9.1 Nothing in this decision instrument shall operate to limit ComReg in the exercise and performance of its statutory powers or duties under any primary or secondary legislation (in force prior to or after the effective date of this decision instrument) from time to time as the occasion requires.

### 10 EFFECTIVE DATE

10.1 This decision instrument shall be effective from the [●] day of [●] 2005 until further notice by ComReg.

ISOLDE GOGGIN CHAIRPERSON THE COMMISSION FOR COMMUNICATIONS REGULATION THE [•] DAY OF [•] 2005

# Appendix C - The Broadband 'New Entrant' Discounted Cash Flow (DCF) Model

The DCF model consists of a set of spreadsheets. The costs are based on those of eircom, in line with the principle set out earlier in the document of a similarly efficient operator and therefore the cost parameters and the model itself are confidential.

The model calculates the margins required between wholesale and retail for individual wholesale products such that a return of 11.5% is achieved.

The product parameters considered in the model are:

- Monthly rental revenue (variable over time)
- Monthly connection revenue (variable over time)
- Upstream bandwidth
- downstream bandwidth
- contention ratio
- delivery technology (IP or ATM)

The model is built on a discounted cash flow basis and includes start up costs, terminal value and fixed and variable operating costs including capital costs.

The costs are categorised under the following headings:

- Marketing
- Sales
- Product management & development
- Accommodation
- Help Desk
- Billing
- Modems
- Order Handling
- Backhaul charges
- Servers and collocation
- Corporate overhead
- Internet connectivity (peering charges)
- Wholesale connection
- Wholesale rental

Revenues are limited to connection charges and rental charges. No value added service revenue is included since the model is limited to an internet connection business.

The model is designed to allow considerable flexibility in setting scenarios. The factors that can be varied include:

- Market size
- Product take-up over time (sales growth curve)
- Product mix
- Customer lifetime
- All costs over both time and volume as appropriate
- Retail revenues, by product, over time

The model provides outputs in the form of retail minus controls of the type (R-X%)-C for each retail product (or product group) based on a single wholesale product. In the formula:

- R is the retail price of the product (where necessary by weighted average price of retail products within the group),
- X% is a percentage reduction on the retail price, and
- C is a constant monetary value reduction on the retail price.

Neither X nor C can be less than zero.

Appendix E – European Commission letter – Case IE/2005/0313: Details of remedies related to case IE/2004/0093

### **EUROPEAN COMMISSION**



Brussels, 21/12/2005

SG-Greffe (2005) D/207603

Commission for Communications Regulation Block DEF Abbey Court, Irish Life Centre, Lower Abbey Street, Dublin 1 Ireland

For the attention of: Ms Isolde Goggin, Chairperson of the Commission Fax: +353 1 804 9680

Dear Ms Goggin,

SUBJECT: CASE IE/2005/0313: DETAILS OF REMEDIES RELATED TO CASE IE/2004/0093

**Article 7(3) of Directive 2002/21/EC<sup>1</sup>: No Comments** 

### I. PROCEDURE

On 24 November 2005, the Commission registered a notification from the Irish national regulatory authority, Commission for Communications Regulations ("ComReg"), concerning the details of remedies related to the market review of the wholesale broadband access ("WBA") market in Ireland, notified to and assessed by the Commission under case number IE/2004/0093.

The national consultation<sup>2</sup> runs in parallel with the Community consultation under Article 7 of the Framework Directive and the deadline for the Community consultation is 2 January 2006.

Directive 2002/21/EC of the European Parliament and of the Council of 7 March 2002 on a common regulatory framework for electronic communications networks and services ("Framework Directive"), OJ L 108, 24.4.2002, p. 33.

Commission européenne, B-1049 Bruxelles / Europese Commissie, B-1049 Brussel - Belgium. Telephone: (32-2) 299 11 11.

<sup>&</sup>lt;sup>2</sup> In accordance with Article 6 of the Framework Directive.

Pursuant to Article 7(3) of the Framework Directive, national regulatory authorities (NRAs) and the Commission may make comments on notified draft measures to the NRA concerned.

### II. DESCRIPTION OF THE DRAFT MEASURE

The notified draft measure relates to the amendment of obligations which were imposed on an interim basis following the market analysis undertaken by ComReg in accordance with Article 16 of the Framework Directive for the market review of the WBA market (i.e. market 12 of the Commission's Recommendation on Relevant Markets)<sup>3</sup> in Ireland. In that case, ComReg proposed an <u>interim</u> price control mechanism, whereby eircom was required to apply a "retail minus" price control to each of the connection and rental charges for its wholesale bitstream products at that time. Additionally, eircom was also made subject to the obligation of notifying ComReg in writing of any proposed amendments to both retail and wholesale terms and conditions, including price, in respect of its existing and future broadband access products. In that notification, ComReg also indicated that a further consultation would be held on the detailed implementation of wholesale price control, accounting separation obligations and cost accounting obligations.<sup>4</sup>

The present notification has been issued in response to this announced consultation on the detailed implementation of wholesale price control, as described above, and on the basis of the responses received, ComReg now proposes a definitive retail minus price control mechanism.

For the establishment of the "minus", ComReg proposes that eircom's costs and revenues will be used as the basis for establishing those of a similarly efficient operator (taking account of differences in scale, costs and evolution of prices). For the assessment of margins, ComReg proposes the application of the discounted cash flow ("DCF") methodology to be carried over a 5-year period, and that the margin squeeze test should be applied on a product-by-product basis, with a separate control for each wholesale and retail product pair. ComReg will review the "minus" margin annually.

Unlike where eircom makes changes to the prices of existing products or to promotions around existing products, eircom will be required to submit any proposals to ComReg in advance where it introduces new products or new discount schemes, and will be required to provide the information necessary to assess whether or not there is a potential margin squeeze. Any proposed change to eircom's wholesale charges should also be notified to ComReg in advance.

Commission Recommendation 2003/311/EC of 11 February 2003 on relevant product and services

markets within the electronic communications sector susceptible for *ex ante* regulation in accordance with the Framework Directive, OJ L 114, 8.5.2003, p. 45 ("the Recommendation").

The Commission notes that when ComReg will implement the details underpinning the accounting separation and cost accounting obligations, they will have to be notified under Article 7(3) of the Framework Directive.

### III. NO COMMENTS

The Commission has examined the draft measure and has no comments. <sup>5</sup>

Pursuant to Article 7(5) of the Framework Directive, ComReg may adopt the resulting draft measures and, where it does so, shall communicate them to the Commission.

The Commission's position on the particular notification is without prejudice to any position it may take *vis-à-vis* other notified draft measures.

Pursuant to Point 12 of Recommendation 2003/561/EC<sup>6</sup> the Commission will publish this document on its website. The Commission does not consider the information contained herein to be confidential. You are invited to inform the Commission<sup>7</sup> within three working days following receipt whether you consider that, in accordance with Community and national rules on business confidentiality, this document contains confidential information which you wish to have deleted prior to such publication. You should give reasons for any such request.

Yours faithfully, For the Commission, Neelie Kroes Member of the Commission

\_

Pursuant to Article 7(3) of the Framework Directive.

<sup>&</sup>lt;sup>6</sup> Commission Recommendation 2003/561/EC of 23 July 2003 on notifications, time limits and consultations provided for in Article 7 of Directive 2002/21/EC, OJ L 190, 30.7.2003, p. 13.

Your request should be sent either by email: INFSO-COMP-ARTICLE7@cec.eu.int or by fax: +32.2.298.87.82.